## UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF ALABAMA SOUTHERN DIVISION

		)	
In re:		)	Chapter 11
		)	
WALTER ENERGY, INC., et al.		)	Case No. 15-02741-TOM11
		)	
I	Debtors. <sup>1</sup>	)	Jointly Administered
		)	
		)	

ORDER (A) AUTHORIZING THE DEBTORS TO OBTAIN SENIOR SECURED POSTPETITION FINANCING, (B) AUTHORIZING CONTINUED POSTPETITION USE OF CASH COLLATERAL, (C) GRANTING ADEQUATE PROTECTION TO PREPETITION SECURED PARTIES AND (D) GRANTING RELATED RELIEF

Upon the motion (the "Motion") [Docket No. 1646]<sup>2</sup> dated January 14, 2016 of the above-captioned debtors and debtors in possession (collectively, the "Debtors"), pursuant to sections 105, 361, 362, 363, 364, 507 and 552(b) of title 11 of the United States Code (the "Bankruptcy Code") and Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules") 2002, 4001, 6004 and 9014, seeking, among other things, entry of an order (this "Order"):

(I) authorizing the Debtors to obtain senior secured postpetition financing in the form of a non-amortizing multiple draw term loan facility (the "<u>DIP Facility</u>") in an aggregate principal amount not to exceed \$50 million (the loans under the DIP Facility, the "<u>DIP Loans</u>") subject to and pursuant to

The Debtors in these cases, along with the last four digits of each Debtor's federal tax identification number, are: Walter Energy, Inc. (9953); Atlantic Development and Capital, LLC (8121); Atlantic Leaseco LLC (5308); Blue Creek Coal Sales, Inc. (6986); Blue Creek Energy, Inc. (0986); J.W. Walter, Inc. (0648); Jefferson Warrior Railroad Company, Inc. (3200); Jim Walter Homes, LLC (4589); Jim Walter Resources, Inc. (1186); Maple Coal Co. LLC (6791); Sloss-Sheffield Steel & Iron Company (4884); SP Machine, Inc. (9945); Taft Coal Sales & Associates, Inc. (8731); Tuscaloosa Resources, Inc. (4869); V Manufacturing Company (9790); Walter Black Warrior Basin LLC (5973); Walter Coke, Inc. (9791); Walter Energy Holdings, LLC (1596); Walter Exploration & Production LLC (5786); Walter Home Improvement, Inc. (1633); Walter Land Company (7709); Walter Minerals, Inc. (9714); and Walter Natural Gas, LLC (1198). The location of the Debtors' corporate headquarters is 3000 Riverchase Galleria, Suite 1700, Birmingham, Alabama 35244-2359.

<sup>&</sup>lt;sup>2</sup> Capitalized terms used but not otherwise defined herein have the meanings ascribed to them in the Motion or the *Amended Final Order (A) Authorizing Postpetition Use of Cash Collateral, (B) Granting Adequate Protection to Prepetition Secured Parties and (C) Granting Related Relief* [Docket No. 797] (the "Amended Final Cash Collateral Order"), as applicable.

the terms and conditions set forth in this Order and the *Senior Secured Superpriority Debtor-in-Possession Credit Agreement* (as amended, supplemented, or otherwise modified from time to time, which shall be substantially in the form annexed hereto as **Exhibit A**, the "DIP Credit Agreement" and, together with all agreements, documents, and instruments executed and delivered in connection therewith, as hereafter amended, supplemented, or otherwise modified from time to time, in accordance with this Order, the "DIP Financing Documents"), among Walter Energy, Inc., as borrower (the "Borrower" or "Walter"), each domestic subsidiary of the Borrower (collectively, the "Guarantors"), each lender from time to time party thereto (collectively, the "DIP Lenders"), and Citibank, N.A., as administrative agent, collateral agent, and escrow agent (collectively, the "DIP Agent") on behalf of the DIP Lenders;

- (II) authorizing the Debtors to execute and deliver, and to perform under, the DIP Credit Agreement and the other DIP Financing Documents, and to perform such other and further acts as may be necessary or appropriate in connection therewith;
- (III) granting the DIP Agent, for the benefit of itself and the DIP Lenders, allowed superpriority administrative expense claims in each of the Debtors' Chapter 11 Cases and any successor cases, including any chapter 7 cases, with respect to the DIP Facility and all obligations and indebtedness owing thereunder, under the DIP Financing Documents and this Order (collectively, and including all "Obligations" as defined in the DIP Credit Agreement, the "DIP Obligations"), subject to the priorities set forth herein;
- (IV) granting the DIP Agent, for the benefit of itself and the DIP Lenders, automatically perfected priming security interests in, and liens on, all of the DIP Collateral (as defined herein), including, but not limited to, Cash Collateral, subject to the priorities set forth herein;
- (V) authorizing the Debtors, subject to and pursuant to the terms and conditions set forth in this Order, to continue to (a) use Cash Collateral and (b) provide adequate protection on account of the diminution in the value of the Prepetition Collateral, including Cash Collateral, as a consequence of the Debtors' use, sale or lease of the Prepetition Collateral, including any Cash Collateral, the imposition of the automatic stay, the Debtors' incurrence of debt under the DIP Credit Agreement and the priming of the Prepetition Secured Parties' liens by the DIP Facility, to the Prepetition Secured Parties (as defined herein) who have been granted prepetition liens and security interests under the following documents (collectively, the "Prepetition Debt Documents"):

- (a) the Credit Agreement, dated as of April 1, 2011 (as amended, restated, amended and restated, waived, supplemented or otherwise modified from time to time, the "Credit Agreement" and together with all mortgage, security, pledge, guaranty and collateral agreements, including the 1L Security Agreement (as defined below) and all other documentation executed in connection with any of the foregoing, each as amended, restated, amended and restated, waived, supplemented or otherwise modified from time to time, the "First Lien Credit Documents"), among the Walter, as U.S. borrower, Western Coal Corp. 3 and Walter Energy Canada Holdings, Inc., as Canadian borrowers, the lenders from time to time party thereto (collectively, the "First Lien Lenders"), and Morgan Stanley Senior Funding, Inc., as administrative agent (in such capacity, the "Administrative Agent");
- (b) the Indenture, dated as of September 27, 2013 (as amended, waived, supplemented or otherwise modified from time to time, the "First Lien Indenture" and together with all mortgage, security, pledge, guaranty and collateral agreements, including the 1L Notes Collateral Agreement (as defined below) and all other documentation executed in connection with the foregoing, each as amended, restated, amended and restated, waived, supplemented or otherwise modified from time to time, the "First Lien Indenture Documents" and together with the First Lien Credit Documents, the "First Lien Documents"), among Walter, as issuer, certain of its subsidiaries, as guarantors, and Wilmington Trust, National Association (as successor to Union Bank, N.A.), as trustee (in such capacity, the "First Lien Trustee") and collateral agent for the noteholders from time to time of the 9.50% Senior Secured Notes due 2019 (collectively, the "First Lien Noteholders" and, together with the First Lien Trustee, the Administrative Agent and the First Lien Lenders, the "First Lien Secured Parties");
- (c) the Indenture, dated as of March 27, 2014 (as amended, waived, supplemented or otherwise modified from time to time, the "Second Lien Indenture" and together with all mortgage, security, pledge, guaranty and collateral agreements, including the 2L Collateral Agreement (as defined below) and all other documentation executed in connection with the foregoing, each as amended, restated, amended and restated, waived, supplemented or

Western Coal Corp. was a Canadian borrower at the time of entry into the First Lien Credit Agreement and related documents. In connection with a 2012 restructuring, substantially all of Western Coal Corp.'s assets were transferred to Walter Canadian Coal Partnership, and Western Coal Corp. was dissolved, with its remaining assets (including its partnership interest in Walter Canadian Coal Partnership) distributed to Walter Energy Canada Holdings, Inc.

otherwise modified from time to time, the "Second Lien Indenture Documents"), among Walter, as issuer, certain of its subsidiaries, as guarantors, and BOKF, N.A. (as successor to Wilmington Trust, National Association), as trustee (in such capacity, the "Second Lien Trustee" and, together with the First Lien Trustee, the "Indenture Trustees") and collateral agent for the noteholders from time to time of the 11.0%/12.0% Senior Secured Second Lien PIK Toggle Notes due 2020 (collectively, the "Second Lien Noteholders" and, together with the Second Lien Trustee and the First Lien Secured Parties, collectively, the "Prepetition Secured Parties");

- (VI) authorizing the DIP Agent and the DIP Lenders to exercise remedies under the DIP Financing Documents upon the occurrence and during the continuance of an Event of Default (as defined herein);
- (VII) waiving the Debtors' right to surcharge against the DIP Collateral or the Prepetition Collateral pursuant to Bankruptcy Code section 506(c);
- (VIII) modifying the automatic stay imposed by Bankruptcy Code section 362 to the extent necessary to implement and effectuate the terms of this Order; and
- (IX) waiving any applicable stay with respect to the effectiveness and enforceability of this Order (including a waiver pursuant to Bankruptcy Rule 6004(h)).

The Court having considered the Motion, the evidence submitted or proffered and the arguments of counsel made at the hearing on the Motion held on January 28, 2016 (the "Hearing"); and proper and sufficient notice of the Motion and the Hearing having been given in accordance with Bankruptcy Rules 2002, 4001(b), (c) and (d), and 9014; and the Hearing to consider the relief requested in the Motion having been held and concluded; and all objections, if any, to the relief requested in the Motion and to the entry of this Order having been withdrawn, resolved, or overruled by the Court; and it appearing to the Court that granting the relief requested is fair and reasonable and in the best interests of the Debtors, their estates, creditors and parties-in-interest; and after due deliberation and consideration, and for good and sufficient cause appearing therefor;

## IT IS FOUND, DETERMINED, ORDERED AND ADJUDGED, that:

- 1. The Motion. The Motion is granted as set forth herein. The DIP Facility is authorized and approved, and the grant of adequate protection for the consensual use of Prepetition Collateral, including Cash Collateral, is authorized, subject to the terms and conditions set forth in this Order. All objections to the Motion and entry of this Order, to the extent not withdrawn or resolved, are hereby overruled.
- 2. Jurisdiction. This Court has core jurisdiction over the above-captioned Chapter 11 cases (the "Chapter 11 Cases") commenced on July 15, 2015 (the "Petition Date"), the Motion, the objections, and the parties and property affected hereby pursuant to 28 U.S.C. §§ 157(b) and 1334. Venue is proper before this Court pursuant to 28 U.S.C. §§ 1408 and 1409.
- 3. Statutory Committees. On July 30, 2015, the Office of the Bankruptcy Administrator for the Northern District of Alabama (the "Bankruptcy Administrator") appointed the official committee of unsecured creditors (the "Creditors' Committee") in the Chapter 11 Cases [Docket Nos. 268 and 342]. Pursuant to an order entered on July 30, 2015 [Docket No. 264], a committee of retired employees (the "Retiree Committee") has been formed pursuant to Bankruptcy Code sections 1114(c)(2) and 1114(d). As of the date hereof, no other official committees have been appointed in the Chapter 11 Cases.
- 4. *Notice*. The Debtors have caused notice of the Motion, the relief requested therein and the Hearing to be served on: (a) the Bankruptcy Administrator; (b) counsel to the Administrative Agent; (c) counsel to the First Lien Trustee; (d) counsel to the Second Lien Trustee; (e) counsel to the Steering Committee; (f) counsel to certain lenders and noteholders holding the majority in amount of the First Lien Obligations that have provided commitments to fund the DIP Loans (collectively, the "Backstop Parties"); (g) counsel to the Creditors'

Committee; (h) counsel to the Retiree Committee; (i) the Internal Revenue Service; (j) the Securities and Exchange Commission; (k) the U.S. Environmental Protection Agency; (l) counsel to the United Mine Workers of America (the "<u>UMWA</u>"); (m) counsel to the United Steelworkers; (n) the United States Attorney for the Northern District of Alabama; and (o) all persons or entities that have filed a request for service of filings in these Chapter 11 Cases pursuant to Bankruptcy Rule 2002 (the "<u>Notice Parties</u>"). Under the circumstances, the notice given by the Debtors of the Motion, the relief requested therein and the Hearing constitutes due and sufficient notice thereof and complies with Bankruptcy Rules 2002, 4001(b), (c) and (d), and 9014.

- 5. Debtors' Stipulations. Except as modified by this Order and subject to the Creditors' Committee's right to assert Claims and Defenses (as defined below) by the Challenge Deadline (as defined herein) as set forth in paragraph 21 below, all admissions, stipulations and agreements of the Debtors set forth in paragraph 5 of the Amended Final Cash Collateral Order are incorporated herein by reference as if fully set forth herein. Such admissions, stipulations and agreements remain binding on the Debtors and their respective representatives, successors and assigns, including any chapter 7 trustee, on each of the Debtors' estates, all creditors thereof and each of their respective representatives, successors and assigns, including, but not limited to, any trustee or other representative appointed by the Court, whether such trustee or representative is appointed in cases under chapter 11 or chapter 7 of the Bankruptcy Code.
- 6. Effect on Amended Final Cash Collateral Order. The Amended Final Cash Collateral Order is amended and superseded by this Order solely to the extent set forth herein; provided, however, that the Adequate Protection Obligations, Superpriority Claims, Adequate Protection Liens, waivers, releases and other rights and protections granted to the Prepetition

Secured Parties in the Amended Final Cash Collateral Order remain in full force and effect, except as modified herein.

- 7. Findings Regarding the DIP Loans and the Use of Prepetition Secured Parties' Collateral, Including the Cash Collateral.
  - (a) Good cause has been shown for the entry of this Order.
- (b) The Debtors require postpetition financing and continued use of the Prepetition Secured Parties' Collateral, including the Cash Collateral, in order to, among other things, fund their operations and pay for the costs and expenses related to the administration of the Chapter 11 Cases until the closing of the sale(s) pursuant to the Sale Motion. Absent granting the relief set forth in this Order, the Debtors' estates and their business will be irreparably harmed.
- (c) The Debtors are unable to obtain financing on more favorable terms and conditions from sources other than the DIP Lenders pursuant to, and for the purposes set forth in, the DIP Financing Documents and are unable to obtain unsecured credit allowable under Bankruptcy Code section 503(b)(1) as an administrative expense. The Debtors also are unable to obtain secured credit allowable under Bankruptcy Code sections 364(c)(1), 364(c)(2), and 364(c)(3) without granting priming liens under Bankruptcy Code section 364(d)(1) and a DIP Superpriority Claim (as defined herein) on the terms and conditions set forth in this Order and the DIP Financing Documents.
- (d) The terms upon which the Debtors will be permitted to continue to use Cash Collateral pursuant to this Order and the terms of the DIP Loans are fair and reasonable, reflect the Debtors' exercise of prudent business judgment consistent with their fiduciary duties and constitute reasonably equivalent value and fair consideration.

- (e) The Steering Committee, the Administrative Agent, the Credit Agreement Collateral Agent, the First Lien Trustee, the Second Lien Trustee, the 1L Notes Collateral Agent, and the 2L Notes Collateral Agent, as applicable, have consented to (or, as applicable, have been directed to, or it has been asserted that they are deemed to have consented to pursuant to the Amended and Restated Intercreditor Agreement, dated as of March 27, 2014 (as amended, supplemented or otherwise modified from time to time, the "Intercreditor Agreement")), conditioned upon the entry of this Order, the Debtors' entry into the DIP Facility, including the granting of priming liens and claims in connection therewith, and the continued use of Cash Collateral, on the terms and conditions set forth in this Order.
- Parties' Collateral, including the continued use of Cash Collateral, have been the subject of extensive negotiations conducted in good faith and at arm's length among the Debtors, the DIP Agent, the DIP Lenders, the Credit Agreement Collateral Agent (on behalf of the First Lien Lenders), the Administrative Agent, and the Backstop Parties, and (ii) the DIP Obligations, and this Order shall be deemed to have been extended by the DIP Agent and the DIP Lenders in "good faith" as such term is used in Bankruptcy Code sections 363(m) and 364(e), and in express reliance upon the protections set forth therein, and shall be entitled to the full protections of Bankruptcy Code sections 363(m) and 364(e) in the event that this Order or any provision hereof is vacated, reversed or modified on appeal or otherwise.
  - 8. Authorization of the DIP Financing Loans and the DIP Financing Documents.
- (a) The DIP Financing Documents are hereby approved. The Debtors are authorized and empowered to enter into, and perform under, the DIP Financing Documents and, in the case of Borrower, to borrow, subject to the terms and conditions of this Order and the DIP

Financing Documents, the DIP Loans in an aggregate principal amount not to exceed \$50 million solely for the purposes specified in the DIP Term Sheet filed with the Motion, and any other purposes specified in the DIP Credit Agreement to be filed with the Court. Upon the Closing (as defined in the Asset Purchase Agreement) of the transactions under the Asset Purchase Agreement, the DIP Obligations shall be indefeasibly and immediately paid in full in cash from the cash consideration provided by the Buyer under the Asset Purchase Agreement.

- (b) In furtherance of the foregoing and without further approval of this Court, each Debtor is authorized and directed to perform all acts (and to the extent such acts have already occurred, such acts are hereby ratified) and to execute and deliver all instruments and documents that the DIP Agent determines to be reasonably required or necessary for the Debtors' performance of their obligations under the DIP Financing Documents, including without limitation:
  - (i) the execution, delivery and performance of the DIP Financing Documents;
  - (ii) the non-refundable payment to the DIP Agent and the DIP Lenders, as the case may be, of the fees set forth in the DIP Financing Documents, including, but not limited to, the Upfront Fee, Drawdown Fee and Put Option Premium and the fees set forth in the Agent Fee Letter and the other fees, costs and expenses as may be due from time to time as provided in this Order and the DIP Financing Documents, including, but not limited to, reasonable attorneys' fees and expenses and the fees and expenses of financial advisors, consultants and other professionals as provided in the DIP Financing Documents;
  - (iii) the granting of all liens and claims with respect to, and the making any payments in respect of, Adequate Protection Obligations (as defined herein) to the extent provided for in this Order; and
  - (iv) the performance of all other acts required under or in connection with this Order and/or the DIP Financing Documents.

- (c) The DIP Financing Documents constitute legal, valid, binding and non-avoidable obligations of the Debtors, enforceable against each of the Debtors and their estates in accordance with the terms of this Order and the DIP Financing Documents. The failure to specifically include any particular provision of the DIP Financing Documents in this Order shall not diminish or impair the efficacy of such provision, it being the intent of the Court that the DIP Financing Documents be authorized and approved in their entirety.
- (d) No obligation, payment, transfer or grant of security or superpriority claim by the Debtors under the DIP Financing Documents, this Order or the Amended Final Cash Collateral Order, to the extent not amended by this Order, shall be subject to contest, attack, objection, recoupment, defense, setoff, counterclaim, avoidance, recharacterization, reclassification, reduction, disallowance, recovery, disgorgement, attachment, "claim" (as defined in the Bankruptcy Code), impairment, or subordination (whether equitable, contractual, or otherwise) or other challenge of any kind pursuant to the Bankruptcy Code or applicable nonbankruptcy law.
- 9. Budget Compliance and Reporting. Attached as **Exhibit B** hereto and incorporated by reference herein is the 11-week budget, for the period starting on the week ending January 23, 2016, setting forth the projected financial operations of the Debtors, which budget has been approved by the Backstop Parties (the "Approved Budget"). As soon as available but in any event within five (5) business days after the end of each Budget Testing Period (with the first variance report due on February 19, 2016), the Debtors shall deliver to the DIP Agent and the Backstop Parties who have signed a non-disclosure agreement, with copies to the Steering Committee Advisors, the advisors to the Administrative Agent, the advisors to the Creditors' Committee and counsel to the Second Lien Trustee, a variance report showing

comparisons of actual amounts for each line item against the budgeted amounts for such line item in the Approved Budget for the preceding two-week period and on a cumulative basis from the immediately preceding Budget Testing Period accompanied by a narrative explanation; provided that, for purposes of the first Budget Testing Period, such variance report shall compare cumulative actuals and budgeted amounts from the first week of the Approved Budget through the week ending February 13, 2016; provided further that the variance report may be shared on a confidential basis with those DIP Lenders who have signed a non-disclosure agreement, and those Prepetition Secured Parties that have signed a non-disclosure agreement or are otherwise subject to confidentiality restrictions pursuant to the Prepetition Debt Documents. The Debtors shall not allow the: (a) aggregate amount of the actual net cash flow of the Debtors of the type set forth in the line item "Cumulative Net Cash Flow" on the Approved Budget for any Budget Testing Period to be lower than the budgeted amount set forth on the line item "Cumulative Net Cash Flow" on the Approved Budget for such Budget Testing Period by more than \$10,000,000; and (b) aggregate amount of the actual disbursements of the Debtors of the type set forth in the line item "Cumulative Disbursements" on the Approved Budget for any Budget Testing Period to be greater than the budgeted amount set forth on the line item "Cumulative Disbursements" on the Approved Budget for such Budget Testing Period by more than the greater of (x) \$5,000,000 and (y) 5.0% of such budgeted amount. For purposes of the Cumulative Net Cash Flow and Cumulative Disbursements variance tests set forth above, (i) "Budget Testing Period" shall mean the two-week period beginning with the week in which the Effective Date (as defined in the DIP Credit Agreement) occurs, and each cumulative period beginning with the week in which the Effective Date occurs and ending every two weeks after the first two week period and (ii) "Cumulative Net Cash Flow" and "Cumulative Disbursements" shall not include (w) the fees

and expenses of the Backstop Parties' Advisors (as defined herein) and the Steering Committee Advisors and the Consultants (each as defined herein), and fees and expenses of the professionals retained by the DIP Agent, the Administrative Agent, the First Lien Trustee and the Debtors, (x) the Cash Collateral used with respect to the Approved Collateralized Obligations, (y) any key employee retention payments pursuant to the Memorandum Opinion and Order (A) Granting the Debtors' Motion Approving the Debtors' Key Employee Retention Plan and (B) Granting Related Relief [Docket No. 1490] (the "KERP Order") and any other key employee retention payments approved by the Required Lenders and the Court, and (z) any other amounts excluded from "Cumulative Net Cash Flow" or "Cumulative Disbursements" in the Amended Final Cash Collateral Order. The Debtors shall not allow the cumulative amount of the actual capital expenditures, as calculated on a GAAP basis, for any Capital Expenditure Testing Period to exceed the budgeted amount set forth in the Capital Expenditure Budget for such Capital Expenditure Testing Period, also calculated on a GAAP basis, by \$1.5 million. For purposes of the capital expenditure variance test, (i) "Capital Expenditure Testing Period" shall mean each cumulative period beginning January 1, 2016 and ending at the end of each calendar month thereafter and (ii) "Capital Expenditure Budget" shall mean the budget projecting capital expenditures of the Debtors attached hereto as **Exhibit C** for period starting from January 1, 2016 through March 31, 2016 on a monthly basis, which "Capital Expenditure Budget" shall be in form and substance acceptable by the Backstop Parties in their sole discretion.

10. Except as otherwise agreed to by the Backstop Parties, or as expressly permitted herein or in the DIP Credit Agreement, the Debtors' use of the DIP Loans and Cash Collateral shall be only permitted pursuant to the terms of, and in accordance with, the Approved Budget subject to permitted variances. The Backstop Parties, the DIP Agent, the Steering Committee,

the Administrative Agent and the First Lien Trustee, as applicable, shall have no obligation with respect to the Debtors' use of the DIP Loans and Cash Collateral, and shall not be obligated to ensure or monitor the Debtors' compliance with the Approved Budget or to pay (directly or indirectly from the DIP Loans or Cash Collateral) any expenses incurred or authorized to be incurred pursuant to the Approved Budget. The consent of the Backstop Parties, the Steering Committee, the DIP Agent, the Administrative Agent and the First Lien Trustee, as applicable, to the use of DIP Loans or Cash Collateral pursuant to this Order and the DIP Financing Documents shall not be construed as consent to the use of any DIP Loans or Cash Collateral after the occurrence of an Event of Default, regardless of whether the aggregate funds shown on the Approved Budget have been expended.

- 11. *DIP Claims and Liens*. As security for the DIP Obligations, the DIP Agent, for the benefit of the DIP Lenders, is hereby granted the following claims, liens, and security interests ((a) and (b) below, collectively, the "<u>DIP Claims and Liens</u>"):
- (a) *DIP Superpriority Claim*. Pursuant to Bankruptcy Code section 364(c)(1), an allowed joint and several superpriority administrative expense claim against each of the Debtors on account of the DIP Obligations (collectively, the "<u>DIP Superpriority Claim</u>"), with priority over any and all administrative expenses and claims asserted against the Debtors or their estates, at any time existing or arising, of any kind or nature whatsoever, including, but not limited to, the Prepetition Secured Parties' Superpriority Claims (as defined herein) and the administrative expenses of the kinds specified in or ordered pursuant to Bankruptcy Code sections 105, 326, 327, 328, 330, 331, 361, 362, 363, 364, 365, 503, 506, 507(a), 507(b), 546, 552, 726, 1113 and 1114, and any other provision of the Bankruptcy Code, whether or not such

expenses or claims may become secured by a judgment lien or other non-consensual lien, levy or attachment, subject and subordinate only to the Carve-Out.

(b) DIP Liens. Pursuant to Bankruptcy Code sections 361, 362, 364(c)(2), 364(c)(3) and 364(d), a valid, binding, continuing, enforceable, fully-perfected, non-avoidable, automatically and properly perfected first priority senior priming lien on, and security interest in (such liens and security interests, the "DIP Liens"), all property, whether now owned or hereafter acquired or existing and wherever located, of each Debtor and each Debtor's "estate" (as created pursuant to Bankruptcy Code section 541(a)), of any kind or nature, whatsoever, real or personal, tangible or intangible, and now existing or hereafter acquired or created, including without limitation, all cash, accounts, inventory, goods, contract rights, mineral rights, instruments, documents, chattel paper, patents, trademarks, copyrights, and licenses therefor, accounts receivable, receivables and receivables records, general intangibles, payment intangibles, tax or other refunds, insurance proceeds, letters of credit, contracts, owned real estate, real property leaseholds, fixtures, deposit accounts, commercial tort claims, securities accounts, instruments, investment property, letter-of-credit rights, supporting obligations, machinery and equipment, real property, leases (and proceeds from the disposition thereof), all of the issued and outstanding capital stock of each Debtor (other than Walter), other equity or ownership interests held by a Debtor, including equity interests in subsidiaries and non-wholly-owned subsidiaries, money, investment property, causes of action (including causes of action and the proceeds thereof arising under Bankruptcy Code section 549 and any related action under Bankruptcy Code section 550) and proceeds of any causes of action under Bankruptcy Code sections 502(d), 544, 545, 547, 548, 550 or 553 or any other avoidance action under the Bankruptcy Code or applicable nonbankruptcy law (collectively, the "Avoidance Actions"), Prepetition Collateral, including Cash Collateral, and all cash and non-cash proceeds, rents, products, substitutions, accessions, and profits of any of the collateral described above, documents, vehicles, intellectual property, securities, partnership or membership interests in limited liability companies and capital stock, including, but not limited to, the products, proceeds and supporting obligations thereof, whether in existence on the Petition Date or thereafter created, acquired, or arising and wherever located (all such property, the "DIP Collateral"), which liens and security interests shall be senior to any and all other liens and security interests, including the Adequate Protection Liens (as defined herein) and the liens granted to the Prepetition Secured Parties under the Prepetition Debt Documents, other than the (i) Carve-Out, (ii) Permitted Priority Liens, if any, and (iii) Approved Liens. The DIP Liens shall not be (i) subject or subordinate to, or pari passu with, (x) any lien or security interest that is avoided and preserved for the benefit of the Debtors and their estates under Bankruptcy Code section 551 or (y) any lien or security interest arising on or after the Petition Date but subject to the Carve-Out or (ii) other than as set forth herein, subordinated to or made pari passu with any other lien, claim or security interest under Bankruptcy Code sections 363 or 364 or otherwise.

(c) For purposes of this Order and the Amended Final Cash Collateral Order, prior to the Closing, the "Carve-Out" shall mean the sum of: (A) all fees required to be paid to the Clerk of the Court and the Bankruptcy Administrator (without regard to the Carve-Out Trigger Notice (as defined herein)); (B) reasonable fees and expenses up to \$25,000 in the aggregate incurred by a trustee appointed under Bankruptcy Code section 726(b) (without regard to the Carve-Out Trigger Notice); (C) subject to the Committee Monthly Cap (as defined herein) with respect to Professional Fees incurred by Professional Persons (as defined herein) retained by the Creditors' Committee, the Retiree Committee or any other statutory committee appointed

in the Chapter 11 Cases, to the extent allowed, whether by interim order, procedural order or otherwise, all accrued and unpaid reasonable fees, costs, and expenses (the "Professional Fees") incurred by persons or firms retained by the Debtors, the Creditors' Committee, the Retiree Committee or any other statutory committee appointed in the Chapter 11 Cases pursuant to Bankruptcy Code section 327, 328, or 363 (collectively, the "Professional Persons") at any time before or on the day of delivery by the DIP Agent or the Backstop Parties of a Carve-Out Trigger Notice (the "Pre-Trigger Date Fees"); and (D) after the delivery by the DIP Agent or the Backstop Parties of the Carve-Out Trigger Notice (the "Trigger Date"), to the extent allowed at any time, whether by interim order, procedural order or otherwise, the payment of (1) all Professional Fees of Professional Persons retained by the Debtors and (2) subject to the Committee Monthly Cap, the payment of Professional Fees of Professional Persons incurred by the Creditors' Committee, the Retiree Committee or any other statutory committee appointed in the Chapter 11 Cases, not to exceed \$5 million in the aggregate for clauses (1) and (2) incurred after the Trigger Date (the amount set forth in this clause (D) being the "Post-Carve Out Trigger Notice Cap"); provided that nothing herein shall be construed to impair the ability of any party to object to the fees, expenses, reimbursement or compensation described in clauses (C) or (D) above, on any grounds. On the day on which a Carve-Out Trigger Notice is given to the Debtors, such Carve-Out Trigger Notice also shall constitute a demand to the Debtors to utilize all cash on hand as of such date and any available cash thereafter held by any Debtor to fund a reserve in an aggregate amount equal to the accrued and unpaid Pre-Trigger Date Fees plus the Post-Carve Out Trigger Notice Cap, and the Debtors shall deposit and hold any such amounts in a segregated account at a financial institution selected by the Debtors for such purpose and solely for the benefit of the Professional Persons entitled thereto. The reserved funds shall be released

from time to time from the segregated account to pay when due any Pre-Trigger Date Fees and any fees and expenses incurred after Post-Carve Out Trigger Notice that are included in the Post-Carve Out Trigger Notice Cap under clause (D) above. Such account and amounts therein shall be free and clear of all liens, claims and interests of any party other than the Professional Persons entitled thereto. Notwithstanding the foregoing, (X) the Carve-Out shall not include, apply to or be available for any fees or expenses incurred by any party in connection with (1) the investigation, preparation, initiation or prosecution of any claims, causes of action, proceeding, adversary proceeding or other litigation against any of the Prepetition Secured Parties (in such capacity), any of the DIP Financing Lenders or the DIP Agent, including challenging the amount, validity, perfection, priority or enforceability of or asserting any defense, counterclaim or offset to, the obligations, liens and security interests granted under the Prepetition Debt Documents, in favor of the Prepetition Secured Parties, the Amended Final Cash Collateral Order, the DIP Financing Documents, in favor of the DIP Financing Lenders or the DIP Agent, or under this Order, including, but not limited to, for lender liability or pursuant to Bankruptcy Code section 105, 510, 544, 547, 548, 549, 550 or 552, applicable nonbankruptcy law or otherwise; (2) attempts to modify any of the rights granted to the Prepetition Secured Parties, the DIP Lenders or the DIP Agent hereunder or under the Amended Final Cash Collateral Order; (3) attempts to prevent, hinder or otherwise delay any of the Prepetition Secured Parties', the DIP Lenders' or the DIP Agent's enforcement or realization upon any Collateral in accordance with the Prepetition Debt Documents, the DIP Financing Documents or this Order; or (4) paying any amount on account of any claims arising before the Petition Date unless such payments are approved by an order of this Court, in the Approved Budget or otherwise consented to by the Backstop Parties in their sole discretion, and (Y) so long as the Carve-Out Trigger Notice shall

not have been delivered, the Carve-Out shall not be reduced by the payment of Professional Fees allowed at any time by this Court. Any claim incurred in connection with any of the activities described above shall not be allowed, treated or payable as an administrative expense claim for purposes of Bankruptcy Code section 1129(a)(9)(A). For purposes of the foregoing, "Carve-Out Trigger Notice" shall mean a written notice delivered by the DIP Agent or the Backstop Parties to the Debtors, Debtors' counsel, the Administrative Agent, the First Lien Trustee, the Bankruptcy Administrator, counsel to the Creditors' Committee and counsel to the Retiree Committee, upon the occurrence and during the continuance of an Event of Default, stating that the Post-Carve Out Trigger Notice Cap has been invoked. For the avoidance of doubt and notwithstanding anything to the contrary herein, in the Prepetition Debt Documents or the DIP Financing Documents, the Carve-Out shall be senior to (i) all liens and claims arising out of the Prepetition Debt Documents, including the Prepetition Liens, (ii) the Adequate Protection Liens, (iii) the First Lien Superpriority Claim, (iv) the Second Lien Superpriority Claim, (v) all liens and claims arising out of the DIP Financing Documents and granted to the DIP Agent and the DIP Lenders under this Order, including the DIP Claims and Liens, and (vi) any and all other forms of adequate protection, liens or claims securing or relating to the Prepetition Obligations or the DIP Obligations. From and after the Closing, however, the "Carve-Out" shall mean zero dollars (\$0) and shall be deemed deleted from this Order and the Amended Final Cash Collateral Order in any and all places and for any and all purposes there shall be no "Carve-Out" or similar concept in these Chapter 11 Cases or any successor case, including any chapter 7 case.

12. Events of Default. Unless expressly waived in accordance with the consents required in the DIP Financing Documents, each of the following shall constitute an event of default (each, an "Event of Default" and collectively, the "Events of Default"):

(a) the Debtors' failure to: (i) use the DIP Loans and DIP Collateral, including but not limited to Cash Collateral, in a manner consistent with the Approved Budget and otherwise comply in any respect with any provision of this Order and the DIP Financing Documents; or (ii) comply with any other covenant or agreement specified in this Order, in each case where such failure shall have continued unremedied for five (5) business days following receipt of written notice by the Debtors from the DIP Agent or the Backstop Parties of such failure;

- (b) (i) filing of an application, motion or other pleading by any Debtor seeking to amend, stay, supplement, vacate, extend or modify in any manner the DIP Lenders', the DIP Agent's or the Prepetition Secured Parties' rights or benefits granted under this Order; or (ii) entry of an order reversing, amending, supplementing, extending, staying, vacating, or otherwise modifying in any manner this Order, in each case, without the prior written consent of the Backstop Parties and the DIP Agent, each in their/its sole discretion;
- (c) the date any provision of this Order for any reason ceases to be valid and binding or any Debtor so asserts in any pleading filed in any court;
- (d) the date (i) any Chapter 11 Case is dismissed or converted to a case under chapter 7 of the Bankruptcy Code or any Debtor files a motion or other pleading seeking the dismissal or conversion of any Chapter 11 Case pursuant to Bankruptcy Code section 1112 or otherwise; or (ii) a trustee, responsible officer, or an examiner (other than a fee examiner) pursuant to Bankruptcy Code section 1104 is appointed or elected, as applicable, in any Chapter 11 Case, any Debtor applies for, consents to, or acquiesces in, any such appointment, or the Court enters an order providing for such appointment, in each case without the prior written consent of the Backstop Parties in their sole discretion;

(e) the Court enters an order or orders granting relief from the automatic stay

to the holder or holders of any security interest to proceed against, including foreclosure (or the

granting of a deed in lieu of foreclosure or the like) on any of the Debtor's assets (other than in

respect of insurance proceeds or with respect to assets having a fair market value of less than

\$1,000,000);

(f) any Debtor files a motion or application for the approval of any

superpriority claim, lien or security interest in the Chapter 11 Cases (other than such claim, lien

or security interest granted or permitted pursuant to this Order and the Approved Liens), which is

pari passu with, or senior to, the DIP Liens, DIP Superpriority Claim, Prepetition Liens,

Adequate Protection Liens, First Lien Superpriority Claim, or the Second Lien Superpriority

Claim, without the prior consent of the DIP Agent, Backstop Parties, the Steering Committee, or

the Administrative Agent, as applicable, each in its/their sole discretion;

(g) other than with respect to the Carve-Out, the Approved Liens and the

claims, liens and security interests granted herein to the DIP Lenders and the DIP Agent, any

Debtor creates or incurs, or the Court enters an order granting, any claim, lien or security interest

that is pari passu with, or senior to, any of the DIP Liens, DIP Superpriority Claim, Prepetition

Liens, Adequate Protection Liens, First Lien Superiority Claim or Second Lien Superpriority

Claim granted under this Order;

(h) except as contemplated by the Sale Motion, unless otherwise agreed to in

writing by the Backstop Parties in their sole discretion, the consummation of a sale or disposition

of any material assets of the Debtors other than in the ordinary course of business;

(i) commencement of any action, including the filing of any pleading, by any

Debtor, or direct or indirect non-debtor affiliate or subsidiary of a Debtor, against any of the DIP

Lenders, DIP Agent, or Prepetition Secured Parties with respect to any of the DIP Obligations, DIP Liens, DIP Superpriority Claim, First Lien Adequate Protection Obligations, First Lien Adequate Protection Liens, Prepetition Obligations or Prepetition Liens, as applicable;

- (j) except as contemplated by the Sale Motion, the Court enters an order approving the sale of all or substantially all of the DIP Collateral that does not provide for the payment in respect thereof to be remitted to the DIP Agent, unless and until the DIP Facility shall have been indefeasibly paid in full in cash, and then to the Administrative Agent consistent with the priorities set forth in this Order and the Prepetition Debt Documents;
- (k) the Debtors' breach of, or failure to perform, any of their agreements, covenants, representations or warranties contained in the Asset Purchase Agreement (unless cured or waived to the extent permitted under and in accordance with the Asset Purchase Agreement) or the Sale Order, without the prior written consent of the Backstop Parties in their sole discretion;
- (l) the occurrence of any termination event under the Asset Purchase Agreement, without the prior written consent of the Backstop Parties in their sole discretion;
- (m) entry of an order reversing, amending, staying, vacating, terminating or otherwise modifying in any manner the Sale Order, without the prior written consent of the Backstop Parties in their sole discretion; and
- (n) any "Event of Default" in the DIP Credit Agreement, unless cured or waived to the extent permitted under the DIP Credit Agreement.
- 13. Remedies Upon an Event of Default (DIP Financing Lenders and DIP Agent).

  The Debtors shall promptly provide notice to counsel to each of the Backstop Parties and the DIP Agent (with a copy to counsel to each of the Creditors' Committee, Retiree Committee,

Administrative Agent, First Lien Trustee, Second Lien Trustee and Bankruptcy Administrator) of the occurrence of any Event of Default, at which time the DIP Obligations, as applicable, shall become due and payable and the Debtors' ability to use Cash Collateral shall terminate. Upon the occurrence of an Event of Default, to the extent not cured or waived as permitted under the DIP Credit Agreement or this Order, as applicable, and following the giving of not less than five (5) business days' written notice (such notice, the "DIP Enforcement Notice" and such date, the "Termination Date") to counsel to the Debtors, counsel to the Creditors' Committee, counsel to the Retiree Committee, and the Bankruptcy Administrator (the "DIP Financing Notice Period"), the DIP Lenders and the DIP Agent (acting at the direction of the Required Lenders) may exercise any remedies available to them under this Order, the DIP Financing Documents and applicable non-bankruptcy law, including, but not limited to, foreclosing upon the DIP Collateral or otherwise enforcing the DIP Obligations and the DIP Claims and Liens on any or all of the DIP Collateral and/or taking any other actions or exercising any other default-related rights and remedies under the DIP Financing Documents, this Order or applicable law to effect the repayment and satisfaction of the DIP Obligations. The only permissible basis for the Debtors, Creditors' Committee, Bankruptcy Administrator or any other party to contest, challenge or object to the DIP Enforcement Notice shall be solely with respect to the validity of the Event(s) of Default giving rise to such DIP Enforcement Notice (i.e., whether such Event(s) of Default validly occurred or have not been cured or waived in accordance with this Order or the DIP Credit Agreement). The automatic stay pursuant to Bankruptcy Code section 362 shall be automatically terminated on the Termination Date, without further notice or order of the Bankruptcy Court, unless the DIP Agent (at the direction of the Required Lenders) elects otherwise in a written notice to the Debtors, and the DIP Agent (at the direction of the Required Lenders) shall be permitted to exercise all rights and remedies, including with respect to the DIP Collateral, set forth in the Order, the DIP Financing Documents, and as otherwise available at law without further order or application or motion to the Court, and without restriction or restraint by any stay under Bankruptcy Code sections 362 or 105 or otherwise.

- upon presentment, but in no event later than 10 days of presentment, of an applicable invoice to the Debtors (with a copy of such invoice to be presented contemporaneously to both the Bankruptcy Administrator and counsel for each of the Creditors' Committee and Retiree Committee), all reasonable, actual, and documented (in customary detail, redacted for privilege and work product) fees, costs and expenses of Akin Gump Strauss Hauer & Feld LLP ("Akin Gump"), as lead counsel, Burr Forman LLP ("Burr Forman"), as Alabama counsel, and Lazard Frères & Co. LLC ("Lazard" and together with Akin Gump and Burr Forman, the "Backstop Parties' Advisors"), as financial advisor, to the Backstop Parties. In addition, the Debtors shall promptly reimburse each Backstop Party in cash for all reasonable and documented out-of-pocket costs and expenses (without limiting the Debtors' obligations pursuant to the previous sentence, which out-of-pocket costs and expenses should not include any advisor or professional fees for such individual Backstop Party) incurred by such party in connection with the Chapter 11 Cases.
- 15. Entitlement to Adequate Protection. The Prepetition Secured Parties are entitled to adequate protection of their respective interests in the Prepetition Collateral (including, but not limited to, the Cash Collateral) on which the Prepetition Secured Parties hold perfected security interests as of the Petition Date in an amount equal to the aggregate postpetition diminution in value of the Prepetition Collateral, including any Cash Collateral, from and after the Petition

Date (such diminution in value, the "<u>Diminution in Value</u>"), including, but not limited to, to the extent such diminution results from the sale, lease or use by the Debtors of the Prepetition Collateral, including any Cash Collateral, the subordination of the Prepetition Liens to the Carve-Out and the DIP Claims and Liens, the Debtors' incurrence of the DIP Loans, or the imposition of the automatic stay pursuant to Bankruptcy Code section 362 (such adequate protection, as set forth in paragraph 16 below, the "<u>Adequate Protection Obligations</u>"). Nothing contained herein shall be deemed a finding by the Court or an acknowledgement by the Prepetition Secured Parties or the Steering Committee, respectively, that the adequate protection granted herein does in fact adequately protect the Prepetition Secured Parties against the Diminution in Value.

- 16. Adequate Protection Granted to Prepetition Secured Parties.
- (a) <u>First Lien Superpriority Claim and Adequate Protection Liens</u>. The First Lien Secured Parties are hereby granted the following claims, liens and security interests:
  - (i) First Lien Superpriority Claim. The Adequate Protection Obligations due to the First Lien Secured Parties (the "First Lien Adequate Protection Obligations") shall constitute allowed joint and several superpriority claims against each of the Debtors as provided in Bankruptcy Code section 507(b) (collectively, the "First Lien" Superpriority Claim"), with priority over any and all administrative expenses and all other claims asserted against the Debtors or their estates, at any time existing or arising, of any kind or nature whatsoever, including, but not limited to, the administrative expenses and other claims of the kinds specified in or ordered pursuant to Bankruptcy Code sections 105, 326, 327, 328, 330, 331, 365, 503, 506, 507(a), 507(b), 546, 552, 726, 1113 and 1114, and any other provision of the Bankruptcy Code, whether or not such expenses or claims may become secured by a judgment lien or other non-consensual lien, levy or attachment, subject and subordinate in all respects to the Carve-Out and the DIP Superpriority Claim. Except to the extent expressly set forth in this Order, the First Lien Secured Parties shall not receive or retain any payments, property or other amounts in respect to the First Lien Superpriority Claim unless and until all DIP Obligations shall have been indefeasibly paid in full in cash.

- (ii) First Lien Adequate Protection Liens. As security for the First Lien Adequate Protection Obligations, effective as of the Petition Date and perfected without the necessity of the execution by the Debtors (or recordation or other filing) of security agreements, control agreements, pledge agreements, financing statements, mortgages or other similar documents, or the possession or control by the First Lien Secured Parties, the First Lien Secured Parties are hereby granted valid, binding, continuing, enforceable, fully-perfected, non-avoidable additional and replacement liens on, and security interests in, the DIP Collateral (the "First Lien Adequate Protection Liens"), subject and subordinate only to the (w) Carve-Out, (x) DIP Liens, (y) Permitted Priority Liens (if any) and (z) Approved Liens. The First Lien Adequate Protection Liens shall not be subject or subordinate to, or pari passu with, any lien or security interest that is avoided and preserved for the benefit of the Debtors and their estates under Bankruptcy Code section 551 but shall be subject to the DIP Liens.
- (b) <u>Second Lien Secured Parties Superpriority Claim and Adequate Protection</u>

  <u>Liens.</u> The Second Lien Trustee and the Second Lien Noteholders (collectively, the "<u>Second Lien Secured Parties</u>") are hereby granted the following claims, liens and security interests:
  - (i) Second Lien Superpriority Claims. The Adequate Protection Obligations due to the Second Lien Secured Parties (the "Second Lien Adequate Protection Obligations") shall constitute joint and several superpriority claims against the Debtors as provided in Bankruptcy Code section 507(b) (the "Second Lien Superpriority Claim" and together with the First Lien Superpriority Claim, the "Prepetition Secured Parties' Superpriority Claims"), with priority over any and all administrative expenses and claims asserted against the Debtors or their estates, at any time existing or arising, of any kind or nature whatsoever, including, but not limited to, the administrative expenses of the kinds specified in or ordered pursuant to Bankruptcy Code sections 105, 326, 327, 328, 330, 331, 365, 503, 506, 507(a), 507(b), 546, 552, 726, 1113 and 1114, and any other provision of the Bankruptcy Code, whether or not such expenses or claims may become secured by a judgment lien or other non-consensual lien, levy or attachment, subject and subordinate in all respects only to the (v) Carve-Out, (w) DIP Superpriority Claim, (x) First Lien Superpriority Claim, (y) Permitted Priority Liens (if any) and (z) Approved Liens. Except to the extent expressly set forth in this Order, the Second Lien Secured Parties shall not receive or retain any payments, property or other amounts in respect to the Second Lien Superpriority Claim unless and until all DIP Obligations and the First Lien Obligations

- shall have been indefeasibly paid in full in cash in accordance with the priorities set forth herein.
- (iii) Second Lien Adequate Protection Liens. As security for the Second Lien Adequate Protection Obligations, effective as of the Petition Date and perfected without the necessity of the execution by the Debtors (or recordation or other filing) of security agreements, control agreements, pledge agreements, financing statements, mortgages or other similar documents, or the possession or control by the Second Lien Secured Parties, security interests in, and liens on, the DIP Collateral are hereby granted to the Second Lien Secured Parties, subject and subordinate only to the (u) Carve-Out, (v) DIP Liens, (w) First Lien Adequate Protection Liens, (x) liens and security interests securing the First Lien Obligations, (y) the Permitted Priority Liens (if any) and (z) Approved Liens, and subject further to the Intercreditor Agreement (all such liens and security interests, the "Second Lien Adequate Protection Liens," and collectively with the First Lien Adequate Protection Liens, the "Adequate Protection Liens"). The Second Lien Adequate Protection Liens shall not be subject or subordinate to, or pari passu with, any lien or security interest that is avoided and preserved for the benefit of the Debtors and their estates under Bankruptcy Code section 551 but shall be subject to the DIP Liens, First Lien Adequate Protection Liens and Prepetition First Priority Liens.

- (c) <u>Additional Adequate Protection Granted to Prepetition Secured Parties.</u>
  - (i) Adequate Protection Payments: The adequate protection payments specified in paragraph 11(a) of the Amended Final Cash Collateral Order that have accrued but remain unpaid as of the date hereof shall be deferred and shall not be paid unless and until the DIP Obligations have been indefeasibly paid in full in cash. No additional adequate protection payments shall accrue under paragraph 11(a) of the Amended Final Cash Collateral Order from and after the date hereof. For the avoidance of doubt, nothing contained herein shall preclude or impact the Administrative Agent's (on behalf of the First Lien Lenders) and the First Lien Trustee's (on behalf of the First Lien Noteholders) ability or right to credit bid the First Lien Adequate Protection Obligations for the Acquired Assets (as defined in the Asset Purchase Agreement) pursuant to the Asset Purchase Agreement, the Amended Final Cash Collateral Order, the Sale Order and the Order Granting Steering Committee's Motion to Determine the Value of the First Lien Secured Parties' Adequate Protection Claims as a Result of the Diminution in Value of the First Lien Secured Parties' Collateral [Docket No. 1450]; provided that the DIP Obligations are indefeasibly paid in full in cash at Closing.
  - (ii) Agent/Indenture Trustee Fees and Expenses: The Debtors shall promptly pay, in cash, (x) all Letter of Credit Fees and Facing Fees (each as defined in the Credit Agreement), and any annual administrative agent fees and other fees set forth in Section 4.01(b) through (e) of the Credit Agreement on the respective dates for the payment of all such fees as provided in the Credit Agreement, at the applicable non-default rate provided for in the First Lien Credit Documents with respect to such fees and (y) upon presentment, but in no event later than 10 days of presentment, of an applicable invoice to the Debtors (with a copy of such invoice to be presented contemporaneously to the Bankruptcy Administrator, counsel to the Backstop Parties, and counsel for each of the Creditors' Committee and Retiree Committee), all reasonable, actual, and documented (in customary detail, redacted for privilege and work product) fees, costs and expenses incurred by each of the Administrative Agent and the First Lien Trustee, including, but not limited to, the fees, costs and expenses of one lead counsel, one local counsel (if necessary) and, if needed, one Canadian counsel for each of the Administrative Agent and the First Lien Trustee, in each case in accordance with the applicable engagement letters (if any) and the Prepetition Debt Documents and without further order of, or application to, the Court or notice to any party other than as provided in this paragraph 16(c)(ii).

- (iii) Budget Compliance/Additional Reporting. With respect to the Prepetition Secured Parties, the Debtors shall comply with the budget and reporting requirements set forth in this Order and the DIP Financing Documents; provided that, such compliance and rights set forth in this paragraph 16(c)(iii) shall only extend to the Prepetition Secured Parties to the extent that the Debtors have indefeasibly paid and satisfied in full in cash the DIP Obligations but continue to use Cash Collateral.
- (iv) Steering Committee Fees and Expenses. The Debtors shall promptly pay in cash upon presentment, but in no event later than 10 days of presentment, of an applicable invoice to the Debtors (with a copy of such invoice to be presented contemporaneously to both the Bankruptcy Administrator and counsel for both the Creditors' Committee and Retiree Committee), all reasonable, actual, and documented (in customary detail, redacted for privilege and work product) fees, costs and expenses of (i) Akin Gump, as lead counsel, Burr Forman, as Alabama counsel, Cassels Brock & Blackwell LLP, as Canadian counsel, Jackson Kelly LLP, as West Virginia counsel, Lazard, as financial advisor (including the Restructuring Fee as defined in Lazard's engagement letter), Stephen Douglas Williams Consulting LLC, as consultant, Golder Associates Inc., as mining engineer, Duff & Phelps LLC, as consultant, NRI Management Group LLC, as consultant, AME Consulting Limited, as consultant, JD Consulting LLC, as consultant, and Henry Consulting, LLC, as consultant, to the Steering Committee (collectively, the "Steering Committee Advisors") and (ii) any other consultants or advisors retained by the Steering Committee (and not by individual Steering Committee members) (the parties described in this paragraph 16(c)(iv), collectively, the "Consultants"); provided that the Steering Committee shall provide notice to the Debtors prior to retaining any such Consultants, in each case, in accordance with engagement letters (if any) of such consultant or advisor, and in each case, without further order of, or application to, the Court or notice to any party other than as provided in this paragraph 16(c)(iv); provided that no success fees shall be payable to any Steering Committee Advisor (except for the Restructuring Fee payable to Lazard) or Consultant. In addition, the Debtors shall promptly reimburse each Steering Committee member in cash for all reasonable and documented out-of-pocket costs and expenses (without limiting the Debtors' obligations pursuant to the previous sentence, which out-of-pocket costs and expenses should not include any advisor or professional fees for such individual Steering Committee member) incurred by such member in connection with these cases.

- (v) Access to Records/Financial Reporting. In addition to, and without limiting, whatever rights of access the Prepetition Secured Parties have under the Prepetition Debt Documents, upon reasonable notice, at reasonable times and subject to appropriate confidentiality protections, the Debtors shall permit representatives and agents of the Backstop Parties, the Creditors' Committee, the Retiree Committee, and the Administrative Agent (i) to have access to and inspect the Debtors' properties, subject to reasonable safety precautions, (ii) to examine the Debtors' books and records, and (iii) to discuss the Debtors' affairs, finances and condition with the Debtors' officers and financial advisors. In addition, the Debtors shall provide to the Backstop Parties' Advisors, and the advisors to each of the Administrative Agent, the Creditors' Committee, the Retiree Committee, and the UMWA, on a monthly basis, reports setting forth (i) substantive mine-by-mine details of the Debtors' operating performance, including its non-debtor subsidiaries and (ii) a detailed comparison, including commentary, of the month's actual operating performance against the projections, substantially in form and substance consistent with the Company's historical monthly reporting to the Board of Directors, as modified to include summary mine-level operating and financial data.
- (vi) **Right to Seek Additional Adequate Protection**. This Order is without prejudice to, and does not constitute a waiver of, expressly or implicitly, the rights of the Prepetition Secured Parties to request additional forms of adequate protection at any time, subject to the consent of the Backstop Parties in their sole discretion, or any party-in-interest's right to object thereto. Any such request must be consistent with the Intercreditor Agreement.
- avoidance of doubt, (i) to the extent ACE American Insurance Company and/or any of its affiliates (collectively, and together with each of their successors "Chubb") had valid and perfected liens and/or security interests on any of the DIP Collateral as of the Petition Date, which liens and/or security interests were senior to the liens and/or security interests of each of the Prepetition Secured Parties, such liens and/or security interests shall be senior to any liens and/or security interests granted pursuant to this Order, (ii) the DIP Agent, DIP Lenders and Prepetition Secured Parties do not have liens and/or security interests on any letter(s) of credit

for which Chubb is the beneficiary or any proceeds thereof and (iii) this Order does not grant the Debtors any right to use any property (or the proceeds thereof) held by Chubb as collateral to secure obligations under insurance policies and related agreements.

18. Release. Each of the Debtors and the Debtors' estates, on its own behalf and on behalf of its past, present and future predecessors, successors, heirs, subsidiaries, and assigns (collectively, the "Releasors") shall, to the maximum extent permitted by applicable law, unconditionally, irrevocably, fully and forever release, remise, acquit, relinquish, irrevocably waive and discharge each of the DIP Lenders, the DIP Agent, the Backstop Parties, the Steering Committee, the Prepetition Secured Parties and each of their respective former, current, or future officers, employees, directors, agents, representatives, owners, members, partners, financial advisors, legal advisors, shareholders, managers, consultants, accountants, attorneys, affiliates, and predecessors in interest (collectively, the "Releasees"), of and from any and all claims, demands, liabilities, responsibilities, disputes, remedies, causes of action, indebtedness and obligations, rights, assertions, allegations, actions, suits, controversies, proceedings, losses, damages, injuries, attorneys' fees, costs, expenses, or judgments of every type, whether known, unknown, asserted, unasserted, suspected, unsuspected, accrued, unaccrued, fixed, contingent, pending, or threatened including, but not limited to, all legal and equitable theories of recovery, arising under common law, statute or regulation or by contract, of every nature and description that existed as of the Effective Date relating to any of the Prepetition Debt Documents, the DIP Financing Documents, or the transactions contemplated under such documents, or the Chapter 11 Cases, as applicable, including, but not limited to, (i) any so-called "lender liability," equitable subordination, equitable disallowance or recharacterization claims or defenses, (ii) any and all claims and causes of action arising under the Bankruptcy Code, and (iii) any and all claims and causes of action regarding the validity, priority, perfection or nonavoidability of the liens or claims of the DIP Lenders, the DIP Agent and the Prepetition Secured Parties; <u>provided</u>, <u>however</u>, that solely with respect to the Prepetition Secured Parties, the foregoing releases are subject to the rights of the Creditors' Committee in paragraph 21 below.

## 19. *Perfection of DIP Liens and Adequate Protection Liens.*

(a) The DIP Agent, the Administrative Agent, the Credit Agreement Collateral Agent, the First Lien Trustee, the 1L Notes Collateral Agent, the Second Lien Trustee and the 2L Notes Collateral Agent, are each hereby authorized, but not required, to file or record financing statements, intellectual property filings, mortgages, notices of lien or similar instruments in any jurisdiction in order to validate and perfect the liens and security interests granted to it hereunder. Whether or not the DIP Agent, the Administrative Agent, the Credit Agreement Collateral Agent, the First Lien Trustee, the 1L Notes Collateral Agent, the Second Lien Trustee or the 2L Notes Collateral Agent, each in its respective sole discretion, chooses to file such financing statements, intellectual property filings, mortgages, notices of lien or similar instruments, such liens and security interests shall be deemed valid, perfected, allowed, enforceable, non-avoidable and not subject to challenge, dispute, subordination, contest, attack, objection, recoupment, defense, setoff, counterclaim, avoidance, recharacterization, reclassification, reduction, disallowance, recovery, disgorgement, attachment, "claim" (as defined in the Bankruptcy Code), impairment, subordination (whether equitable, contractual or otherwise) or other challenge of any kind pursuant to the Bankruptcy Code or applicable nonbankruptcy law as of the Petition Date. If the DIP Agent, the Administrative Agent, the Credit Agreement Collateral Agent, the First Lien Trustee, the 1L Notes Collateral Agent, the Second Lien Trustee or the 2L Notes Collateral Agent, as applicable, determines to file any

financing statements, notices of liens or similar instruments, the Debtors will cooperate and assist in any such filings as reasonably requested by the DIP Agent, the Administrative Agent, the Credit Agreement Collateral Agent, the First Lien Trustee, the 1L Notes Collateral Agent, the Second Lien Trustee or the 2L Notes Collateral Agent, as applicable, and the automatic stay shall be modified to allow such filings.

- (b) The DIP Agent, the Administrative Agent, the Credit Agreement Collateral Agent, the First Lien Trustee, the 1L Notes Collateral Agent, the Second Lien Trustee or the 2L Notes Collateral Agent, may, each in its respective discretion, cause a certified copy of this Order to be filed with or recorded in filing or recording offices in addition to or in lieu of such financing statements, mortgages, notices of lien or similar instruments, and all filing offices are hereby authorized to accept such certified copy of this Order for filing and recording.
- (c) The Debtors shall execute and deliver to the DIP Agent, the Administrative Agent, the Credit Agreement Collateral Agent, the First Lien Trustee, the 1L Notes Collateral Agent, the Second Lien Trustee or the 2L Notes Collateral Agent all such agreements, financing statements, instruments and other documents as each such party may reasonably request to evidence, confirm, validate or perfect the DIP Liens and the Adequate Protection Liens, as applicable.
- (d) Notwithstanding anything to the contrary in the Motion or this Order, for purposes of this Order, in no event shall the DIP Collateral or Prepetition Collateral include or the DIP Liens or the Adequate Protection Liens granted under this Order attach to, any lease, license, permit, contract, or agreement (including any operating and joint venture agreements) or other property right, to which any Debtor is a party, or any of such relevant Debtor's rights or interests thereunder, if and for so long as the grant of such security interest would constitute or

result in: (i) the abandonment, invalidation, unenforceability, or other impairment of any right, title, or interest of any Debtor therein, or (ii) a breach or termination pursuant to the terms of, or a default under, any such lease, license, contract, agreement, or other property right pursuant to any provision thereof, unless, in the case of each of clauses (i) and (ii), the applicable provision is rendered ineffective by applicable non-bankruptcy law or the Bankruptcy Code (such leases, licenses, contracts or agreements, or other property rights are collectively referred to as the "Specified Contracts"); provided that the foregoing shall not preclude any counterparty to a Specified Contract from an opportunity to be heard in this Court on notice with respect to whether applicable non-bankruptcy law or the Bankruptcy Code renders such provision ineffective. Notwithstanding the foregoing, the DIP Liens and the Adequate Protection Liens shall in all events attach to all proceeds, products, offspring, or profits from all sales, transfers, dispositions, or monetizations of any and all Specified Contracts.

## 20. Preservation of Rights Granted Under this Order.

(a) Notwithstanding any order dismissing any of these Chapter 11 Cases under Bankruptcy Code section 1112 or otherwise entered at any time, (i) the DIP Claims and Liens, the Prepetition Secured Parties' Superpriority Claims, the Adequate Protection Liens, the other administrative claims granted pursuant to this Order, and the Carve-Out shall continue in full force and effect and shall maintain their priorities as provided in this Order until all DIP Obligations, Adequate Protection Obligations and the Carve Out shall have been indefeasibly paid and satisfied in full in cash, as applicable (and such DIP Claims and Liens, Prepetition Secured Parties' Superpriority Claims, other administrative claims granted pursuant to this Order, Carve Out and Adequate Protection Liens shall, notwithstanding such dismissal, remain binding on all parties in interest), and (ii) this Court shall retain jurisdiction, notwithstanding

such dismissal, for the purposes of enforcing the claims, liens and security interests referred to in clause (i) above.

(b) If any or all of the provisions of this Order are hereafter reversed, modified, vacated or stayed, such reversal, stay, modification or vacatur shall not affect: (i) the validity, priority or enforceability of any DIP Claims and Liens, DIP Obligations, Prepetition Secured Parties' Superpriority Claims and Adequate Protection Obligations incurred prior to the date of the entry of an order granting such reversal, stay, modification or vacatur (the "Reversal Order"); or (ii) the validity, priority or enforceability of the DIP Liens and the Adequate Protection Liens. Notwithstanding any such reversal, stay, modification or vacatur, any borrowings under the DIP Facility, use of the Prepetition Collateral (including the Cash Collateral) or DIP Collateral, or the DIP Obligations or the Adequate Protection Obligations incurred by the Debtors hereunder prior to the date of the entry of the Reversal Order shall be governed in all respects by the original provisions of this Order, and (x) the DIP Lenders, the DIP Agent, and the First Lien Secured Parties shall be entitled to all of the rights, remedies, privileges and benefits granted in Bankruptcy Code sections 363(m) and 364(e) with respect to all uses of the Prepetition Collateral and DIP Collateral, all DIP Obligations and First Lien Adequate Protection Obligations for periods prior to the date of the entry of the Reversal Order and (y) subject to the Intercreditor Agreement, the Second Lien Trustee and the Second Lien Noteholders shall be entitled to all of the rights, remedies, privileges and benefits granted in Bankruptcy Code section 363(m) with respect to all uses of the Prepetition Collateral (including the Cash Collateral) (other than Collateral constituting setoff rights of the First Lien Secured Parties) and all Second Lien Adequate Protection Obligations for periods prior to the date of the entry of the Reversal Order.

Except as expressly provided in this Order or in the DIP Financing (c) Documents, the DIP Claims and Liens, the Adequate Protection Liens and Prepetition Secured Parties' Superpriority Claims and all other rights, claims, security interests and remedies of the DIP Lenders and the Prepetition Secured Parties granted by the provisions of this Order and the DIP Financing Documents shall survive, and shall not be modified, impaired or discharged by the entry of an order (i) converting any of these Chapter 11 Cases to cases under chapter 7 of the Bankruptcy Code, dismissing any of these Chapter 11 Cases or by any other act or omission or (ii) confirming a plan of reorganization in any of the Chapter 11 Cases, and, pursuant to section 1141(d)(4) of the Bankruptcy Code, the Debtors have waived any discharge as to any remaining DIP Claims and Liens, Adequate Protection Liens or the Prepetition Secured Parties' Superpriority Claims. The terms and provisions of this Order shall continue in these Chapter 11 Cases, in any successor cases, including any chapter 7 cases, if these Chapter 11 Cases cease to be jointly administered, or in any successor or superseding chapter 7 cases under the Bankruptcy Code, and the DIP Claims and Liens, Adequate Protection Liens, the administrative claims granted pursuant to this Order, and all other rights, claims, security interests and remedies of the DIP Lenders, DIP Agent and Prepetition Secured Parties granted by the provisions of this Order shall continue in full force and effect as provided herein.

21. Effect of Stipulations on Third Parties. The stipulations, releases and admissions contained in this Order, including in paragraph 5 hereof, shall be binding upon the Debtors and any successor thereto in all circumstances other than as expressly set forth below. The stipulations, releases and admissions contained in this Order, including in paragraph 5 hereof, shall be binding upon all other parties in interest, including the Creditors' Committee or any chapter 7 or chapter 11 trustee appointed or elected for any of the Debtors (a "Trustee"), unless

solely with respect to the Prepetition Secured Parties (a) the Creditors' Committee has duly filed an adversary proceeding challenging in whole or part the validity, enforceability, priority or extent of the Prepetition Obligations or the liens on the Prepetition Collateral securing the Prepetition Obligations held by or on behalf of the Prepetition Secured Parties or otherwise asserting or prosecuting any Avoidance Actions, recharacterization, subordination, "lender liability", or any other claims, counterclaims or causes of action, objections, contests or defenses (collectively, the "Claims and Defenses") against the Prepetition Secured Parties in connection with any matter related to the Prepetition Obligations, or the Prepetition Collateral or the Prepetition Liens by no later than fourteen days (14) from the date the Creditors' Committee receives written notice of the termination of the Asset Purchase Agreement (the "Challenge Deadline") or as otherwise provided in the Global Settlement (as defined below); provided, however, pursuant to that certain Settlement Term Sheet (as approved in the Order Approving Global Settlement Among the Debtors, Official Committee of Unsecured Creditors, Steering Committee and Stalking Horse Purchaser Pursuant to Fed. R. Bank. P. 9019 [Docket No. 1456], the "Global Settlement"), the Creditors' Committee has agreed to waive its right to, and shall not, assert any Claims and Defenses, Challenges (as defined herein) or any other claims against the Prepetition Secured Parties upon the Closing of the transactions contemplated under the Asset Purchase Agreement (the "Condition to Effectiveness of Global Settlement"), and (b) an order is entered by a court of competent jurisdiction and becomes final and non-appealable in favor of the Creditors' Committee sustaining any such Challenge or claim in any such duly filed adversary proceeding. If the Condition to Effectiveness of Global Settlement does not occur, and the Creditors' Committee fails to timely file such an adversary proceeding by the Challenge Deadline or (b) the Condition to Effectiveness of Global Settlement occurs: (x) the Prepetition Obligations shall constitute allowed claims, not subject to counterclaim, setoff, subordination, recharacterization, defense, avoidance, contest, attack, objection, recoupment, reclassification, reduction, disallowance, recovery, disgorgement, attachment, "claim" (as defined in the Bankruptcy Code), impairment, subordination (whether equitable, contractual or otherwise) or other challenge of any kind pursuant to the Bankruptcy Code or applicable nonbankruptcy law, for all purposes in these Chapter 11 Cases and any subsequent chapter 7 cases; and (y) the Prepetition Obligations, the Administrative Agent's, the First Lien Trustee's and the Second Lien Trustee's respective Prepetition Liens on the Prepetition Collateral and the respective Prepetition Secured Parties in such capacity shall not be subject to any other or further challenge or claim and any party in interest shall be forever enjoined and barred from seeking to exercise the rights of the Debtors' estates or taking any such action, including any successor thereto (including any estate representative or a Trustee, whether such Trustee is appointed or elected prior to or following the expiration of the Challenge Period). If the Condition to Effectiveness of Global Settlement does not occur and the Creditors' Committee does file such an adversary proceeding by the Challenge Deadline, the stipulations and admissions contained in this Order, including in paragraph 5 hereof, shall nonetheless remain binding and preclusive (as provided in the second sentence of this paragraph) on the Creditors' Committee and any other Person (as defined in the Credit Agreement), including any Trustee, except as to any such findings and admissions that were expressly and successfully challenged in such adversary proceeding. Nothing in this Order vests or confers on any Person, including the Creditors' Committee or Trustee, standing or authority to pursue any cause of action belonging to the Debtors or their estates; provided, however, that the Creditors' Committee shall be automatically vested with standing to initiate a Challenge prior to the expiration of the Challenge Period without any need to first seek a Court order granting such standing. For the avoidance of doubt, the Challenge Period (as defined in the Amended Final Cash Collateral Order) has expired with respect to all other parties in interest, other than the Creditors' Committee, and no such party in interest has commenced a Challenge or filed an adversary proceeding prior to the expiration of such period.

- 22. 506(c) Waiver. Except to the extent of the Carve-Out, no costs or expenses of administration of the Chapter 11 Cases or any successor cases, including any chapter 7 cases, that may result therefrom, including liquidation in bankruptcy or other proceedings under the Bankruptcy Code, shall be charged against or recovered from (i) the DIP Agent or any DIP Lender, any of the DIP Obligations, any Prepetition Secured Party, or any of the Prepetition Obligations, (ii) any of the DIP Lenders', DIP Agent's or the Prepetition Secured Parties' respective claims, or (iii) the DIP Collateral or Prepetition Collateral, as applicable, pursuant to Bankruptcy Code sections 105(a) or 506(c), or otherwise, without the prior written consent of the affected DIP Agent, Administrative Agent, Steering Committee, First Lien Trustee or Second Lien Trustee and the Backstop Parties, as applicable, each in its/their sole discretion, and no such consent shall be implied from any other action, inaction, or acquiescence by any of the DIP Lenders, the DIP Agent, the Prepetition Secured Parties, the Steering Committee, the Backstop Parties, as applicable, or their respective representatives.
- 23. Section 552(b). Each of the DIP Lenders, the DIP Agent, and the Prepetition Secured Parties shall be entitled to all of the rights and benefits of Bankruptcy Code section 552(b). The "equities of the case" exception under Bankruptcy Code section 552(b) shall not apply to the DIP Lenders, the DIP Agent and Prepetition Secured Parties with respect to (i) proceeds, products or profits of any of the Prepetition Collateral, including Cash Collateral, or

DIP Collateral, as applicable or (ii) the extension of the Adequate Protection Liens to cover proceeds of the Prepetition Collateral.

24. Credit Bidding. (a) The DIP Agent (acting at the direction of the Required Lenders) shall have the unqualified right to credit bid up to the full amount of any remaining DIP Obligations in the sale of any of the Debtors' assets, including (i) pursuant to Bankruptcy Code section 363, (ii) a plan of reorganization or a plan of liquidation under Bankruptcy Code section 1129, or (iii) a sale or disposition by a chapter 7 trustee for any Debtor under Bankruptcy Code section 725, and (b) subject to the indefeasible payment in full in cash of the DIP Obligations, the Administrative Agent (on behalf of the First Lien Lenders), the First Lien Trustee (on behalf of the First Lien Noteholders) and the Second Lien Trustee (on behalf of the Second Lien Noteholders) (but only if any such credit bid provides, to the extent set forth in the Intercreditor Agreement, for the payment in full and in cash of all Prepetition Obligations owed to the First Lien Secured Parties and any amounts due and owing to the First Lien Secured Parties under this Order, and provides for the cash collateralization of any letters of credit in accordance with the First Lien Credit Documents and this Order), as applicable, shall have the right to credit bid (X) up to the full amount of the remaining Prepetition Obligations under the First Lien Credit Documents, First Lien Indenture Documents and the Second Lien Indenture Documents, respectively and (Y) the Prepetition Secured Parties' Superpriority Claims, and any unpaid amounts due and owing under paragraph 16 hereof, as applicable, in the sale of any of the Debtors' assets, including, but not limited to, (i) pursuant to Bankruptcy Code section 363, (ii) a plan of reorganization or a plan of liquidation under Bankruptcy Code section 1129, or (iii) a sale or disposition by a chapter 7 trustee for any Debtor under Bankruptcy Code section 725.

- 25. *No Marshaling*. None of the DIP Collateral, the Prepetition Collateral, the DIP Lenders, the DIP Agent or the Prepetition Secured Parties shall be subject to the equitable

doctrine of "marshaling" or any other similar doctrine.

Management Motion.

- 26. Information and Other Covenants. The Debtors shall comply with the reporting requirements set forth in the DIP Credit Agreement and the Prepetition Debt Documents, including the reporting requirements set forth in Section 9.01 (except for clauses (d) and (e)) of the Credit Agreement, as applicable, together with such additional information as the DIP Agent, the Administrative Agent or the First Lien Trustee may reasonably request from time to time. The Debtors shall maintain their cash management arrangements in a manner consistent in all material respects with that described in The Debtors' Motion for an Order (A) (I) Approving Continued Use of the Debtors' Existing Cash Management System; (II) Authorizing Use of Existing Bank Accounts and Checks; (III) Waiving the Requirements of 11 U.S.C. 345(b); (IV) Granting Administrative Expense Status to Certain Postpetition Intercompany Claims; and (V) Authorizing the Continuation of Certain Intercompany Transactions; and (B) Granting Related Relief (the "Cash Management Motion") [Docket No. 38], and any orders approving the Cash
- 27. Restrictions on Transfer of DIP Collateral or Prepetition Collateral to Non-Debtor Affiliates. Other than as permitted under section 10.05(1) of the DIP Credit Agreement, the Debtors shall not transfer or use any DIP Collateral or Prepetition Collateral, including Cash Collateral, to or for the benefit of any direct or indirect foreign or non-debtor affiliate or subsidiary of the Debtors.
- 28. *Indemnification and Expenses*. Each of the Debtors, jointly and severally, shall indemnify and hold harmless the DIP Agent, the Backstop Parties, the DIP Lenders, the First

Lien Secured Parties, their respective affiliates, successors and assigns and the officers, directors, employees, agents, advisors, controlling persons and members of each of the foregoing (each, an "Indemnified Person") from and against all costs, expenses (including reasonable and documented fees, disbursements and other charges of outside counsel but subject to the limitations set forth below) and liabilities of such Indemnified Person arising out of or relating to any investigation, claim or any litigation or other proceeding (regardless of whether such Indemnified Person is a party thereto and regardless of whether such matter is initiated by a third party or by the Borrower or any of its affiliates) that relates to the DIP Financing Documents or this Order or the transactions contemplated thereby and hereby or the exercise of any of their rights provided in the DIP Financing Documents or this Order; provided that, no Indemnified Person will be indemnified for any cost, expense or liability to the extent determined in the final, non-appealable judgment of a court of competent jurisdiction to have resulted solely from its gross negligence or willful misconduct. No Indemnified Person shall have any liability (whether direct or indirect, in contract, tort or otherwise) to the Debtors or any of their subsidiaries or any shareholders or creditors of the foregoing for or in connection with the transactions contemplated hereby, except to the extent such liability is found in a final non-appealable judgment by a court of competent jurisdiction to have resulted solely from such Indemnified Person's gross negligence, willful misconduct or material breach of its obligations under the DIP Financing Documents or this Order. In no event, however, shall any Indemnified Person be liable on any theory of liability for any special, indirect, consequential or punitive damages. In addition, whether accrued on, prior to, or after the Effective Date, and whether or not the Closing occurs, but subject to the limitations in Section 14.01 of the DIP Credit Agreement, (a) all out-of-pocket expenses (including, but not limited to, reasonable and documented fees, disbursements and

other charges of outside counsel and financial advisors) of the DIP Agent and the Backstop Parties (but excluding the fees and expenses of any outside counsel or financial advisor retained by any individual Backstop Party), in connection with the Chapter 11 Cases or any other insolvency or bankruptcy proceedings of the Debtors, the DIP Facility and the transactions contemplated thereby and hereby shall be paid by the Debtors from time to time (b) all out-ofpocket expenses (including, but not limited to, fees, disbursements and other charges of outside counsel and financial advisors) of the DIP Agent and the DIP Lenders for enforcement costs and documentary taxes associated with the DIP Facility and the transactions contemplated thereby and hereby shall be paid by the Debtors, and (c) all fees of the DIP Agent charged in connection with any "seasoning" of the DIP Facility shall be paid by the Debtors. Notwithstanding the foregoing, in no event shall (i) the DIP Agent and (ii) the DIP Lenders, in each case, be entitled to the reimbursement of costs and expenses of more than one primary counsel, one local Alabama counsel and other special counsel and advisors, as needed. Nothing herein is meant to limit the scope of any indemnity provided for the benefit of the DIP Agent, the Backstop Parties or the DIP Lenders in the DIP Financing Documents, which indemnity is hereby approved and authorized. For the avoidance of doubt, this paragraph 28 does not limit or otherwise affect any indemnification rights or obligations in respect of the Prepetition Secured Parties under the Prepetition Debt Documents.

29. Limitation on Use of the DIP Loans, the DIP Collateral, and the Prepetition Collateral (including the Cash Collateral). The Debtors shall use the DIP Loans, the DIP Collateral, and the Prepetition Collateral (including the Cash Collateral), solely as provided in this Order, the Approved Budget and the DIP Financing Documents. Notwithstanding anything herein or in any other order of this Court to the contrary, the Debtors shall not be authorized to

use the DIP Loans, the DIP Collateral, or the Prepetition Collateral, including the Cash Collateral, to pay fees or expenses (x) in excess of \$175,000 per month for each of the Creditors' Committee and the Retiree Committee (the "Committee Monthly Cap") on account of all Professional Persons retained by each such committee in the aggregate; provided that any unused amounts may be carried forward to a subsequent month, (y) in excess of \$100,000 (the "Investigation Budget"), which Investigation Budget replaces and does not supplement or replenish the Investigation Budget provided under the Amended Final Cash Collateral Order, for the Creditors' Committee to investigate (but not prepare, initiate or prosecute) Claims and Defenses against the Prepetition Secured Parties before the Challenge Deadline, or (z) to initiate or prosecute proceedings or actions (the "Challenge") on account of any Claims and Defenses against the Prepetition Secured Parties. For the avoidance of doubt and notwithstanding any other provision of this Order, other than the Investigation Budget (which may be used solely for the purposes authorized in this paragraph 29), no DIP Loans, DIP Collateral, Prepetition Collateral, including Cash Collateral, or any portion of the Carve-Out may be used directly or indirectly by any Debtor, any official committee appointed in the case, including the Creditors' Committee and Retiree Committee, or any trustee appointed in the Chapter 11 Cases or any successor case, including any chapter 7 case, or any other person, party or entity to (i) investigate, object, contest, or raise any defense to the validity, perfection, priority, extent, or enforceability of the DIP Obligations, the Prepetition Obligations or the liens, claims or rights granted under the Amended Final Cash Collateral Order, this Order, the DIP Financing Documents or the Prepetition Debt Documents, or take any action purporting to do any of the foregoing; (ii) investigate, assert or prosecute any Claims and Defenses against the DIP Agent, the DIP Lenders, the Prepetition Secured Parties or their respective predecessors-in-interest,

agents, affiliates, representatives, attorneys, or advisors or take any action purporting to do the foregoing in respect of the DIP Obligations, Prepetition Obligations, DIP Claims and Liens, Prepetition Liens, and/or the Adequate Protection Obligations, Adequate Protection Liens and Superpriority Claims granted to the Prepetition Secured Parties under this Order or the Amended Final Cash Collateral Order, as applicable; (iii) prevent, hinder, or otherwise delay the DIP Agent's, the DIP Lenders' or the Prepetition Secured Parties', as applicable, enforcement, or realization on the DIP Obligations, DIP Collateral, Prepetition Obligations, Prepetition Collateral, and the liens, claims and rights granted to such parties under the Amended Final Cash Collateral Order and this Order, in accordance with the DIP Financing Documents, the Prepetition Debt Documents, the Amended Final Cash Collateral Order or this Order, as applicable; (iv) seek to modify any of the rights granted to the DIP Agent, the DIP Lenders, or Prepetition Secured Parties hereunder (other than with the consents contemplated hereunder) or under the Amended Final Cash Collateral Order, the DIP Financing Documents or the Prepetition Debt Documents, as applicable; (v) apply to the Court for authority to approve superpriority claims or grant liens (other than the Approved Liens) or security interests in the DIP Collateral or any portion thereof that are senior to, or pari passu with, the DIP Liens, DIP Claims, Adequate Protection Liens, Prepetition Secured Parties' Superpriority Claims or Prepetition Liens, unless all DIP Obligations, Prepetition Obligations, Adequate Protection Obligations, and claims granted to the DIP Agent, DIP Lenders or Prepetition Secured Parties under the Amended Final Cash Collateral Order and this Order, as applicable, have been refinanced or paid in full in cash (including the cash collateralization of any letters of credit) or otherwise agreed to in writing by the Required Lenders, DIP Agent or the Steering Committee and the Administrative Agent, as applicable, each in its sole discretion; or (vi) seek to pay any

amount on account of any claims arising prior to the Petition Date unless such payments are agreed to in writing by the Required Lenders in their sole discretion or are otherwise included in the Approved Budget.

- 30. *Employee Incentive/Retention Plans*. Other than the key employee retention plan approved by the KERP Order, the Debtors will not seek approval of any employee incentive or retention plans (or any similar sort of retention or incentive program) without the prior written consent of the Backstop Parties, which consent shall be in their sole discretion.
- 31. Executory Contracts and Unexpired Leases. The Debtors will confer with the Steering Committee, the Backstop Parties, and its/their advisors to determine which executory contracts and unexpired leases should be assumed or rejected by the Debtors. The Debtors will provide the Steering Committee, the Backstop Parties, and its/their advisors with all necessary information in order to analyze such a decision. Other than as contemplated by the Sale Motion, the Debtors shall not make any decision with regard to the assumption or rejection of executory contracts and unexpired leases without first obtaining the consent of the Steering Committee and the Backstop Parties and which consent shall each be in its/their respective sole discretion.
- 32. Restrictions on Disposition of Material Assets Outside the Ordinary Course of Business. Except as contemplated by the Sale Motion or expressly permitted under the "first day" pleadings, the Debtors shall not use, sell or lease any material assets outside the ordinary course of business, or seek authority of this Court to the extent required by Bankruptcy Code section 363, without obtaining the prior written consent of the Backstop Parties, which consent shall be in their sole discretion, at least five (5) business days prior to the date on which the Debtors seek the Court's authority for such use, sale or lease. Subject to paragraph 11(c) hereof and the rights of any holder of a Permitted Priority Lien thereon, in the event of any such sale,

lease, transfer, license, or other disposition of property of the Debtors (other than a disposition of all or substantially all of the Debtors' assets) that constitutes DIP Collateral outside the ordinary course of business (to the extent permitted by the DIP Financing Documents and this Order), the Debtors are authorized and shall promptly pay, without further notice or order of this Court, the DIP Agent, for the benefit of the DIP Lenders, 100% of the net cash proceeds resulting therefrom no later than the second business day following receipt of such proceeds. In the event of any casualty, condemnation, or similar event with respect to property that constitutes DIP Collateral, the Debtors are authorized and shall promptly pay to the DIP Agent, for the benefit of the DIP Lenders, any insurance proceeds, condemnation award, or similar payment (excluding any amounts on account of any D&O policies) in excess of \$2,000,000 no later than the second business day following receipt of payment by the Debtors, unless the DIP Agent and the Backstop Parties consent, each in its/their sole discretion, in writing, to the funds being reinvested by the Debtors.

- 33. Prepetition Intercreditor Agreements. Nothing in this Order shall amend or otherwise modify the terms and enforceability of the Intercreditor Agreement, which shall remain in full force and effect. The rights of the Prepetition Secured Parties shall at all times remain subject to the Intercreditor Agreement and any other applicable intercreditor agreements.
- 34. Binding Effect; Successors and Assigns. The provisions of this Order, the DIP Credit Agreement and the other DIP Financing Documents shall be binding upon all parties in interest in the Chapter 11 Cases and any successor cases, including any chapter 7 cases, including the Prepetition Secured Parties, the Backstop Parties, the Steering Committee, the Creditors' Committee, the Retiree Committee, the Debtors and their respective successors and assigns (including any Trustee hereinafter appointed or elected for the estates of any of the

Debtors, an examiner appointed pursuant to Bankruptcy Code section 1104, or any other fiduciary appointed as a legal representative of any of the Debtors or with respect to the property of the estate of any of the Debtors) as provided herein.

35. Limitation of Liability. In determining to make any loan under the DIP Financing Documents, permitting use of the Cash Collateral, or exercising any rights or remedies as and when permitted pursuant to this Order, the DIP Financing Documents or the Prepetition Debt Documents, the DIP Agent, the DIP Lenders, and the Prepetition Secured Parties shall not be deemed to be in control of the operations of the Debtors or to be acting as a "responsible person" or "owner or operator" with respect to the operation or management of the Debtors or their businesses (as such terms, or any similar terms, are used in the United States Comprehensive Environmental Response, Compensation and Liability Act, 29 U.S.C. §§ 9601 et seq. as amended, or any similar federal or state statute), nor shall they owe any fiduciary duty to any of the Debtors, their creditors or estates, or constitute or be deemed to constitute a joint venture or partnership with any of the Debtors. Furthermore, nothing in this Order, the DIP Financing Documents or the Prepetition Debt Documents shall in any way be construed or interpreted to impose or allow the imposition upon the DIP Agent, the DIP Lenders or the Prepetition Secured Parties of any liability for any claims arising from the prepetition or postpetition activities of any of the Debtors and their respective affiliates (as defined in Bankruptcy Code section 101(2)).

36. No Modification of Order. Until and unless the DIP Obligations, DIP Superpriority Claim, Prepetition Secured Parties' Superpriority Claims, and Prepetition Obligations have been indefeasibly paid in full in cash (such payment being without prejudice to any terms or provisions contained in the DIP Facility which survive such discharge by their terms), and all commitments to extend credit under the DIP Facility have been terminated, the

Debtors irrevocably waive the right to seek and shall not seek or consent to, directly or indirectly: (a) except as permitted under the DIP Financing Documents, and with the prior written consent of the Backstop Parties (i) any modification, stay, vacatur, or amendment to this Order; (ii) a priority claim for any administrative expense or unsecured claim against the Debtors (now existing or hereafter arising of any kind or nature whatsoever, including, but not limited to any administrative expense of the kind specified in Bankruptcy Code sections 503(b), 507(a), or 507(b)) in any of the Chapter 11 Cases or successor cases, including any chapter 7 cases, equal or superior to the DIP Superpriority Claim, the DIP Obligations, Prepetition Secured Parties' Superpriority Claims, or the Prepetition Obligations, other than the Carve-Out, or (iii) any other order allowing use of the DIP Collateral or Prepetition Collateral; and (b) except as permitted under the DIP Financing Documents, any lien on any of the DIP Collateral or Prepetition Collateral with priority equal or superior to the DIP Liens, Adequate Protection Liens or Prepetition Liens, respectively. Each Debtor irrevocably waives any right to seek any amendment, modification or extension of this Order without the prior written consent of the DIP Agent, the Backstop Parties and the Steering Committee, as applicable, each in its/their sole discretion, and no such consent shall be implied by any action, inaction or acquiescence of the applicable DIP Agent, DIP Lenders, the Backstop Parties or the Steering Committee.

37. Priorities Among Prepetition Secured Parties. Notwithstanding anything to the contrary herein or in any other order of this Court, in determining the relative priorities among, and rights of, the Prepetition Secured Parties (including, but not limited to, the relative priorities and rights of the Prepetition Secured Parties with respect to the adequate protection granted hereunder), such relative priorities and rights shall continue to be governed by the Prepetition Debt Documents and the Intercreditor Agreement.

38. Rights of Administrative Agent and First Lien Trustee. Nothing in this Order shall be construed to limit or affect the (i) Administrative Agent's right to request instructions from the Required Lenders (as defined in the Credit Agreement) in accordance with Section 12.04 of the Credit Agreement and (ii) First Lien Trustee's right to request directions from holders of a majority in aggregate principal amount of the then outstanding First Lien Notes (the "Majority First Lien Noteholders") in accordance with Sections 6.05, 7.01, and 10.02 of the First Lien Indenture. It being understood that since the Steering Committee constitutes the Required Lenders and the Majority First Lien Noteholders and can therefore direct the Administrative Agent in taking actions in connection with the Credit Agreement and the First Lien Trustee in taking actions in connection with the First Lien Indenture, in any instance in this Order where the Administrative Agent or the First Lien Trustee is indicated as having given its consent, as the Steering Committee has also given its consent in such case, the Administrative Agent and the First Lien Trustee shall be deemed to have given its consent at the direction of the Required Lenders and Majority First Lien Noteholders, as applicable.

39. No Waiver. This Order shall not be construed in any way as a waiver or relinquishment of any rights that the DIP Agent, the DIP Lenders or the Prepetition Secured Parties may have to bring or be heard on any matter brought before this Court. Notwithstanding anything herein to the contrary, the entry of this Order is without prejudice to, and does not constitute a waiver of, expressly or implicitly, the rights, claims and defenses of the issuers of surety bonds on which the Debtors are principals or indemnitors under applicable bankruptcy and non-bankruptcy law, including any such issuer's rights, claims and defenses under any existing indemnity agreements, surety bonds or related agreements or any letters of credit related thereto, all of which are expressly reserved.

40. Automatic Stay Modified. The automatic stay shall be modified or lifted to the extent necessary to allow the DIP Agent, the Backstop Parties or the Prepetition Secured Parties, as applicable, to provide any notices to the Debtors or take any other action as contemplated by

and in accordance with this Order.

41. Rights Preserved. Notwithstanding anything herein to the contrary, the entry of

this Order is without prejudice to, and does not constitute a waiver of, expressly or implicitly the

DIP Agent's, the DIP Lenders', the Prepetition Secured Parties', the Steering Committee's or the

Backstop Parties' respective right to seek any other or supplemental relief in respect of the

Debtors. Nothing in this Order shall relieve the Debtors of any obligations under federal, state or

local police or regulatory laws or under 28 U.S.C. § 959(b).

42. Federal Securities Laws. Nothing in this Order shall relieve or excuse the

Debtors, the DIP Lenders, the DIP Agent, or any other party from complying with any and all

applicable federal securities laws (including relevant rules and regulations promulgated

thereunder) with respect to the matters set forth herein, and nothing in this Order shall impair the

Securities and Exchange Commission's police or regulatory powers with respect thereto.

43. Effectiveness. This Order shall constitute findings of fact and conclusions of law

pursuant to Bankruptcy Rule 7052 and shall take effect immediately upon entry hereof, and there

shall be no stay of execution of effectiveness of this Order. Any finding of fact shall constitute a

finding of fact even if it appears as a conclusion of law, and any conclusion of law shall

constitute a conclusion of law even if it appears as a finding of fact.

44. Controlling Effects of Order. To the extent of any conflict between or among (a)

the Motion, any other order of this Court, the DIP Credit Agreement or any other agreements, on

the one hand, and (b) the terms and provisions of this Order, on the other hand, unless such term

or provision herein is phrased in terms of "as defined in" or "as more fully described in" or "as

provided in" or words to that effect with respect to the DIP Financing Documents or the

Prepetition Debt Documents, as applicable, the terms and provisions of this Order shall govern.

45. Jurisdiction. This Court shall retain jurisdiction to enforce the terms of this Order

and to adjudicate any and all matters arising from or related to the interpretation, implementation

or enforcement of this Order.

Dated: January 28, 2016

/s/ Tamara O. Mitchell

THE HONORABLE TAMARA O. MITCHELL UNITED STATES BANKRUPTCY JUDGE

# Exhibit A

**DIP Credit Agreement** 

### SENIOR SECURED SUPERPRIORITY

## DEBTOR-IN-POSSESSION CREDIT AGREEMENT

among

WALTER ENERGY, INC.,

as Borrower

THE LENDERS FROM TIME TO TIME PARTY HERETO,

and

CITIBANK, N.A.,

as ADMINISTRATIVE AGENT and COLLATERAL AGENT

Dated as of [●]

## TABLE OF CONTENTS

		<b>Page</b>			
SECTION 1	Definitions and Accounting Terms	7			
1.01	Defined Terms	7			
1.02	Other Definitional Provisions				
SECTION 2	Amount and Terms of Credit	26			
2.01	The Commitments	26			
2.02	Conditions to Each Withdrawal				
2.03	Notice of Withdrawal				
2.04	Disbursement of Funds				
2.05	Notes	28			
2.06	[Reserved]	28			
2.07	Pro rata Fundings	29			
2.08	Interest	29			
2.09	[Reserved]	29			
2.10	Increased Costs				
2.11	[Reserved]	30			
2.12	[Reserved]				
SECTION 3	[Reserved]	30			
SECTION 4	Fees	30			
4.01	Fees	30			
SECTION 5	Prepayments; Payments; Taxes	31			
5.01	Voluntary Prepayments	31			
5.02	Mandatory Prepayments	31			
5.03	Repayment of Loans	32			
5.04	Method and Place of Payment	32			
5.05	Net Payments				
SECTION 6	Conditions Precedent to Credit Events on the Initial Funding Date	35			
6.01	Credit Agreement	35			
6.02	Officer's Certificate				
6.03	[Reserved]	35			
6.04	Secretary's Certificate; Organization Documents; etc.				
6.05	Adverse Change, Approvals				
6.06	[Reserved]				
6.07	UCC Filings				
6.08	[Reserved]				

6.09	Insurance	36				
6.10	Fees.					
6.11	Patriot Act					
6.12	[Reserved]					
6.13	Approved Budget					
6.14	Capital Expenditure Budget	37				
6.15	Bankruptcy Matters	37				
SECTION 7	Conditions Precedent to All Credit Events	38				
7.01	No Default; Representations and Warranties					
7.02	Notice of Withdrawal					
7.03	Withdrawal Conditions					
7.04	Sale Order					
7.05	Asset Purchase Agreement					
7.06	Final DIP Order					
7.07	Fees and Expenses, Etc.					
7.08	Bankruptcy Matters					
7.09	Letter of Direction.	39				
SECTION 8	Representations, Warranties and Agreements	39				
8.01	Organization; Powers					
8.02	Authorization; Enforceability					
8.03	No Violation					
8.04	Approvals					
8.05	Financial Statements; Financial Condition					
8.06	Litigation; Defaults					
8.07	True and Complete Disclosure					
8.08	Ownership of Property; Liens; Investments					
8.09	Tax Returns and Payments					
8.10	Compliance with ERISA					
8.11	[Reserved]					
8.12	Insurance					
8.13	Subsidiaries					
8.14	Compliance with Statutes, etc.					
8.15	Investment Company Act; Margin Regulations					
8.16 8.17	Environmental Matters					
8.18	Employment and Labor Relations					
	Intellectual Property, etc.					
8.19 8.20	Representations and Warranties in Asset Purchase Agreement.					
8.20 8.21	Foreign Assets Control Regulations, Etc.					
8.21 8.22	[Reserved] Permits; Licenses					
8.22 8.23	Bank Accounts and Securities Accounts					
8.24	[Reserved]					
8.25	Reorganization Matters; Administrative Priority; Lien Priority					
0.4 <i>3</i>	ixongamzanon manors, manimishanyo i nonty, Lich i nolity,	,				

SECTION 9	Affirmative Covenants			
9.01	Information Covenants	48		
9.02	Books, Records and Inspections	52		
9.03	Maintenance of Property; Insurance	52		
9.04	Existence; Franchises	53		
9.05	Compliance with Statutes, etc.	53		
9.06	Compliance with Environmental Laws	53		
9.07	ERISA	54		
9.08	End of Fiscal Years; Fiscal Quarters	54		
9.09	Payment of Taxes	54		
9.10	Use of Proceeds	55		
9.11	Additional Security; Further Assurances; etc.	56		
9.12	Pleadings	58		
9.13	Bi-Weekly Calls	58		
9.14	Cooperation	58		
9.15	Cash Management			
SECTION 10	Negative Covenants	58		
10.01	Liens	58		
10.02	Fundamental Changes	61		
10.03	Restricted Payments	61		
10.04	Indebtedness	62		
10.05	Investments	63		
10.06	Transactions with Affiliates	64		
10.07	Asset Sales	65		
10.08	Limitation on Certain Restrictions on Subsidiaries	66		
10.09	Business; etc	66		
10.10	Financial Covenants	66		
10.11	Amendments of Organization Documents	67		
10.12	Chapter 11 Cases	67		
10.13	Prepayments, Etc. of Indebtedness			
10.14	Sale Leasebacks			
10.15	Anti-Terrorism Laws	68		
10.16	Intercompany Limitation	68		
SECTION 11	Events of Default	68		
11.01	Payments			
11.02	Representations, etc.			
11.03	Covenants	69		
11.04	Default Under Other Agreements	69		
11.05	Bankruptcy, etc.			
11.06	[ERISA			
11.07	Security Documents	70		
11.08	Guaranties	70		

11.09	Judgments	70			
11.10	Change of Control				
11.11	Sale of all or Substantially all Assets				
11.12	Chapter 11 Cases				
11.13	Application of Proceeds				
SECTION 12	The Administrative Agent	74			
12.01	Appointment	74			
12.02	Nature of Duties				
12.03	Lack of Reliance on the Administrative Agent				
12.04	Certain Rights of the Administrative Agent				
12.05	Reliance				
12.06	Indemnification				
12.07	Resignation by the Administrative Agent				
12.08	Collateral Matters	76			
12.09	Delivery of Information	77			
SECTION 13	Guaranty	78			
13.01	Guaranty				
13.02	Amendments, etc. with respect to the Obligations	78			
13.03	Guaranty Absolute and Unconditional	79			
13.04	Reinstatement	80			
13.05	Payments	80			
13.06	Information	80			
13.07	Indemnity and Subrogation	81			
13.08	Contribution and Subrogation	81			
13.09	Subordination	81			
SECTION 14	Miscellaneous	82			
14.01	Payment of Expenses; Indemnifications etc.	82			
14.02	Right of Setoff	84			
14.03	Notices				
14.04	Benefit of Agreement; Assignments; Participations				
14.05	No Waiver; Remedies Cumulative	87			
14.06	Payments Pro Rata	88			
14.07	Calculations; Computations	88			
14.08	GOVERNING LAW; SUBMISSION TO JURISDICTION; VENUE;				
	WAIVER OF JURY TRIAL	89			
14.09	Counterparts	90			
14.10	Headings Descriptive	90			
14.11	Amendment or Waiver; etc.	90			
14.12	Survival	92			
14.13	Domicile of Loans.	92			
14.14	[Reserved]	93			

11.15	Community	wiity	, 5
14.16	Patriot Act		93
		te Limitation	94
14.18	Waiver of Immunities		94
14.19	No Adviso	ry or No Fiduciary Duty	94
14.20		Lender Instructions	
14.21	Final DIP	Order	95
SCHEDULE 1.01	(a)	Commitments	
SCHEDULE 1.01	· /	Lender Addresses	
SCHEDULE 1.01	` /	Guarantors	
SCHEDULE 8.08	· /	Owned Real Property	
SCHEDULE 8.08	` '	Leased Real Property	
SCHEDULE 8.11	` /	Financing Statements	
SCHEDULE 8.12		Real Property	
SCHEDULE 8.13		Subsidiaries	
SCHEDULE 8.16		Environmental Matters	
SCHEDULE 8.21		Material Contracts	
SCHEDULE 8.23		Bank Accounts and Securities Accounts	
SCHEDULE 8.25		Senior Liens	
SCHEDULE 10.0		Existing Liens	
SCHEDULE 10.0	4	Existing Indebtedness	
SCHEDULE 10.0	5	Existing Investments	
SCHEDULE 10.0	8	Existing Restrictions	
EXHIBIT A		Form of Notice of Withdrawal	
EXHIBIT B		Form of Note	
EXHIBIT C		Form of Tax Compliance Certificate	
EXHIBIT D		Form of Compliance Certificate	
EXHIBIT E		Form of Assignment and Assumption Agreement	
EXHIBIT F		Form of Intercompany Note	

14.15

Confidentiality .......93

This SENIOR SECURED SUPERPRIORITY DEBTOR-IN-POSSESSION CREDIT AGREEMENT, dated as of [●], among Walter Energy, Inc., a Delaware corporation, as a debtor and debtor-in-possession, as borrower (the "Borrower"), each domestic subsidiary of the Borrower (each a "Guarantor" and collectively, "Guarantors"), each Lender from time to time party hereto (collectively, the "Lenders" and individually a "Lender"), and Citibank, N.A., as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and as collateral agent for the Secured Creditors (in such capacity, the "Collateral Agent" and together with the Administrative Agent, collectively, the "Agent"). All capitalized terms used herein and defined in Section 1.01 are used herein as therein defined.

### WITNESSETH:

WHEREAS, on July 15, 2015 (the "Petition Date"), the Borrower and certain Subsidiaries of the Borrower (collectively, the "Debtors" and each individually, a "Debtor") have commenced cases (the "Chapter 11 Cases") under Chapter 11 of Title 11 of the Bankruptcy Code in the United States Bankruptcy Court for the Northern District of Alabama (the "Bankruptcy Court"), and the Debtors have retained possession of their assets and are authorized under the Bankruptcy Code to continue the operations of their businesses as debtors-in-possessions; and

WHEREAS, the Borrower and the Guarantors have asked the Lenders to make postpetition term loans and provide other financial or credit accommodations to the Borrower, and the Lenders have agreed, subject to the conditions set forth herein, to extend a senior secured credit facility to the Borrower, comprised of term loans in the aggregate principal amount of up to \$50,000,000. The Lenders have severally, and not jointly, agreed to extend such credit to the Borrower, subject to the terms and conditions hereinafter set forth.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained, the parties hereto and agree as follows:

### SECTION 1 Definitions and Accounting Terms.

- 1.01 <u>Defined Terms</u>. As used in this Agreement, the following terms shall have the following meanings (such meanings to be equally applicable to both the singular and plural forms of the terms defined):
  - "Additional Security Documents" shall have the meaning provided in Section 9.11(b).
  - "Adequate Protection Liens" shall have the meaning provided in the Final DIP Order.
- "<u>Adequate Protection Obligations</u>" shall have the meaning provided in the Final DIP Order.
- "Adjusted Net Worth" of any Guarantor at any time, shall mean the greater of (x) \$0 and (y) the amount by which the fair saleable value of such Guarantor's assets on the date of the respective payment hereunder exceeds its debts and other liabilities (including contingent liabilities, but without giving effect to any of its obligations under this Agreement or any other Credit Document).

"<u>Administrative Agent</u>" shall have the meaning provided in the introductory paragraph hereto.

"Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly controlling (including, but not limited to, all directors and officers of such Person), controlled by, or under direct or indirect common control with, such Person. A Person shall be deemed to control another Person if such Person possesses, directly or indirectly, the power (i) to vote 10% or more of the securities having ordinary voting power for the election of directors (or equivalent governing body) of such Person or (ii) to direct or cause the direction of the management or policies of such other Person, whether through the ownership of voting securities, by contract or otherwise; provided, however, that none of the Administrative Agent, the Collateral Agent, any Lender or any of their respective Affiliates shall be considered an Affiliate of the Borrower or any Subsidiary thereof.

"Agent" shall have the meaning provided in the introductory paragraph hereto.

"Agency Fees" shall mean have the meaning provided in Section 4.01(d).

"Agent Fee Letter" shall mean [●].

"Agreement" shall mean this Senior Secured Superpriority Debtor-in-Possession Credit Agreement, as modified, supplemented, amended, restated (including any amendment and restatement hereof), extended or renewed from time to time.

"<u>APA Closing Date</u>" shall mean the "Closing Date" as defined in the Asset Purchase Agreement.

"Approved Budget" shall have the meaning provided in <u>Section 6.13</u>. For the avoidance of doubt, once delivered and approved as set forth in <u>Section 6.13</u>, the Approved Budget may not be amended, or modified without the prior written consent of the Required Lenders.

"Approved Collateralized Obligations" shall have the meaning provided in the Cash Collateral Order.

"Approved Fund" shall mean with respect to any Lender which is a fund that invests in loans, any other fund that invests in loans that is managed by the same investment advisor as such Lender or by an Affiliate of such Lender or such investment advisor.

"Approved Liens" shall have the meaning provided in the Cash Collateral Order.

"Asset Purchase Agreement" shall mean the Asset Purchase Agreement, dated as of November 5, 2015, by and among Coal Acquisition LLC, as buyer (the "Buyer"), the Borrower and certain subsidiaries of Borrower, as sellers, as the same may be amended, amended and restated, supplemented or otherwise modified from time to time in accordance with the terms thereof.

"Asset Sale" shall mean any sale, transfer, license, lease or sublease, assignment, conveyance or other disposition (including any sale and leaseback transaction) by the Borrower

or any of its Domestic Subsidiaries of any asset or property of any kind, whether real, personal or mixed and whether tangible or intangible, whether now owned or hereafter acquired, by an Person (or the granting of any option or other right to do any of the foregoing), including without limitation, any sale, assignment, transfer or other disposal, with or without recourse, of any notes or accounts receivable or any rights and claims associated therewith, or the Equity Interests of any Person and any issuance of Equity Interests of any Subsidiary of the Borrower, but excluding sales of assets pursuant to Sections 10.07(b), (c), (d), (f), (g) and (h).

"Assignment and Assumption Agreement" shall mean an Assignment and Assumption Agreement substantially in the form of Exhibit E (appropriately completed).

"Authorized Officer" shall mean, with respect to (i) delivering Notices of Borrowing and similar notices, any person or persons that has or have been authorized by the board of directors of the Borrower to deliver such notices pursuant to this Agreement, (ii) delivering financial information and officer's certificates pursuant to this Agreement, the chief executive officer, president, chief financial officer, treasurer, assistant treasurer, controller or principal accounting officer of the Borrower, and (iii) any other matter in connection with this Agreement or any other Credit Document, any officer (or a person or persons so designated by any two officers) of the Borrower.

"Availability Period" shall mean the period commencing on the Effective Date and ending on (but excluding) the Maturity Date.

"Bankruptcy Code" shall have the meaning provided in Section 11.05.

"<u>Bankruptcy Court</u>" shall have the meaning provided in the introductory paragraph hereto.

"Borrower" shall have the meaning provided in the introductory paragraph hereto.

"Borrower Materials" shall have the meaning provided in Section 9.01(j).

"Budget Testing Period" shall mean the two week period beginning with the week in which the Effective Date occurs, and each cumulative period beginning with the week in which the Effective Date occurs and ending every two weeks after the first two week period.

"Business Day" shall mean for all purposes, any day except Saturday, Sunday and any day which shall be in New York, New York, a legal holiday or a day on which banking institutions are authorized or required by law or other government action to close.

"Buyer" shall have the meaning provided in the definition of "Asset Purchase Agreement."

"Canadian Entities" shall have the meaning provided in the Cash Collateral Order.

"Capital Expenditure Budget" shall have the meaning provided in Section 6.14.

"Capital Expenditure Testing Period" shall mean each cumulative period beginning January 1, 2016 and ending at the end of each calendar month thereafter.

"Capitalized Lease Obligations" shall mean, with respect to any Person, all rental obligations of such Person which, under GAAP, are or will be required to be capitalized on the books of such Person, in each case, taken at the amount thereof accounted for as indebtedness in accordance with such principles, <u>provided</u> that notwithstanding the foregoing, in no event will any obligation in respect of a lease that would have been categorized as an operating lease in accordance with GAAP on the Effective Date be considered a Capitalized Lease Obligation.

"Cash Collateral" shall have the meaning provided in the Cash Collateral Order.

"Cash Collateral Order" shall mean the Amended Final Order (A) Authorizing Postpetition Use of Cash Collateral, (B) Granting Adequate Protection to Postpetition Secured Parties and (C) Granting Related Relief [Docket No. 797], as amended from time to time with the consent of the Steering Committee and the Prepetition Administrative Agent, each in their sole discretion.

"Carve-Out" shall have the meaning provided in the Final DIP Order.

"Cash Equivalents" shall mean, as to any Person,

- (a) securities issued or unconditionally guaranteed by the United States government or the government of Canada and, in each case, any agency or instrumentality thereof, in each case having maturities of not more than twenty-four months from the date of acquisition thereof;
- (b) securities issued by any state of the United States of America, any province of Canada or any political subdivision of any such state, province or any public instrumentality thereof or any political subdivision of any such state or any public instrumentality thereof having maturities of not more than twenty-four months from the date of acquisition thereof and, at the time of acquisition, having an investment grade rating generally obtainable from either S&P or Moody's (or, if at any time neither S&P nor Moody's shall be rating such obligations, then from another nationally recognized rating service);
- (c) commercial paper maturing no more than 12 months after the date of creation thereof and, at the time of acquisition, having a rating of at least A-2 or P-2 from either S&P or Moody's (or, if at any time neither S&P nor Moody's shall be rating such obligations, an equivalent rating from another nationally recognized rating service);
- (d) domestic, LIBOR certificates and CDOR certificates of deposit maturing no more than two years (and with respect to bankers' acceptances, twelve months) after the date of acquisition thereof issued by any Lender or any other bank having combined capital and surplus of not less than \$500,000,000 in the case of domestic banks and \$100,000,000 (or the Dollar Equivalent thereof) in the case of foreign banks;

- (e) repurchase agreements with a term of not more than 90 days for underlying securities of the type described in clauses (a), (b) and (d) above entered into with any bank meeting the qualifications specified in clause (d) above or securities dealers or recognized national standing;
- (f) marketable short-term money market and similar funds (x) either having assets in excess of \$500,000,000 or (y) having a rating of at least A-2 or P-2 from either S&P or Moody's (or, if at any time neither S&P nor Moody's shall be rating such obligations, an equivalent rating from another nationally recognized rating service);
- (g) shares of investment companies that are registered under the Investment Company Act of 1940 and substantially all the investments of which are one or more of the types of securities described in clauses (a) through (f) above; and
- (h) Investment of assets made pursuant to any non-qualified deferred compensation plan sponsored by the Borrower or its Subsidiaries.

"Change of Control" shall mean (i) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act) is or shall become the "beneficial owner" (as defined in Rules 13(d)-3 and 13(d)-5 under the Exchange Act, except that a person or group shall be deemed to have "beneficial ownership" of all securities that such person or group has the right to acquire, whether such right is exercisable immediately or only after the passage of time (such right, an "option right")), directly or indirectly, of 35% or more on a fully diluted basis of the economic or voting interests in the Borrower's capital stock (and taking into account all such securities that such "person" or "group" has the right to acquire pursuant to any option right); or (ii) the Board of Directors of the Borrower shall cease to consist of a majority of Continuing Directors.

"Chapter 11 Cases" shall have the meaning provided in the introductory paragraph hereto

"Claims" shall have the meaning provided in the definition of "Environmental Claims".

"Code" shall mean the Internal Revenue Code of 1986, as amended from time to time, and the regulations promulgated and rulings issued thereunder.

"Collateral" shall mean all property (whether real or personal) with respect to which any security interests have been granted (or purported to be granted) pursuant to the Final DIP Order and/or any Security Document.

"Collateral Agent" shall have the meaning provided in the introductory paragraph hereto.

"Commitment" shall mean, with respect to any Lender, such Lender's commitment to make a Loan to the Borrower hereunder in a principal amount not to exceed the amount set forth opposite such Lender's name on (a) as to any Lender that becomes a party to this Agreement on the Effective Date, Schedule 1.01(a) or (b) in the case of any Lender that becomes a party to this Agreement after the Effective Date, the Assignment and Assumption Agreement pursuant to which such Lender becomes a party hereto and assumes a portion of the Commitments, in either

case, as the same may be changed from time to time pursuant to the terms hereof. The original aggregate amount of the Commitments is \$50,000,000.

"Commitment Letter" shall mean the Commitment Letter dated as of January 14, 2016 among the commitment parties party thereto and the Borrower.

"Commitment Parties" shall have the meaning provided in the Commitment Letter.

"Company" shall mean any corporation, limited liability company, partnership or other business entity (or the adjectival form thereof, where appropriate).

"Consultants" shall have the meaning provided in the Final DIP Order.

"Contingent Obligation" shall mean, as to any Person, any obligation of such Person as a result of such Person being a general partner of any other Person, unless the underlying obligation is expressly made non-recourse as to such general partner, and any obligation of such Person guaranteeing or intended to guarantee any Indebtedness, leases, dividends or other obligations ("primary obligations") of any other Person (the "primary obligor") in any manner, whether directly or indirectly, including, without limitation, any obligation of such Person, whether or not contingent, (i) to purchase any such primary obligation or any property constituting direct or indirect security therefor, (ii) to advance or supply funds (x) for the purchase or payment of any such primary obligation or (y) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (iii) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation or (iv) otherwise to assure or hold harmless the holder of such primary obligation against loss in respect thereof; provided, however, that the term Contingent Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Contingent Obligation shall be deemed to be an amount equal to the stated or determinable amount of the primary obligation in respect of which such Contingent Obligation is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof (assuming such Person is required to perform thereunder) as determined by such Person in good faith.

"Continuing Director" shall mean an individual (a) who is a member of the board of directors of the Borrower on the Effective Date, (b) who, as of the date of determination, has been a member of such board of directors for at least the twelve preceding months or (c) has been nominated to be a member of such board of directors by a majority of the other Continuing Directors then in office.

"Contractual Obligation" shall mean, as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

"Credit Documents" shall mean this Agreement, each Note, each other Security Document and the Commitment Letter.

"Credit Event" shall mean the making of any Loan and the making of any Withdrawal.

"Credit Party" shall mean the Borrower and each Guarantor.

"<u>Debtor</u>" shall have the meaning provided in the introductory paragraph hereto.

"<u>Default</u>" shall mean any event or condition that constitutes an Event of Default or that, with the giving of any notice, the passage of time, or both, would constitute an Event of Default.

"<u>Defaulting Lender</u>" shall mean any Lender with respect to which a Lender Default is in effect.

"<u>Default Rate</u>" shall mean when used with respect to Obligations, an interest rate equal to 14% per annum, which 14% per annum shall be payable upon demand in cash.

"DIP Claims and Liens" shall have the meaning provided in the Final DIP Order.

"Disqualified Equity Interest" shall mean, with respect to any Person, any Equity Interest of such Person which, by its terms or by the terms of any security into which it is convertible or for which it is putable or exchangeable, or upon the happening of any event or condition (a) matures or is mandatorily redeemable (other than solely for Qualified Equity Interests), pursuant to a sinking fund obligation or otherwise (except as a result of a change of control or asset sale event, which shall be subject to the prior repayment in full of the Loans and all other Obligations that are accrued and payable, and the termination of the Commitments), (b) is redeemable at the option of the holder thereof (other than solely for Qualified Equity Interests), in whole or in part, (c) provides for the mandatory or scheduled payments of dividends in cash, or (d) is or becomes convertible into or exchangeable for Indebtedness or any other Equity Interests that would constitute Disqualified Equity Interests, in each case, prior to the date that is ninety-one days after the Maturity Date.

"Dividend" shall mean, with respect to any Person, that such Person has declared or paid a dividend (other than dividends payable solely in its Qualified Equity Interests), distribution or returned any equity capital to its stockholders, partners or members or authorized or made any other distribution, payment or delivery of property (other than common Equity Interests of such Person) or cash to its stockholders, partners or members in their capacity as such, or redeemed, retired, purchased or otherwise acquired (other than common Equity Interests of such Person), directly or indirectly, for a consideration any shares of any class of its capital stock or any other Equity Interests outstanding on or after the Effective Date (or any options or warrants issued by such Person with respect to its capital stock or other Equity Interests), or set aside any funds for any of the foregoing purposes, or shall have permitted any of its Subsidiaries to purchase or otherwise acquire for a consideration any shares of any class of the capital stock or any other Equity Interests of such Person outstanding on or after the Effective Date (or any options or warrants issued by such Person with respect to its capital stock or other Equity Interests). Without limiting the foregoing, "Dividends" with respect to any Person shall also include all payments made or required to be made by such Person with respect to any stock appreciation rights, plans, equity incentive or achievement plans or any similar plans or setting aside of any funds for the foregoing purposes.

"<u>Dollars</u>" and the sign "<u>\$</u>" shall each mean freely transferable lawful money of the United States.

"<u>Domestic Subsidiary</u>" of any Person shall mean any Subsidiary of such Person incorporated or organized in the United States or any State or territory thereof or the District of Columbia; <u>provided</u> that no Subsidiary of a Foreign Subsidiary shall be deemed to be a Domestic Subsidiary; <u>provided</u> <u>further</u> that any Subsidiary that would otherwise constitute a Domestic Subsidiary and is a holding company which owns Equity Interests in one or more Foreign Subsidiaries, but owns no other material assets and does not engage in any trade or business (other than acting as a holding company for such Equity Interests in Foreign Subsidiaries) shall not constitute a Domestic Subsidiary hereunder.

"Effective Date" shall mean the first date all conditions precedent in Section 6 are satisfied or waived in accordance with Section 14.11.

"Eligible Transferee" shall mean and include a commercial bank, an insurance company, a finance company, a financial institution, any fund that invests in loans or any other "accredited investor" (as defined in Regulation D of the Securities Act), but in any event excluding the Borrower or any of its Subsidiaries.

"Environmental Claims" shall mean any and all administrative, regulatory or judicial actions, suits, petitions, pleas, charges, demands, demand letters, directives, claims, liens, notices of noncompliance or violation, investigations and/or proceedings (hereafter "Claims") relating in any way to any actual or alleged noncompliance with, or liability arising under, Environmental Law or to any permit issued, or any approval given, under any Environmental Law, including, (a) any and all Claims by any Governmental Authority for enforcement, cleanup, removal, response, remedial or other actions or damages pursuant to any Environmental Law, and (b) any and all Claims by any third party seeking damages, contribution, indemnification, cost recovery, compensation or injunctive relief arising out of or relating to an alleged injury or threat of injury to human health, safety or the environment due to the presence of Hazardous Materials.

"Environmental Law" shall mean any applicable Federal, state, local or foreign law (including principles of common law), constitution, treaty, convention, statute, rule, regulation, ordinance, code, directive, judgment, order, agreement, or other similar authority, now or hereafter in effect and in each case as amended, and any legally binding judicial or administrative interpretation thereof, relating to pollution or protection of the environment and human health and safety (as it relates to the exposure to environmental hazards), including those relating to the presence, Release or threatened Release, or the manufacture, generation, handling, use, transportation, treatment, storage, labeling, disposal, recycling, control, cleanup, or other action or failure to act involving Hazardous Materials, or the arrangement for any such activities.

"Environmental Permit" shall mean any permit, approval, consent, waiver, franchise, registration, filing, accreditation, identification number, license, certificate or other authorization required under any Environmental Law.

"Equity Interests" of any Person shall mean any and all shares, interests, rights to purchase or receive, warrants, options, participation or other equivalents of or interest in (however designated) equity of such Person, including any common stock, preferred stock, any limited or general partnership interest and any limited liability company membership interest.

"ERISA" shall mean the U.S. Employee Retirement Income Security Act of 1974, as amended from time to time, and the regulations promulgated and rulings issued thereunder.

"ERISA Affiliate" shall mean any person that for purposes of Title I or Title IV of ERISA or Section 412 of the Code would be deemed at any relevant time to be a single employer or otherwise aggregated with the Borrower or any of its Subsidiaries under Section 414(b) or (c) of the Code or Section 4001 of ERISA.

"ERISA Event" shall mean any one or more of the following:

- (a) any Reportable Event;
- (b) the filing of a notice of intent to terminate any Plan, if such termination would require material additional contributions in order to be considered a standard termination within the meaning of Section 4041(b) of ERISA, the filing under Section 4041(c) of ERISA of a notice of intent to terminate any Plan or the termination of any Plan under Section 4041(c) of ERISA;
- (c) the institution of proceedings, or the occurrence of an event or condition which constitutes grounds for the institution of proceedings by the PBGC under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan;
- (d) the failure to make a required contribution to any Plan that would result in the imposition of a lien or other encumbrance or the provision of security under Section 430 of the Code or Section 303 or 4068 of ERISA, or the arising of such a lien or encumbrance; there being or arising any "unpaid minimum required contribution" or "accumulated funding deficiency" (as defined or otherwise set forth in Section 4971 of the Code or Part 3 of Subtitle B of Title I of ERISA), whether or not waived; or the filing of any request for or receipt of a minimum funding waiver under Section 412 of the Code with respect to any Plan;
- (e) engaging in a non-exempt prohibited transaction within the meaning of Section 4975 of the Code or Section 406 of ERISA;
- (f) the complete or partial withdrawal of the Borrower or any ERISA Affiliate from a Multiemployer Plan, the reorganization or insolvency under Title IV of ERISA of any Multiemployer Plan; or the receipt by the Borrower or any ERISA Affiliate, of any notice, or the receipt by any Multiemployer Plan from any of the Borrower or any ERISA Affiliate of any notice, that a Multiemployer Plan is in endangered or critical status under Section 305 of ERISA; or
- (g) Borrower or an ERISA Affiliate incurring any liability under Title IV of ERISA with respect to any Plan (other than PBGC premiums due and not delinquent under Section 4007 of ERISA).

"<u>Escrow Account</u>" shall mean the account maintained by the Administrative Agent for purposes of holding the Loans in escrow as set forth in Section 2.01.

"Event of Default" shall have the meaning provided in Section 11.

"Executive Order" shall have the meaning provided in Section 8.19.

"Extraordinary Receipt" shall mean any cash received by or paid to or for the account of any Person not in the ordinary course of business, including tax refunds, pension plan reversions, cash proceeds from any Recovery Event, proceeds of insurance (other than proceeds of business interruption insurance to the extent such proceeds constitute compensation for lost earnings), condemnation awards (and payments in lieu thereof), and indemnity payments.

"<u>Facility</u>" shall mean the aggregate amount of the Commitments and the Loans of all Lenders outstanding at such time.

"FATCA" shall mean Sections 1471 through 1474 of the Code, as enacted on the Effective Date, any amended or successor provisions that are substantively similar thereto and which do not impose criteria that are materially more onerous than those contained in such Sections as enacted on the Effective Date, any current or future Treasury regulations thereunder or official interpretations thereof, any agreements entered into pursuant to Section 1471(b)(1) of the Code (or any amended or successor version described above) and any intergovernmental agreements (together with any laws implementing such agreements) implementing the foregoing.

"Federal Funds Rate" shall mean, for any period, a fluctuating interest rate equal for each day during such period to the weighted average of the rates on overnight Federal Funds transactions with members of the Federal Reserve System arranged by Federal Funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by the Administrative Agent from three Federal Funds brokers of recognized standing selected by the Administrative Agent.

"Fees" shall have the meaning provided in Section 4.01(d).

"Final DIP Order" shall mean the Order (A) Authorizing the Debtors to Obtain Senior Secured Postpetition Financing, (B) Authorizing Continued Post Petition Use of Cash Collateral, (C) Granting Adequate Protection to Prepetition Secured Parties and (D) Granting Related Relief, entered by the Bankruptcy Court, which Final DIP Order shall be in form and substance acceptable to the Debtors, the Commitment Parties and the Administrative Agent.

"Fiscal Quarter" shall mean, for any Fiscal Year, (i) the fiscal period commencing on January 1 of such Fiscal Year and ending on March 31 of such Fiscal Year, (ii) the fiscal period commencing on April 1 of such Fiscal Year and ending on June 30 of such Fiscal Year, (iii) the fiscal period commencing on July 1 of such Fiscal Year and ending on September 30 of such Fiscal Year and (iv) the fiscal period commencing on October 1 of such Fiscal Year and ending on December 31 of such Fiscal Year.

"<u>Fiscal Year</u>" shall mean the fiscal year of the Borrower and its Subsidiaries ending on December 31 of each calendar year.

"Foreign Assets Control Regulations" shall have the meaning provided in Section 8.19.

"Foreign Lender" shall have the meaning provided in Section 5.05(b).

"Foreign Subsidiary" of any Person shall mean any Subsidiary of such Person that is not a Domestic Subsidiary.

"<u>FRB</u>" means the Board of Governors of the Federal Reserve System of the United States.

"Funding Date" means the Initial Funding Date or the Second Escrowed DIP Loan Funding Date, as applicable.

"GAAP" shall mean generally accepted accounting principles in the United States as in effect from time to time; provided, however, that if there occurs after the Effective Date any change in GAAP that affects in any respect the calculation of Applicable Margin or any covenant contained in Section 10, the Lenders and the Borrower shall negotiate in good faith amendments to the provisions of this Agreement that relate to the calculation of such covenant with the intent of having the respective positions of the Lenders and the Borrower after such change in GAAP conform as nearly as possible to their respective portions as of the Effective Date and until such amendments have been agreed upon the definition of Applicable Margin and the covenants in Section 10 shall be calculated as if no change in GAAP had occurred.

"Governmental Authority" shall mean the government of the United States of America, any other nation or any political subdivision thereof, whether state, provincial or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.

"Guarantor" shall have the meaning provided in the introductory paragraph hereto.

"Guaranty" shall mean, as to any Person, without duplication, any (a) any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Indebtedness or other monetary obligation payable or performable by another Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of such Person, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other monetary obligation, (ii) to purchase or lease property, securities or services for the purpose of assuring the obligee in respect of such Indebtedness or other monetary obligation of the payment or performance of such Indebtedness or other monetary obligation, (iii) to maintain working capital, equity capital or any other financial statement condition or liquidity or level of income or cash flow of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other monetary obligation, or (iv) entered into for the purpose of assuring in any other manner the obligee in respect of such Indebtedness or other monetary obligation of the payment or performance thereof or to protect such obligee against loss in respect thereof (in whole or in part), or (b) any Lien on any assets of such Person securing any Indebtedness or other monetary obligation of any other Person, whether or not such Indebtedness or other monetary obligation is assumed by such Person (or any right, contingent or otherwise, of any holder of such Indebtedness to obtain any such Lien). The term "Guarantee" as a verb has a corresponding meaning.

"<u>Hazardous Materials</u>" shall mean any chemicals, materials, wastes, pollutants, contaminants or substances in any form that are prohibited, limited or regulated pursuant to any Environmental Law by virtue of their toxic or otherwise deleterious characteristics, including any petroleum or petroleum products, radioactive materials, explosives, asbestos in any form, urea formaldehyde, polychlorinated biphenyls, and radon gas.

"<u>Hedging Agreements</u>" shall mean any foreign exchange contracts, currency swap agreements, commodity agreements or other similar arrangements, or arrangements designed to protect against fluctuations in currency values or commodity prices.

"Indebtedness" shall mean, as to any Person, without duplication, (a) all indebtedness of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments, (c) the deferred purchase price of assets or services that in accordance with GAAP would be included as a liability on the balance sheet of such Person, (d) the drawn amount of all letters of credit issued for the account of such Person, (e) all Indebtedness of any other Person secured by any Lien on any property owned by such Person, whether or not such Indebtedness has been assumed by such Person, (f) the principal component of all Capitalized Lease Obligations of such Person, (g) all obligations of such Person under interest rate swap, cap or collar agreements, interest rate future or option contracts, currency swap agreements, currency future or option contracts, commodity price protection agreements or other commodity price hedging agreements and other similar agreements, (h) all obligations of such Person in respect of Disqualified Equity Interests and (i) without duplication, all Contingent Obligations of such Person, provided that Indebtedness shall not include (i) trade and other ordinary course payables and accrued expenses arising in the ordinary course of business, (ii) deferred or prepaid revenue, and (iii) purchase price holdbacks in respect of a portion of the purchase price of an asset to satisfy warranty or other unperformed obligations of the respective seller. The amount of Indebtedness of any Person for purposes of clause (e) shall be deemed to be equal to the lesser of (i) the aggregate unpaid amount of such Indebtedness and (ii) the fair market value of the property encumbered thereby as determined by such Person in good faith. The indebtedness of any Person shall include the Indebtedness of any other Person (including any partnership in which such Person is a general partner) to the extent such Person is liable therefor as a result of such Person's ownership interest in or other relationship with such Person except to the extent the terms of such Indebtedness provide that such Person is not liable therefor. The amount of any Capitalized Lease Obligation as of any date shall be deemed to be the capitalized amount of the applicable capital lease that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP in respect thereof as of such date.

"Indemnified Person" shall have the meaning provided in Section 14.01(c).

"Initial Escrowed DIP Loans" shall mean the aggregate principal amount of the Loans equal to \$30,000,000 funded on the Initial Funding Date pursuant to Section 2.01(a).

"<u>Initial Funding Date</u>" shall mean the date occurring on or after the Effective Date on which the Initial Escrowed DIP Loans are funded pursuant to Section 2.01(a).

"Intercompany Loans" shall have the meaning provided in Section 10.05(g).

"Intercompany Note" shall mean a promissory note evidencing Intercompany Loans, duly executed and delivered substantially in the form of Exhibit F (or such other form as shall be satisfactory to the Administrative Agent in its sole discretion), with blanks completed in conformity herewith.

"Investments" shall have the meaning provided in Section 10.05.

"IP Rights" shall have the meaning set forth in Section 8.18.

"<u>Leaseholds</u>" of any Person shall mean all the right, title and interest of such Person as lessee or licensee in, to and under leases or licenses of land, improvements and/or fixtures.

"<u>Lender</u>" shall mean each financial institution listed on <u>Schedule 1.01(a)</u>, as well as any Person that becomes a "Lender" hereunder pursuant to <u>Section 14.04(b)</u>.

"Lender Default" shall mean, as to any Lender, (i) the wrongful refusal (which has not been retracted) of such Lender to make available its portion of any Commitment within three (3) Business Days of a Funding Date, (ii) such Lender having, or having a direct or indirect parent company that has, (x) become the subject of a proceeding under the Bankruptcy Code or has taken any action in furtherance of, or indicating its consent to, approval of or acquiescence in any such proceeding, (y) had appointed for it, or has taken any action in furtherance of, or indicating its consent to, approval of or acquiescence in any such appointment of, a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity, or (iii) such Lender having notified either the Borrower, the Administrative Agent and/or any Credit Party (x) that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect or (y) of the events described in preceding clause (ii); provided that, no Lender Default shall be deemed to have occurred solely by virtue of the ownership or acquisition of an equity interest in such Lender or a parent company thereof by a governmental authority or an instrumentality thereof.

"<u>Lien</u>" shall mean any mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance, lien (statutory or other), preference, priority or other security agreement of any kind or nature whatsoever (including, without limitation, any conditional sale or other title retention agreement, any financing or similar statement or notice filed under the UCC or any other similar recording or notice statute, and any lease having substantially the same effect as any of the foregoing).

"Loan" shall have the meaning provided in Section 2.01.

"Margin Stock" shall have the meaning provided in Regulation U.

"Material Adverse Effect" shall mean a material adverse effect on (i) the business, operations, property, assets, liabilities or condition (financial or otherwise) of the Borrower and its Subsidiaries taken as a whole, other than any change, event or occurrence, arising individually or in the aggregate, from (a) events leading up to the commencement of the Chapter 11 Cases and (b) events that would reasonably be expected to result from the filing, commencement or

administration of proceedings under Chapter 11 of the Bankruptcy Code, (ii) the ability of the Lenders, the Administrative Agent or the Collateral Agent to enforce their rights or remedies hereunder or under any other Credit Document or (iii) the ability of the Credit Parties, taken as a whole, to perform its obligations to the Lenders, the Administrative Agent or the Collateral Agent hereunder or under any other Credit Document.

"Material Contract" shall mean any contract or other arrangement which if terminated could reasonably be expected to have a Material Adverse Effect.

"Maturity Date" shall mean the earliest to occur of (a) February 29, 2016, which date may be extended with the consent of the Required Lenders and shall be extended automatically to match the Outside Date (under and as defined in the Asset Purchase Agreement) if such Outside Date is extended beyond February 29, 2016; provided that in no event shall the Maturity Date be extended, either with the consent of the Required Lenders or automatically, to a date that is later than March 30, 2016, (b) the acceleration of the Loans and termination of the Commitments pursuant to Section 11, (c) the consummation of any sale or other disposition of all or substantially all of the assets of the Debtors (including the consummation of the transactions contemplated by the Asset Purchase Agreement), and (d) the effective date of any chapter 11 plan of the Borrower or any other Debtor.

"Maximum Rate" shall have the meaning provided in Section 14.18.

"Moody's" shall mean Moody's Investors Service, Inc.

"Multiemployer Plan" shall mean any multiemployer plan as defined in Section 4001(a)(3) of ERISA, to which contributions are or within the immediately preceding five year period have been made (or have been required to be made) by the Borrower or any ERISA Affiliate.

"NAIC" shall mean the National Association of Insurance Commissioners.

"Net Cash Proceeds" shall mean the gross cash proceeds (including any cash received by way of deferred payment pursuant to a promissory note, receivable or otherwise, but only as and when received) received from an Extraordinary Receipt or the events described in Section 5.02(a) or (b) of this Agreement, net of reasonable transaction costs (including, as applicable, any underwriting, brokerage or other customary commissions and reasonable legal, advisory and other fees and expenses associated therewith) received from any such event.

"Net Sale Proceeds" shall mean for any Asset Sale, (a) the gross cash proceeds (including any cash received by way of deferred payment pursuant to a promissory note, receivable or otherwise, but only as and when received) received from such sale or other disposition of assets, less (b) the sum of (i) the amount if any, of all taxes paid or estimated to be payable in connection with such sale or other disposition of assets, (ii) the amount of any reasonable reserve established in accordance with GAAP against any liabilities (other than any taxes deducted pursuant to clause (i) above) (x) associated with the assets that are the subject to such sale or other disposition of assets and (y) retained by the Borrower or any of its Subsidiaries, provided that the amount of any subsequent reduction of such reserve (other than in connection with a payment in respect of any such liability) shall be deemed to be Net Sale Proceeds of such sale or

other disposition of assets occurring on the date of such reduction, (iii) the amount of any Indebtedness (other than Indebtedness of the Lenders pursuant to this Agreement) secured by a Lien on the assets that are the subject of such sale or other disposition of assets to the extent that the instrument creating or evidencing such Indebtedness requires that such Indebtedness be repaid upon consummation of such sale or other disposition of assets, (iv) in the case of any sale or other disposition of assets by a non-wholly-owned Subsidiary, the *pro rata* portion of the Net Proceeds thereof (calculated without regard to this clause (iv)) attributable to minority interests and not available for distribution to or for the account of the Borrower or a wholly-owned Subsidiary as a result thereof, and (v) reasonable and customary fees paid by the Borrower or any of its Subsidiaries in connection with any of the foregoing, in each case, only to the extent not already deducted in arriving at the amount referred to in clause (a) above.

"Non-Defaulting Lender" shall mean and include each Lender other than a Defaulting Lender.

"Non-Wholly Owned Subsidiary" shall mean, as to any Person, each Subsidiary of such Person which is not a Wholly-Owned Subsidiary of such Person.

"Note" shall have the meaning provided in Section 2.05(a).

"Notice of Withdrawal" shall have the meaning provided in Section 2.03.

"Notice Office" shall mean the office of the Administrative Agent located at 1615 Brett Road, New Castle, DE 19720 (Attn: [•]; Fax: [•]; Email: [•]) or such other office or person as the Administrative Agent may hereafter designate in writing as such to the other parties hereto.

"Obligations" shall mean all amounts owing to the Administrative Agent, the Collateral Agent or any Lender pursuant to the terms of this Agreement or any other Credit Document (including all interest which accrues after the commencement of any case or proceeding in bankruptcy after the insolvency of, or for the reorganization of the Borrower or any of its Subsidiaries, whether or not allowed in such case or proceeding), including, but not limited to, all fees, expenses (including fees and expenses of counsel, advisors or consultants retained by each of the Commitment Parties, the Administrative Agent and the Collateral Agent), indemnification obligations and whether primary, secondary, direct, indirect, absolute, contingent, fixed or otherwise, now existing or hereafter arising (including obligations of performance) and other obligations to the extent payable hereunder.

"Organization Documents" shall mean, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; and (c) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization of such entity.

- "Payment Office" shall mean the office of the Administrative Agent located at 1615 Brett Road, New Castle, DE 19720 (Attn: [●]; Fax: [●]; Email: [●]) or such other office as the Administrative Agent may hereafter designate in writing as such to the other parties hereto.
  - "PBGC" shall mean the U.S. Pension Benefit Guaranty Corporation.
  - "Permitted Liens" shall have the meaning provided in Section 10.01.
- "<u>Person</u>" shall mean any individual, partnership, joint venture, firm, corporation, association, limited liability company, trust or other enterprise or any Governmental Authority.
  - "Petition Date" shall have the meaning provided in the introductory paragraph hereto.
- "Plan" shall mean an "employee pension benefit plan" as defined in Section 3(2) of ERISA (other than a Multiemployer Plan) subject to the provisions of Title IV of ERISA or Section 412 of the Code or Section 302 of ERISA maintained or contributed to by the Borrower or with respect to which the Borrower has any liability (including on account of an ERISA affiliate).
  - "Platform" shall have the meaning provided in Section 9.01(j).
- "<u>Prepetition Administrative Agent</u>" shall mean the administrative agent under the Prepetition Credit Agreement.
  - "Prepetition Collateral" shall have the meaning provided in the Final DIP Order.
- "Prepetition Credit Agreement" shall mean that certain Credit Agreement, dated as of April 1, 2011, by and among, *inter alios*, the Borrower, Western Coal Corp., Walter Energy Canada Holdings, Inc., the lenders from time to time party thereto, and Morgan Stanley Senior Funding, Inc., as administrative agent.
  - "Prepetition Debt Documents" shall have the meaning provided in the Final DIP Order.
- "<u>Prepetition First Lien Secured Parties</u>" shall mean the "First Lien Secured Parties" as defined in the Final DIP Order.
  - "Prepetition Liens" shall have the meaning provided in the Cash Collateral Order.
  - "Prepetition Obligations" shall have the meaning provided in the Cash Collateral Order.
  - "Prepetition Secured Parties" shall have the meaning provided in the Final DIP Order.
- "<u>Prepetition Trustee</u>" shall mean the "First Lien Trustee" as defined in the Final DIP Order.
  - "Projections" shall have the meaning provided in Section 8.05(c).
  - "Public Lender" shall have the meaning provided in Section 9.01(j).

"Qualified Equity Interest" shall mean any Equity Interest that does not constitute a Disqualified Equity Interest.

"Real Property" of any Person shall mean all the right, title and interest of such Person in and to land, improvements and fixtures, including Leaseholds.

"Recovery Event" shall mean any event that gives rise to the receipt by the Borrower or any of its Subsidiaries of any cash insurance proceeds or condemnation awards payable (i) by reason of theft, loss, physical destruction, damage, taking or any other similar event with respect to any property or assets of the Borrower or any of its Subsidiaries and (ii) under any policy of insurance required to be maintained under Section 9.03.

"Register" shall have the meaning provided in Section 14.04(c).

"Regulation D" shall mean Regulation D of the Securities Act as from time to time in effect and any successor to all or a portion thereof.

"Regulation T" shall mean Regulation T of the Board of Governors of the Federal Reserve System as from time to time in effect and any successor to all or a portion thereof.

"Regulation U" shall mean Regulation U of the Board of Governors of the Federal Reserve System as from time to time in effect and any successor to all or a portion thereof.

"<u>Related Parties</u>" shall mean, with respect to any Person, such Person's Affiliates and the members, partners, directors, officers, employees, controlling persons, agents, trustees, representatives, attorneys, advisors, successors and assigns of such Person and of such Person's Affiliates.

"Release" shall mean disposing, discharging, injecting, spilling, pumping, leaking, leaching, dumping, emitting, escaping, emptying, pouring, seeping, or migrating into, through or upon any land or water or air, or otherwise entering into the environment.

"Reportable Event" shall mean an event described in Section 4043(c) of ERISA with respect to a Plan that is subject to Title IV of ERISA other than those events as to which the 30-day notice period is waived under applicable regulations.

"Required Lenders" shall mean, at any time, at least four (4) unaffiliated Lenders who are Commitment Parties or their Affiliates holding at least 55% of the aggregate outstanding amount of the Loans and Commitments held by all Commitment Parties and their Affiliates on such date, provided that the Loans of any Defaulting Lender shall be disregarded in determining Required Lenders at any time.

"Returns" shall have the meaning provided in Section 8.09.

"S&P" shall mean Standard & Poor's Ratings Services, a division of McGraw-Hill, Inc.

"Sale Motion" shall have the meaning provided to the term "Motion" in the Sale Order.

"Sale Order" shall mean the Order (I) Approving the Sale of the Acquired Assets Free and Clear of Claims, Liens, Interests and Encumbrances; (II) Approving the Assumption and Assignment of Certain Executory Contracts and Unexpired Leases; and (III) Granting Related Relief [Docket No. 1584], entered by the Bankruptcy Court on January 8, 2016.

"SEC" shall have the meaning provided in Section 9.01(h).

"Second Escrowed DIP Loan Funding Date" shall mean the date occurring after the Initial Funding Date on which the Second Escrowed DIP Loans are funded pursuant to Section 2.01(b).

"Second Escrowed DIP Loans" shall mean the aggregate principal amount of the Loans equal to \$20,000,000 funded on the Second Escrowed DIP Loan Funding Date pursuant to Section 2.01(b).

"Secured Creditors" shall mean the Commitment Parties, the Lenders, the Administrative Agent, the Collateral Agent, and their respective successors and permitted assigns.

"Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Security Document" shall mean and include the Final DIP Order and, after the execution and delivery thereof, each Additional Security Document and all other security documents hereafter delivered to the Collateral Agent granting a Lien on any property of any Person for the benefit of the Secured Creditors.

"Steering Committee" shall have the meaning provided in the Cash Collateral Order.

"Steering Committee Advisors" shall have the meaning provided in the Final DIP Order.

"Subsidiary" shall mean, as to any Person, (i) any corporation more than 50% of whose stock of any class or classes having by the terms thereof ordinary voting power to elect a majority of the directors of such corporation (irrespective of whether or not at the time stock of any class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time owned by such Person and/or one or more Subsidiaries of such Person and (ii) any partnership, limited liability company, association, joint venture or other entity in which such Person and/or one or more Subsidiaries of such Person has more than a 50% equity interest at the time. Unless otherwise qualified, all references to a "Subsidiary" or to "Subsidiaries" in this Agreement shall refer to a Subsidiary or Subsidiaries of the Borrower.

"Tax Compliance Certificate" shall have the meaning provided in Section 5.05(b).

"Taxes" shall have the meaning provided in Section 5.05(a).

"<u>Total Commitment</u>" shall mean, at any time, the sum of the Commitments of each of the Lenders at such time.

"Trading With the Enemy Act" shall have the meaning provided in Section 8.19.

"<u>Transaction</u>" shall mean, collectively, (i) the execution, delivery and performance by each Credit Party of the Credit Documents to which it is a party, (ii) the funding of the Loans to the Escrow Account on any Funding Date, (iii) the Withdrawal of Loans from time to time during the Availability Period and the use of proceeds thereof, and (iv) the payment of all fees and expenses in connection with the foregoing.

"<u>UCC</u>" shall mean the Uniform Commercial Code as from time to time in effect in the relevant jurisdiction.

"<u>Unfunded Pension Liability</u>" shall mean the excess of a Plan's benefit liabilities under Section 4001(a)(16) of ERISA, over the fair market value of that Plan's assets (excluding any accrued but unpaid contributions), determined as of the beginning of the most recent plan year in accordance with the assumption used for funding the Plan pursuant to Sections 412 and 430 of the Code for the Plan year.

"United States" and "U.S." shall each mean the United States of America.

"Wholly-Owned Domestic Subsidiary" shall mean, as to any Person, any Wholly-Owned Subsidiary of such Person which is a Domestic Subsidiary.

"Wholly-Owned Subsidiary" shall mean, as to any Person, (i) any corporation 100% of whose capital stock is at the time owned by such Person and/or one or more Wholly-Owned Subsidiaries of such Person and (ii) any partnership, limited liability company, association, joint venture or other entity in which such Person and/or one or more Wholly-Owned Subsidiaries of such Person has a 100% equity interest at such time (other than, in the case of a Foreign Subsidiary of the Borrower with respect to the preceding clauses (i) and (ii), director's qualifying shares and/or other nominal amount of shares required to be held by Persons other than the Borrower and its Subsidiaries under applicable law).

"Withdrawal" shall mean any withdrawal by the Borrower of the Initial Escrowed DIP Loans or the Second Escrowed DIP Loans, as applicable, from the Escrow Account.

"Withdrawal Date" means the date of any Withdrawal.

- 1.02 <u>Other Definitional Provisions</u>. Unless otherwise specified therein, all terms defined in this Agreement shall have the defined meanings when used in the other Credit Documents or any certificate or other document made or delivered pursuant hereto or thereto.
- (a) As used herein and in the other Credit Documents, and any certificate or other document made or delivered pursuant hereto or thereto, (i) accounting terms not defined in Section 1.01 shall have the respective meanings given to them under GAAP, (ii) the words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation", (iii) the word "incur" shall be construed to mean incur, create, issue, assume, become liable in respect of or suffer to exist (and the words "incurred" and "incurrence" shall have correlative meanings), (iv) unless the context otherwise requires, the words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all

tangible and intangible assets and properties, including cash, Equity Interests, securities, revenues, accounts, leasehold interests and contract rights, (v) the word "will" shall be construed to have the same meaning and effect as the word "shall", and (vi) unless the context otherwise requires, any reference herein (A) to any Person shall be construed to include such Person's successors and assigns and (B) to the Borrower or any other Credit Party shall be construed to include the Borrower or such Credit Party as debtor and debtor-in-possession and any receiver or trustee for the Borrower or any other Credit Party, as the case may be, in any insolvency or liquidation proceeding.

- (b) The words "hereof", "herein" and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section, Schedule and Exhibit references are to this Agreement unless otherwise specified. The reference herein to any Schedules to the Asset Purchase Agreement shall refer to those schedules as updated from time to time in accordance with the terms thereof.
- (c) The meanings given to terms defined herein shall be equally applicable to both the singular and plural forms of such terms.

#### SECTION 2 Amount and Terms of Credit.

- 2.01 <u>The Commitments</u>. Subject to and upon the terms and conditions set forth herein and the Final DIP Order, each Lender severally agrees to make a term loan or term loans denominated in Dollars (each, a "<u>Loan</u>" and, collectively, the "<u>Loans</u>") to the Borrower as follows:
- (a) on the Initial Funding Date, each Lender with a Commitment shall fund its pro rata portion of the Initial Escrowed DIP Loans; <u>provided</u> that, (i) the Borrower hereby directs the Lenders to fund the Initial Escrowed DIP Loans directly to the Escrow Account, and (ii) for the avoidance of doubt, interest on the Initial Escrowed DIP Loans shall accrue on the full amount of the Initial Escrowed DIP Loans commencing on the Initial Funding Date in accordance with <u>Section 2.08</u>;
- (b) on the Second Escrowed DIP Loan Funding Date, solely to the extent (i) the Availability Period is extended as a result of an extension of the Maturity Date as set forth in clause (a) of the definition thereof, (ii) less than \$2,500,000 of the proceeds of the Initial Escrowed DIP Loans remain in the Escrow Account, and (iii) the Administrative Agent has received a written request from the Borrower no later than five (5) Business Days prior to the proposed Second Escrowed DIP Loan Funding Date, each Lender with a Commitment shall fund its pro rata portion of the Second Escrowed DIP Loans, provided that, (x) the Borrower hereby directs the Lenders to fund the Second Escrowed DIP Loans directly to the Escrow Account, and (y) for the avoidance of doubt, interest on the Second Escrowed DIP Loans shall accrue on the full amount of the Second Escrowed DIP Loans commencing on the Second Escrowed DIP Loan Funding Date in accordance with Section 2.08;
- (c) subject to the conditions set forth in <u>Section 2.02</u>, (i) on or after the Initial Funding Date, Withdrawals of the proceeds of the Initial Escrowed DIP Loans from the Escrow

Account shall be permitted from time to time during the Availability Period and (ii) on or after the Second Escrowed DIP Loan Funding Date, Withdrawals of the proceeds of the Second Escrowed DIP Loans from the Escrow Account shall be permitted from time to time during the Availability Period; and

- (d) notwithstanding anything to the contrary, (i) any amount of the Loans funded into the Escrow Account hereunder will permanently reduce the Commitment, (ii) once repaid, the Loans incurred hereunder may not be re-borrowed, (iii) regardless of whether or not the Total Commitments are fully borrowed, the Commitments shall terminate at the end of the Availability Period and (iv) in no event shall the aggregate principal amount of Loans exceed \$50,000,000.
- 2.02 <u>Conditions to Each Withdrawal</u>. On each Withdrawal Date, (i) immediately after giving effect to each such proposed Withdrawal (and the use of proceeds thereof on any such Withdrawal Date), the aggregate amount of unrestricted cash and Cash Equivalents of the Borrower and the other Debtors shall not exceed \$40,000,000 in the aggregate, (ii) only one Withdrawal may occur on the same date, (iii) the amount of Withdrawal on each Withdrawal Date shall not be less than \$5,000,000 (or if less than \$5,000,000 the aggregate amount remaining in the Escrow Account), (iv) the Drawdown Fee shall have been paid on each Withdrawal Date from the proceeds of such Withdrawal, and (v) all conditions set forth in Section 7 shall have been satisfied.
- Notice of Withdrawal. The Borrower shall give the Administrative Agent at the Notice Office at least one (1) Business Day's prior notice of each proposed Withdrawal (each, a "Notice of Withdrawal"), other than with respect to the Withdrawal on the Effective Date, provided that (in each case) any such notice shall be deemed to have been given on a certain day only if given before 12:00 Noon (New York City time) on such day. Each Notice of Withdrawal (shall be irrevocable and shall be in writing, or by telephone promptly confirmed in writing, in the form of Exhibit A, appropriately completed to specify: (i) the aggregate principal amount of the Loans to be withdrawn pursuant to such Withdrawal (stated in Dollars), and (ii) the Withdrawal Date (which shall be a Business Day). The Administrative Agent shall promptly give each Lender notice of such proposed Withdrawal, of such Lender's proportionate share thereof and of the other matters required by the immediately preceding sentence to be specified in the Notice of Withdrawal.
- Eucling Date, each Lender with a Commitment will make available its *pro rata* portion (determined in accordance with Section 2.07) of the Initial Escrowed DIP Loans or the Second Escrowed DIP Loans, as applicable. All such amounts will be made available in Dollars, and in immediately available funds at the Payment Office or directly to the Escrow Account. Unless the Administrative Agent shall have been notified in writing by any Lender prior to the applicable Funding Date that such Lender does not intend to make available to the Administrative Agent such Lender's portion of the Loans to be made on such date, the Administrative Agent may assume that such Lender has made such amount available to the Administrative Agent on such Funding Date and the Administrative Agent may at the direction of the Required Lenders (but shall not be obligated to), in reliance upon such assumption, fund the Escrow Account a corresponding amount. If such corresponding amount is not in fact made available to the

Administrative Agent by such Lender, the Administrative Agent shall be entitled to recover such corresponding amount on demand from such Lender. If such Lender does not pay such corresponding amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent shall promptly notify the Borrower and the Borrower shall immediately pay such corresponding amount to the Administrative Agent. The Administrative Agent also shall be entitled to recover on demand from such Lender or the Borrower interest on such corresponding amount in respect of each day from the date such corresponding amount was made available by the Administrative Agent to the Borrower until the date such corresponding amount is recovered by the Administrative Agent, at a rate per annum equal to (i) if recovered from such Lender, the overnight Federal Funds Rate for the first three days and at the interest rate otherwise applicable to such Loans for each day thereafter and (ii) if recovered from the Borrower, the rate of interest applicable to the respective Loan, as determined pursuant to Section 2.08. Nothing in this Section 2.04 shall be deemed to relieve any Lender from its obligation to make Loans hereunder or to prejudice any rights which the Borrower may have against any Lender as a result of any failure by such Lender to make Loans hereunder.

#### 2.05 Notes.

- (a) Borrower's obligation to pay the principal of, and interest on, the Loans made by each Lender shall be evidenced in the Register maintained by the Administrative Agent pursuant to Section 14.04(c) and shall, if requested by such Lender, also be evidenced by a promissory note duly executed and delivered by the Borrower substantially in the form of Exhibit B, with blanks appropriately completed in conformity herewith (each, a "Note" and, collectively, the "Notes").
- (b) Each Lender will note on its internal records the amount of each Loan made by it and each payment in respect thereof and, to the extent applicable, prior to any transfer of any of its Notes will endorse on the reverse side thereof the outstanding principal amount of Loans evidenced thereby. Failure to make any such notation or any error in such notation shall not affect the Borrower's obligations in respect of such Loans.
- (c) Notwithstanding anything to the contrary contained above in this Section 2.05 or elsewhere in this Agreement, Notes shall only be delivered to Lenders which at any time specifically request the delivery of such Notes. No failure of any Lender to request or obtain a Note evidencing its Loans to the Borrower shall affect or in any manner impair the obligations of the Borrower to pay the Loans (and all related Obligations) incurred by such Borrower which would otherwise be evidenced thereby in accordance with the requirements of this Agreement, and shall not in any way affect the security or guaranties therefor provided pursuant to the various Credit Documents. Any Lender which does not have a Note evidencing its outstanding Loans shall in no event be required to make the notations otherwise described in preceding clause (b). At any time when any Lender requests the delivery of a Note to evidence any of its Loans, the respective Borrower shall promptly execute and deliver to the respective Lender the requested Note in the appropriate amount or amounts to evidence such Loans.

#### 2.06 [Reserved].

2.07 <u>Pro rata Fundings</u>. All funding of Loans under this Agreement shall be incurred from the Lenders *pro rata* on the basis of their Commitments. It is understood that no Lender shall be responsible for any default by any other Lender of its obligation to make Loans hereunder and that each Lender shall be obligated to make the Loans provided to be made by it hereunder, regardless of the failure of any other Lender to make its Loans hereunder.

## 2.08 Interest.

- (a) Subject to the provisions of <u>Section 2.08(b)</u>, Loans shall bear interest on the outstanding principal amount thereof (including any Fees or other amounts due and unpaid hereunder) at a rate per annum equal to 12% per annum payable in cash on the Maturity Date.
- (b) Upon the occurrence and during the continuance of an Event of Default, the principal amount of all Loans outstanding and, to the extent permitted by applicable law, any interest payments on the Loans or any Fees or other amounts due and unpaid under any Credit Document, shall thereafter bear interest at the Default Rate. Payment or acceptance of the increased rates of interest provided for in this Section 2.08 is not a permitted alternative to timely payment and shall not constitute a waiver of any Event of Default or otherwise prejudice or limit any rights or remedies of any Agent or any Lender. Any accrued and unpaid interest pursuant to this clause (b) shall be payable upon demand in cash.
- (c) Accrued and unpaid interest shall be payable in cash in respect of each Loan on (x) the date of any repayment or prepayment thereof (on the amount prepaid or repaid), and (y) the Maturity Date (whether by acceleration or otherwise) and thereafter, on demand.

#### 2.09 [Reserved].

## 2.10 <u>Increased Costs</u>

If any Lender determines that after the Effective Date the introduction of or any change in any applicable law or governmental rule, regulation, order, guideline, directive or request (whether or not having the force of law) concerning capital adequacy, or any change in interpretation or administration thereof by the NAIC or any Governmental Authority, central bank or comparable agency, will have the effect of increasing the amount of capital required or expected to be maintained by such Lender or any corporation controlling such Lender based on the existence of such Lender's Commitments hereunder or its obligations hereunder, then the Borrower agrees to pay to such Lender, upon its written demand therefor, such additional amounts as shall be required to compensate such Lender or such other corporation for the increased cost to such Lender or such other corporation or the reduction in the rate of return to such Lender or such other corporation as a result of such increase of capital. In determining such additional amounts, each Lender will act reasonably and in good faith and will use averaging and attribution methods which are reasonable, provided that such Lender's determination of compensation owing under this Section 2.10(a) shall, absent manifest error, be final and conclusive and binding on all the parties hereto. Each Lender, upon determining that any additional amounts will be payable pursuant to this Section 2.10(a), will give prompt written notice thereof to the Borrower, which notice shall show in reasonable detail the basis for calculation of such additional amounts.

- (b) Notwithstanding anything in this Agreement to the contrary, the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines, requirements and directives thereunder, issued in connection therewith or in implementation thereof, shall be deemed to be a change after the Effective Date in a requirement of law or government rule, regulation or order, regardless of the date enacted, adopted, issued or implemented (including for purposes of this <u>Section 2.10</u>).
- (c) Failure or delay on the part of any Lender to demand compensation pursuant to this Section 2.10 shall not constitute a waiver of such Lender's right to demand such compensation; provided that Borrower shall not be required to compensate a Lender pursuant to this Section 2.10 for any increased costs or reductions incurred more than 180-days prior to the date that such Lender notifies the Borrower of the circumstances giving rise to such increased costs or reductions and of such Lender's intention to claim compensation therefor; provided further that if the circumstances giving rise to such increased costs or reductions are retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof.
  - 2.11 [Reserved].
  - 2.12 [Reserved].

SECTION 3 [Reserved].

SECTION 4 Fees.

- 4.01 <u>Fees</u>. The Borrower agrees to pay the fees set forth below. All such fees shall be fully earned when paid and shall not be refundable for any reason whatsoever.
- (a) <u>Put Option Premium</u>. The Borrower shall pay a put option premium (the "<u>Put Option Premium</u>") to each Commitment Party equal to 7.5% of such Commitment Party's Commitments as set forth in the Commitment Letter as of the date of such Commitment Letter. The Put Option Premium shall be fully earned as of the date of the Final DIP Order and shall be paid in cash on the Effective Date from the proceeds of the Loans.
- (b) <u>Upfront Fee</u>. The Borrower shall pay a fee (the "<u>Upfront Fee</u>") to each Lender on the Effective Date equal to 7.5% of the Commitments of such Lender as of the Effective Date. The Upfront Fee shall be fully earned as of the Effective Date and shall be paid in cash on the Effective Date from the proceeds of the Loans.
- (c) <u>Drawdown Fee.</u> On each Withdrawal Date, the Borrower shall pay to each Lender, ratably based on each Lender's respective principal amount of Loans provided with respect to the proceeds withdrawn on such Withdrawal Date, a fee (the "<u>Drawdown Fee</u>") equal to 2.5% of the amount of the Loans provided by such Lender with respect to the proceeds withdrawn on such Withdrawal Date. The Drawdown Fee shall be paid in cash on the date of each such Withdrawal (it being understood that such Drawdown Fee shall be payable in respect of the gross principal amount of the Loans corresponding to the amount actually withdrawn by the Borrower from the Escrow Account, rather than the net proceeds of the Loans actually received by the Borrower pursuant to the applicable Withdrawal). The Lenders, Commitment

Parties and the Credit Parties hereto agree to treat, for federal income tax purposes, (a) the Upfront Fee and the Drawdown Fee as interest and (b) the Put Option Premium as a remuneration to the Commitment Parties for their agreeing to enter into the put option. The Lenders, Commitment Parties and the Credit Parties shall not take any position or action inconsistent with such treatment and/or characterization.

(d) The Borrower shall pay all fees set forth in the Agent Fee Letter (the "<u>Agency Fees</u>", and together with the Drawdown Fee, the Put Option Premium and the Upfront Fee, collectively, the "<u>Fees</u>") at the times when such payments are due and payable.

## SECTION 5 Prepayments; Payments; Taxes.

5.01 <u>Voluntary Prepayments</u>. The Borrower shall have the right to prepay its Loans, without premium or penalty, in whole or in part at any time and from time to time on the following terms and conditions: (i) the Borrower shall give the Administrative Agent prior to 12:00 Noon (New York City time) at the Notice Office at least one (1) Business Day's prior written notice (or telephonic notice promptly confirmed in writing) of its intent to prepay Loans, which notice (in each case) shall specify the amount of such prepayment to be prepaid, and which notice the Administrative Agent shall promptly transmit to each of the Lenders; (ii) each partial prepayment of Loans pursuant to this <u>Section 5.01</u> shall be in an aggregate principal amount of at least \$1,000,000 (or such lesser amount as is acceptable to the Administrative Agent in any given case); (iii) if such notice is given by the Borrower, the Borrower shall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein; and (iv) any prepayment of a Loan shall be accompanied by all accrued interest on the amount prepaid.

#### 5.02 Mandatory Prepayments.

- (a) On each date on or after the Effective Date upon which the Borrower or any of its Domestic Subsidiaries receives any cash proceeds from any issuance, incurrence or sale by the Borrower or any of its Subsidiaries of Indebtedness (other than Indebtedness for borrowed money permitted to be incurred pursuant to Section 10.04), an amount equal to 100% of the Net Cash Proceeds of the respective issuance, incurrence or sale of such Indebtedness shall be applied on such date as a mandatory prepayment in accordance with the requirements of Section 5.02(e);
- (b) On each date on or after the Effective Date upon which the Borrower or any of its Subsidiaries receives any cash proceeds from a capital contribution to, or the issuance of any Equity Interests of, the Borrower or any of its Subsidiaries, an amount equal to 100% of the Net Cash Proceeds of the respective capital contribution or issuance of Equity Interests shall be applied on such date as a mandatory prepayment in accordance with the requirements of Section 5.02(e).
- (c) Subject to the Final DIP Order, on each date on or after the Effective Date upon which the Borrower or any of its Domestic Subsidiaries receives any cash proceeds from any Asset Sale, an amount equal to 100% of the Net Sale Proceeds such Asset Sale shall be

applied on such date as a mandatory prepayment in accordance with the requirements of Section 5.02(e).

- (d) Within one (1) Business Day of each date on or after the Effective Date upon which the Borrower or any of its Domestic Subsidiaries receives any cash proceeds from any Extraordinary Receipt, an amount equal to 100% of the Net Cash Proceeds from such Extraordinary Receipts shall be applied on such date as a mandatory repayment in accordance with the requirements of Section 5.02(e), provided that any portion of the casualty insurance proceeds received with respect to any Acquired Assets (as defined in the Asset Purchase Agreement) and intended to be applied (or are applied) to the repairs or restoration of such Acquired Assets (or any other insurance proceeds to the extent the Required Lenders, in their sole discretion, have consented to such use) shall not be deemed Net Cash Proceeds subject to the mandatory prepayment requirements of this Section 5.02(d).
- (e) Each amount required to be applied pursuant to <u>Sections 5.01</u> and <u>Sections 5.02(a)</u>, (b), (c) and (d) shall first be applied to reduce any interest then accrued and unpaid on the principal so prepaid. To the extent that the Net Cash Proceeds or the Net Sale Proceeds required to prepay the Loans pursuant to clauses (a) through (d) above exceed the amount of Loans outstanding, the unused Commitments will be reduced by an amount equal to such excess. Each prepayment or repayment of the principal of the Loans and reduction of Commitments shall be applied on a pro rata basis to the outstanding Loans or Commitments, as applicable. For the avoidance of doubt, Loans that are optionally or mandatorily prepaid or repaid may not be reborrowed.
- 5.03 <u>Repayment of Loans</u>. The Borrower shall repay on the Maturity Date to the Administrative Agent for the benefit of the Lenders the aggregate principal amount of all Loans and all other Obligations outstanding as of the Maturity Date (which amounts shall be reduced as a result of the application of prepayments in accordance with <u>Section 5.01</u> and <u>Section 5.02</u>) in full in cash.
- 5.04 Method and Place of Payment. Except as otherwise specifically provided herein, all payments under this Agreement and under any Note shall be made to the Administrative Agent for the account of the Lender or Lenders entitled thereto not later than 12:00 Noon (New York City time) on the date when due and shall be made in Dollars in immediately available funds at the Payment Office. Whenever any payment to be made hereunder shall be stated to be due on a day which is not a Business Day, the due date thereof shall be extended to the next succeeding Business Day and, with respect to payments of principal, interest shall be payable at the applicable rate during such extension.

## 5.05 Net Payments.

(a) All payments made by the Borrower hereunder will be made without setoff, counterclaim or other defense, unless so provided under this Agreement or any other Credit Document. Except as provided in Section 5.05(b), 13.04(a) and (b) or as required by applicable law or regulation or the administration or interpretation thereof, all payments made by the Borrower hereunder will be made free and clear of, and without deduction or withholding for, any present or future taxes, levies, imposts, duties, fees, assessments or other charges of

whatever nature now or hereafter imposed by any jurisdiction or by any political subdivision or taxing authority thereof or therein with respect to such payments (but excluding (i) any tax (x) imposed on or measured by the net income, net profits or capital of the recipient pursuant to the laws of the jurisdiction in which it is organized or the jurisdiction in which its principal office or, in the case of a Lender, applicable lending office is located or any subdivision thereof or therein, or (y) imposed by reason of the recipient being a resident, domiciliary or national of, being engaged in business in or maintaining a permanent establishment or other physical presence in, or otherwise having some connection with the jurisdiction imposing the tax unless such connection results solely from such recipient's executing, enforcing, delivering, becoming a party to, or performing its obligations or receiving a payment under or enforcing this Agreement or any other Credit Document, and (ii) any withholding tax that would not have been imposed but for a failure by such recipient (or any financial institution through which any payment is made to such recipient) to comply with the applicable requirements of FATCA or is attributable to such recipient's failure to comply with Section 5.05(b), any withholding tax resulting from any law in effect (including FATCA) on the date such recipient becomes a party to this Agreement or designates a new lending office, except to the extent such recipient (or its assignor, if any) was entitled, at the time of designation of a new lending office (or assignment), to receive additional amounts with respect to such withholding tax pursuant to Section 5.05(a) hereto, and all interest or penalties with respect to such non-excluded taxes, levies, imposts, duties, fees, assessments or other charges (all such non-excluded taxes, levies, imposts, duties, fees, assessments or other charges being referred to collectively as "Taxes")). If any Taxes are so levied or imposed in connection with any payment made by a Borrower hereunder, the respective Borrower agrees to pay the full amount of such Taxes to the appropriate taxing authority, and shall pay to the relevant person such additional amounts as may be necessary so that such payment, after withholding or deduction for or on account of such Taxes, will not be less than the amount such person would have received absent such withholding or deduction. The respective Borrower will furnish to the Administrative Agent within 45 days after the date the payment of any Taxes is due pursuant to applicable law certified copies of tax receipts issued by the applicable taxing authority evidencing such payment by such Borrower, or such other evidence satisfactory to the Administrative Agent. The Borrower agrees to indemnify and hold harmless each Lender, and reimburse such Lender upon its written request, such request stating the amount of Taxes so paid in connection with any payment made by a Borrower hereunder and describing, in reasonable detail, the basis for such indemnification claim, for the amount of any Taxes so levied or imposed in connection with any payment made by a Borrower hereunder and paid by such Lender (including penalties and interest). Such indemnity shall survive indefinitely following the termination of this Agreement or the settlement of the Notes, as applicable. Such payment shall be made within 30 days after the date the Lender makes a written demand therefor.

(b) Each Lender that is not a United States person (as such term is defined in Section 7701(a)(30) of the Code) (each, a "Foreign Lender") for U.S. Federal income tax purposes agrees to deliver to the Borrower and the Administrative Agent on or prior to the Effective Date (i) two accurate and complete original signed copies of Internal Revenue Service Form W-8ECI, Form W-8BEN or Form W-8BEN-E, as applicable (with respect to a complete exemption under an income tax treaty) (or successor forms) certifying to such Lender's entitlement as of such date to a complete exemption from United States withholding tax with respect to payments to be made under any Credit Document, (ii) if the Lender is not a "bank"

within the meaning of Section 881(c)(3)(A) of the Code and is not delivering forms pursuant to clause (i) above, (x) a certificate substantially in the form of Exhibit C (any such certificate, a "Tax Compliance Certificate") and (y) two accurate and complete original signed copies of Internal Revenue Service Form W-8BEN or Form W-8BEN-E, as applicable (with respect to the portfolio interest exemption) (or successor form) certifying to such Lender's entitlement as of such date to a complete exemption from United States withholding tax with respect to payments of interest to be made under any Credit Document, or (iii) in the case of a Foreign Lender that is not the beneficial owner of payments made under any Credit Document (including a partnership or a participating Lender) (1) two accurate and complete original signed copies of Internal Revenue Service Form W-8IMY on behalf of itself and (2) the relevant forms prescribed in clauses (i) and (ii) above, and any other relevant forms pursuant to FATCA, as applicable, that would be required of each such beneficial owner or partner of such partnership if such beneficial owner or partner were a Lender; provided, however, that if the Lender is a partnership and one or more of its partners are claiming the exemption for portfolio interest under Section 881(c) of the Code, such Lender may provide a Tax Compliance Certificate on behalf of such partners. In addition, if a payment made to a Lender under any Credit Document would be subject to U.S. Federal withholding Tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable), such Lender shall deliver to the Borrower and the Administrative Agent, at the time or times prescribed by law and at such time or times reasonably requested by the Borrower or the Administrative Agent, such documentation prescribed by applicable law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) to determine that such Lender has or has not complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. If a Lender is not a Foreign Lender, such Lender shall deliver to the Borrower and the Administrative Agent two accurate and complete original signed copies of Internal Revenue Service Form W-9 certifying that such Lender is exempt from U.S. Federal backup withholding tax. Furthermore, any recipient that is entitled to an exemption from, or reduction of, any applicable non-United States withholding tax with respect to any payments under any Credit Document shall deliver to the Borrower and the Administrative Agent, at the time or times reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation reasonably requested by the Borrower or the Administrative Agent as will permit such payments to be made without, or at a reduced rate of, such non-United States withholding. In addition, each Lender agrees that from time to time after the Effective Date, when a lapse in time or change in circumstances renders the previous certification obsolete or inaccurate in any material respect, such Lender will deliver to the Borrower and the Administrative Agent two new accurate and complete original signed copies of Internal Revenue Service Form W-8ECI, Form W-8BEN, Form W-8BEN-E, Form W-8IMY and/or Form W-9, as applicable, and a Tax Compliance Certificate and such other forms as may be required in order to confirm or establish the entitlement of such Lender to a continued exemption from or reduction in United States withholding tax with respect to payments under this Agreement, or such Lender shall immediately notify the Borrower and the Administrative Agent of its inability to deliver any such Form or Certificate, in which case such Lender shall not be required to deliver any such Form or Certificate pursuant to this <u>Section 5.05(b)</u>. Notwithstanding anything to the contrary contained in Section 5.05(a), but subject to Section 14.04(a) and (b) and the immediately succeeding sentence, the Borrower shall be entitled, to the extent it is required to do so by law or the

administration or official interpretation thereof, to deduct or withhold taxes imposed by any relevant taxing authority from interest, Fees or other amounts payable under any Credit Document to the extent that such Lender has not provided to the Borrower the relevant documentation and certification to establish a complete exemption from such deduction or withholding.

payment under any Credit Document determines, in its sole discretion, that it has received a refund of any Taxes as to which it has been indemnified by the Borrower or with respect to which the Borrower has paid additional amounts pursuant to this Section, it shall pay to the Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrower under this Section with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses of the relevant recipient, as the case may be, and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund), provided that such Borrower, upon the request of the relevant recipient, agrees to repay the amount paid over to such Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to the recipient in the event the recipient is required to repay such refund to such Governmental Authority. This subsection shall not be construed to require any such recipient to make available its tax returns (or any other information relating to its taxes that it deems confidential) to the Borrower or any other Person.

# SECTION 6 Conditions Precedent to Credit Events on the Initial Funding Date.

The effectiveness of this Agreement and the obligation of each Lender to make Loans on the Initial Funding Date is subject at the time of the making of such Loans to the satisfaction of the following conditions in a manner satisfactory to the Commitment Parties:

- 6.01 <u>Credit Agreement</u>. The Administrative Agent shall have received from each party hereto a counterpart of this Agreement signed on behalf of such party and in form and substance satisfactory to the Commitment Parties and the Administrative Agent.
- 6.02 Officer's Certificate. On the Effective Date, the Administrative Agent shall have received a closing certificate signed by an Authorized Officer of the Borrower dated as of the Effective Date, stating that (i) all representations and warranties set forth in this Agreement and the other Credit Documents are true and correct in all material respects (or in all respects with regards to representations and warranties already qualified by materiality) on and as of such date (except to the extent that any such representation or warranty speaks or is referenced to a particular date, in which case stating that such representation or warranty is true and correct as of such date), (ii) the Credit Parties are on such date in compliance with all the terms and provisions set forth in this Agreement and the other Credit Documents and (iii) on such date no Default or Event of Default has occurred and is continuing at the time of, or immediately after giving effect to, the making of the Loans on the Initial Funding Date.

#### 6.03 [Reserved].

6.04 <u>Secretary's Certificate; Organization Documents; etc.</u> On the Effective Date, the Administrative Agent shall have received a certificate of the secretary or assistant secretary (or

other equivalent officer, partner or manager) of each Credit Party, in form and substance satisfactory to the Commitment Parties dated as of the Effective Date, which shall certify (i) copies of resolutions, in form and substance satisfactory to the Commitment Parties in their sole discretion, of the board of directors (or other equivalent governing body, member or partner) of such Credit Party authorizing (1) the execution, delivery and performance of this Agreement and each other Credit Document to which such Credit Party is a party (including, in the case of the Borrower, authorization of the borrowing of the Loans), and (2) the granting by such Credit Party of the security interests in and liens upon the Collateral to secure all of the Obligations (and such certificate shall state that such resolutions have not been amended, modified, revoked or rescinded as of the date of such certificate), (ii) the incumbency and signature of the officers of such Credit Party authorized to execute this Agreement and the other Credit Documents to which such Credit Party is a party, (iii) copies of the Organization Documents of such Credit Party as in effect on such date, complete with all amendments thereto, and (iv) the good standing (or equivalent status to the extent applicable) of such Credit Party in its jurisdiction of organization, as evidenced by good standing certificate(s) (or the equivalent thereof issued by any applicable jurisdiction) dated not more than 30 days prior to the Effective Date, issued by the Secretary of State or other appropriate official of each such jurisdiction as attached to such certificate.

# 6.05 Adverse Change, Approvals.

- (a) Since January 14, 2016, there shall not have occurred a Material Adverse Effect or any event or occurrence that would reasonably be expected to have a Material Adverse Effect.
- (b) On or prior to the Effective Date, except for any approvals or consents, the failure of which to obtain could not reasonably be expected to have a Material Adverse Effect, all necessary governmental (domestic and foreign) and material third party approvals and/or consents in connection with the Transaction, the other transactions contemplated hereby and the granting of Liens under the Credit Documents shall have been obtained and remain in effect, and all applicable waiting periods with respect thereto shall have expired without any action being taken by any competent authority which restrains, prevents or imposes materially adverse conditions upon the consummation of the Transaction or the other transactions contemplated by the Credit Documents or otherwise referred to herein or therein.

## 6.06 [Reserved].

6.07 <u>UCC Filings</u>. If requested by the Administrative Agent (at the direction of the Required Lenders), the Borrower and each other Debtor shall have delivered proper financing statements (Form UCC-1 or the equivalent) fully completed for filing in the appropriate filing offices of each jurisdiction as may be necessary to perfect the security interests in such Collateral in which a security interest in which may be perfected by such a filing.

## 6.08 [Reserved].

6.09 <u>Insurance</u> On the Effective Date, the Administrative Agent shall have received evidence that all insurance required to be maintained pursuant to the Credit Documents has been

obtained and is in effect, together with the certificates of insurance and endorsements, naming the Administrative Agent, on behalf of the Lenders and/or the Collateral Agent on behalf of the Secured Creditors, as an additional insured or loss payee, as the case may be, under all insurance policies maintained with respect to the assets and properties of the Credit Parties;

- 6.10 <u>Fees.</u> On the Effective Date, the Borrower shall have paid (a) to the Administrative Agent (and its relevant Affiliates) (i) for account of the Commitment Parties, the Put Option Premium in accordance with <u>Section 4.01(a)</u>, (ii) for account of the Lenders, the Upfront Fee in accordance with <u>Section 4.01(b)</u>, and (iii) for the account of the Administrative Agent, all fees in accordance with <u>Section 4.01(d)</u>, and (b) all other costs, fees and expenses (including, without limitation, legal fees and expenses and the fees and expenses of any other advisors) and other compensation contemplated hereby payable to the Administrative Agent, Commitment Parties or such Lender to the extent then due.
- 6.11 <u>Patriot Act</u>. The Administrative Agent and the Lenders shall have received all documentation and other information requested by the Administrative Agent or the respective Lenders that is required by bank regulatory authorities under the applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act.

## 6.12 [Reserved].

- 6.13 <u>Approved Budget</u>. The Administrative Agent and the Lenders shall have received an operating budget (the "<u>Approved Budget</u>") projecting financial operations of the Debtors for the eleven-week period starting on the week ending January 23, 2016 and extending though the week in which the Outside Date occurs, which budget shall be in form and substance acceptable to the Commitment Parties in their sole discretion.
- 6.14 <u>Capital Expenditure Budget</u>. The Administrative Agent shall have received for distribution to the Lenders a budget (the "<u>Capital Expenditure Budget</u>") projecting capital expenditures of the Debtors and their Subsidiaries for period starting from January 1, 2016 through March 31, 2016 on a monthly basis, which budget shall be in form and substance acceptable to the Commitment Parties in their sole discretion.

## 6.15 Bankruptcy Matters.

- (a) The Final DIP Order shall have been entered by the Bankruptcy Court, shall be in full force and effect and shall not have been reversed, stayed, vacated, terminated, modified or amended, in each case, without the prior written consent of the Administrative Agent and the Commitment Parties, each in their sole discretion. The Final DIP Order shall grant to the Administrative Agent and the Lenders (i) superpriority claims pursuant to Section 364(c)(1) of the Bankruptcy Code and (ii) security interests in, and liens on, the Collateral securing the Obligations pursuant to Sections 361, 362, 364(c)(2), 364(c)(3) and 364(d)(1) of the Bankruptcy Code. The Debtors shall be in compliance in all respects with the Final DIP Order.
- (b) The Sale Order shall be in full force and effect and shall not have been reversed, stayed, vacated, terminated, modified or amended, in each case, without the prior written consent of the Commitment Parties in their sole discretion. The Debtors shall be in compliance in all respects with the Sale Order.

- (c) [reserved];
- (d) The Administrative Agent and the Lenders shall have a valid, binding, continuing, enforceable, fully-perfected and non-avoidable first priority senior priming lien on and security interest in the Collateral, which liens and security interests shall be senior to any and all other liens and security interests, other than (i) the Carve-Out, (ii) any Permitted Priority Liens as defined in the Cash Collateral Order, if any, and (iii) Approved Liens.
- (e) The Loans shall not cause the aggregate amount of the Loans to exceed the amount then authorized by the Final DIP Order to be borrowed.

## SECTION 7 Conditions Precedent to All Credit Events.

The obligation of each Lender to make Loans on any Funding Date, including any Loans to be made or deemed made on or after the Initial Funding Date, and the right of the Borrower to make any Withdrawal on any Withdrawal Date, is subject to the satisfaction of each of the conditions precedent set forth below:

- 7.01 No Default; Representations and Warranties. At the time of each such event and also after giving effect thereto (i) there shall exist no Default or Event of Default, (ii) no Material Adverse Effect has occurred since January 14, 2016, after giving pro forma effect to the transactions hereunder, (iii) the making of such Loans or the Withdrawal, as applicable, shall not violate any requirement of law and shall not be enjoined, temporarily, preliminarily or permanently, and (iv) all representations and warranties contained herein and in the other Credit Documents shall be true and correct in all material respects with the same effect as though such representations and warranties had been made on the date of such event (it being understood and agreed that (x) any representation or warranty which by its terms is made as of a specified date shall be required to be true and correct in all material respects only as of such specified date and (y) any representation or warranty that is qualified as to "materiality," "Material Adverse Effect" or similar language shall be true and correct in all respects on such date).
- 7.02 <u>Notice of Withdrawal</u>. Prior to each Withdrawal, the Administrative Agent shall have received a Notice of Withdrawal meeting the requirements of Section 2.03.
- 7.03 <u>Withdrawal Conditions</u>. The conditions set forth in <u>Section 2.02</u> shall have been satisfied.
- 7.04 <u>Sale Order</u>. The Sale Order shall not have been reversed, stayed, vacated, terminated, modified or amended, in each case, without the prior written consent of the Commitment Parties in their sole discretion. There shall not have been any breach of, or failure to perform by the Debtors of any of their agreements, covenants, representations or warranties contained in the Sale Order.
- 7.05 <u>Asset Purchase Agreement</u>. There shall not have been (i) any breach of, or failure to perform by the Debtors of any of their agreements, covenants, representations or warranties contained in the Asset Purchase Agreement and such breach or failure to perform shall continue unremedied or unwaived, or (ii) any event of termination of the Asset Purchase Agreement pursuant to Article 11 of the Asset Purchase Agreement.

- 7.06 <u>Final DIP Order</u>. The Final DIP Order shall not have been reversed, stayed, vacated, terminated, modified or amended, in each case, without the prior written consent of the Commitment Parties and the Administrative Agent in their sole discretion. The Debtors shall be in compliance in all respects with the Final DIP Order.
- 7.07 <u>Fees and Expenses, Etc.</u>. The Administrative Agent shall have received for its own account, or for the account of each Lender, as the case may be, the Drawdown Fee in accordance with <u>Section 4.01(c)</u>, all other Fees set forth in <u>Section 4</u> and all other costs, fees and expenses (including, without limitation, legal fees and expenses and the fees and expenses of any other advisors) and other compensation contemplated hereby payable to the Administrative Agent or Lenders to the extent then due.
- 7.08 <u>Bankruptcy Matters</u>. From and after the Effective Date, any and all pleadings filed by the Debtors, and any orders entered, in the Chapter 11 Cases shall be in form and substance acceptable to the Commitment Parties, each in their sole discretion. Such documentation shall include, among other things, all motions and other filings with the Bankruptcy Court made by the Debtors of any kind and at any time, including any proposed and final orders.
- 7.09 <u>Letter of Direction</u>. The Administrative Agent shall have received a letter of direction, in form and substance reasonably satisfactory to the Administrative Agent, delivered by the Required Lenders, directing and instructing the Administrative Agent to fund the Withdrawal requested in the applicable Notice of Withdrawal.

The Notice of Withdrawal submitted by the Borrower shall be deemed to be a representation and warranty to the Administrative Agent and each of the Lenders that all the conditions specified in Section 6 (with respect to Credit Events on the Initial Funding Date) and in this Section 7 (with respect to funding of Loans and Withdrawals on or after the Initial Funding Date) and applicable to such event are satisfied as of that time. All of the certificates and other documents and papers referred to in Section 6 and in this Section 7, unless otherwise specified, shall be delivered to the Administrative Agent at the Notice Office for the account of each of the Lenders.

#### SECTION 8 Representations, Warranties and Agreements.

In order to induce the Lenders to enter into this Agreement and to make the Loans, the Borrower makes the following representations, warranties and agreements, in each case after giving effect to the Transaction, with the occurrence of each Credit Event or Withdrawal on or after the Initial Funding Date being deemed to constitute a representation and warranty that the matters specified in this Section 8 are true and correct in all material respects on and as of the Initial Funding Date and on the date of each such other Credit Event or Withdrawal (it being understood and agreed that (x) any representation or warranty which by its terms is made as of a specified date shall be required to be true and correct in all material respects only as of such specified date and (y) any representation or warranty that is qualified as to "materiality," "Material Adverse Effect" or similar language shall be true and correct in all respects on such date).

- 8.01 Organization; Powers. Each of the Borrower and its Domestic Subsidiaries (i) is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, (ii) subject to any restriction arising on account of Borrower's or each of its Domestic Subsidiaries' status as a "debtor" under the Bankruptcy Code, has all requisite power and authority to own its property and assets and to transact the business in which it is engaged and presently proposes to engage, subject to the entry of the Final DIP Order and (iii) is duly qualified and is authorized to do business and is in good standing in each jurisdiction where the ownership, leasing or operation of its property or the conduct of its business requires such qualifications except for failures to be so qualified or authorized which, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.
- 8.02 <u>Authorization; Enforceability</u>. Subject to the entry of the Final DIP Order, each Credit Party has the Company power and authority to execute, deliver and perform the terms and provisions of each of the Credit Documents to which it is party and has taken all necessary Company action to authorize the execution, delivery and performance by it of each of such Credit Documents. Subject to entry of the Final DIP Order, each Credit Party has duly executed and delivered each of the Credit Documents to which it is party, and each of such Documents constitutes its legal, valid and binding obligation enforceable in accordance with its terms, except to the extent that the enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws generally affecting creditors' rights and by equitable principles (regardless of whether enforcement is sought in equity or at law).
- 8.03 No Violation. Subject to entry of the Final DIP Order, neither the execution, delivery or performance by any Credit Party of each Credit Documents to which it is a party, nor the consummation of the Transaction, nor compliance by it with the terms and provisions thereof, (i) will contravene any provision of any material law, statute, rule or regulation or any order, writ, injunction or decree of any court or Governmental Authority, (ii) will conflict with or result in any breach of any of the terms, covenants, conditions or provisions of, or constitute a default under, or result in the creation or imposition of (or the obligation to create or impose) any Lien (except pursuant to the Security Documents or the Final DIP Order) upon any of the property or assets of any Credit Party pursuant to the terms of any indenture, mortgage, deed of trust, credit agreement or loan agreement, or any other material agreement, contract or instrument, in each case to which any Credit Party is a party or by which it or any its property or assets is bound or to which it may be subject, unless the enforcement of any rights or remedies under such indenture, mortgage, deed of trust, credit agreement or loan agreement, or any other material agreement, contract or instrument are effectively stayed by the filing of the Chapter 11 Cases, or (iii) will violate any provision of the Organization Documents of any Credit Party.
- 8.04 Approvals. Except for the entry of the Final DIP Order, and subject to the terms and conditions thereof, no order, consent, approval, license, authorization or validation of, or filing, recording or registration with (except for those that have otherwise been obtained or made on or prior to the Effective Date and which remain in full force and effect on the date of each funding of Loans and Withdrawal), or exemption by, any Governmental Authority is required to be obtained or made by, or on behalf of, any Credit Party to authorize, or is required to be obtained or made by, or on behalf of, any Credit Party in connection with, (i) the execution, delivery and performance of any Credit Document, (ii) the legality, validity, binding effect or enforceability of any such Credit Document, (iii) the grant by any Credit Party of the Liens

granted by it pursuant to the Final DIP Order and the Security Documents (including the first priority nature thereof) or (iv) the exercise by the Agent, the Required Lenders or any Lender of its rights under the Credit Documents or the remedies in respect of the Collateral pursuant to the Security Documents and the Final DIP Order, except where the failure to obtain such order, consent, approval, license, authorization or validation could not reasonably be expected to have a Material Adverse Effect.

#### 8.05 Financial Statements; Financial Condition.

- (a) The unaudited consolidated balance sheet of the Borrower and its Subsidiaries dated [September 30, 2015] and the related consolidated statements of income and cash flows and changes in shareholders' equity of the Borrower for the Fiscal Quarter ended on such date, furnished to the Lenders prior to the Effective Date, present fairly in all material respects the consolidated financial condition of the Borrower and its Subsidiaries at the date of said financial statements and the results for the period covered thereby, subject to normal year-end adjustments. All such financial statements have been prepared in accordance with GAAP consistently applied except to the extent provided in the notes to said financial statements and subject, in the case of the unaudited financial statements, to normal year-end audit adjustments (all of which are of a recurring nature and none of which, individually or in the aggregate, would be material) and the absence of footnotes.
- (b) Except as fully disclosed in the financial statements delivered pursuant to Section 8.05(a) or set forth in the schedules to the Asset Purchase Agreement or in the Debtors' schedules of financial affairs filed with the Bankruptcy Court and except for the Indebtedness incurred under this Agreement, there are no liabilities or obligations with respect to the Borrower or any of its Domestic Subsidiaries of any nature whatsoever (whether absolute, accrued, contingent or otherwise and whether or not due) which, either individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect on the Borrower or any of its Domestic Subsidiaries.
- (c) All financial projections (the "<u>Projections</u>") prepared by or as directed by the Debtors or any of their representatives in connection with the Facility and that have been made available to the Lenders or their Related Parties in connection with the Transactions contemplated by this Agreement (including the Approved Budget and the Capital Expenditure Budget) have been prepared in good faith and are based on reasonable assumptions.

#### 8.06 Litigation; Defaults.

- (a) Except as disclosed in the Asset Purchase Agreement or pending before the Bankruptcy Court, there are no actions, suits or proceedings pending or, to the knowledge of the Borrower, threatened (i) that purport to affect or pertain to the Transactions or any Credit Document or (ii) that has had, or could reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect.
- (b) Except as disclosed in the Asset Purchase Agreement or pending before the Bankruptcy Court, neither the Borrower nor any of its Domestic Subsidiaries is in default under or with respect to any Contractual Obligation that could, either individually or in in the

aggregate, reasonably be expected to have a Material Adverse Effect. No Default has occurred and is continuing or would result from the consummation of the Transactions or any other transactions contemplated by this Agreement or any other Credit Document.

8.07 True and Complete Disclosure. All factual information (taken as a whole) furnished by or on behalf of the Borrower in writing to any Agent or any Lender (including, without limitation, all information contained in the Credit Documents) for purposes of, or in connection with, this Agreement, the other Credit Documents or any transaction contemplated herein or therein is true and accurate in all material respects on the date as of which such information is dated or certified and not incomplete by omitting to state any material fact necessary to make such information (taken as a whole) not misleading in any material respect at such time in light of the circumstances under which such information was provided, it being understood and agreed that for purposes of this Section 8.07, such factual information shall not include the Projections, any estimates (including financial estimates, forecasts and other forward-looking information) or information of a general economic or general industry nature.

## 8.08 Ownership of Property; Liens; Investments.

- (a) The Borrower and each of its Domestic Subsidiaries has good and marketable title in fee simple to, or valid leasehold interests in, all Real Property necessary or used to conduct its business in the ordinary course except for (i) Permitted Liens and (ii) minor defects in title that, in the aggregate, are not substantial in amount and do not materially detract from the value of the property subject thereto or interfere in any material respect with the Borrower's ability to conduct its business as currently conducted or to utilize such properties and assets for their intended purposes. Since January 14, 2016, no portion of the Real Property of any Credit Party has suffered any material damage by fire or other casualty loss that has not been repaired and restored to the extent required by this Agreement.
- (b) [Schedule 5.4(a)(i) of the Asset Purchase Agreement sets forth a complete and accurate list of all material Real Property owned by each Credit Party on the Effective Date, showing as of the date hereof the street address, county or other relevant jurisdiction, state, and record owner. Each Credit Party has good and insurable fee simple title to the Real Property owned by such Credit Party, free and clear of all Liens, other than Permitted Liens.]
- (c) [Schedule 5.4(c)] of the Asset Purchase Agreement sets forth a complete and accurate list of all material leases of real property under which any Credit Party is the lessee on the Effective Date, showing as of the Effective Date the street address, county or other relevant jurisdiction, state, lessor, lessee, expiration date and annual rental cost thereof. Each such lease is the legal, valid and binding obligation of the lessor thereof, enforceable in accordance with its terms.]
- (d) Except as set forth on <u>Schedule 5.4(a)(ii)</u> of the Asset Purchase Agreement, no Credit Party is obligated under any right of first refusal, option or other contractual right to sell, assign or otherwise dispose of any Real Property of any Credit Party or any interest therein.
- (e) [There are no pending or, to the knowledge of any Credit Party, proposed special or other assessments for public improvements or otherwise affecting any material portion

of any owned Real Property of the Credit Parties, nor are there any contemplated improvements to such owned Real Property of the Credit Parties that may result in such special or other assessments. No Credit Party has suffered, permitted or initiated the joint assessment of any owned Real Property of the Credit Parties with any other real property constituting a separate tax lot. Each owned parcel of Real Property of the Credit Parties is composed of one or more parcels, each of which constitutes a separate tax lot and none of which constitutes a portion of any other tax lot.]

- (f) As of the Effective Date, the Credit Parties do not own any Real Property on which any improvement is located that is subject to federal flood insurance laws and requirements, [except for those improvements described on <u>Schedule 8.08(f)</u>.]
- 8.09 Tax Returns and Payments. The Borrower and each of its Domestic Subsidiaries has timely filed or caused to be timely filed with the appropriate taxing authority all Federal, state, and all other returns, statements, forms and reports for taxes (the "Returns") required to be filed by the Borrower and/or any of its Domestic Subsidiaries. The Returns accurately reflect in all material respects all liability for taxes of the Borrower and its Domestic Subsidiaries, as applicable, for the periods covered thereby. The Borrower and each of its Domestic Subsidiaries has paid all taxes and assessments payable by it which have become due, other than those that are being contested in good faith and adequately disclosed and provided for on the financial statements of each Credit Party and its Domestic Subsidiaries in accordance with GAAP or the payment thereof has been stayed by the Chapter 11 Cases. There is no material action, suit, proceeding, investigation, audit or claim now pending or, to the best knowledge of the Borrower or any of its Domestic Subsidiaries, threatened by any authority regarding any taxes relating to each Credit Party or any of its Domestic Subsidiaries.

#### 8.10 [Compliance with ERISA.

Each Plan is in compliance in form and operation with its terms and with ERISA and the Code (including without limitation the Code provisions compliance with which is necessary for any intended favorable tax treatment) and all other applicable laws and regulations, except where any failure to comply could not reasonably be expected to have a Material Adverse Effect. Except as would not reasonably be expected to result in a Material Adverse Effect, each Plan (and each related trust, if any) which is intended to be qualified under Section 401(a) of the Code (x) has received a favorable determination letter from the IRS to the effect that it meets the requirements of Sections 401(a) of the Code and that the applicable trust qualifies for exemption from taxation under Section 501(a) of the Code or an application for such a letter has been filed within the remedial amendment period and is currently being processed by the IRS with respect thereto, or (y) is comprised of a master or prototype plan that has received a favorable opinion letter from the IRS, and to the knowledge of the Borrower, nothing has occurred since the date of such determination that would reasonably be expected to adversely affect such determination (or, in the case of a Plan with no determination, to the knowledge of the Borrower, nothing has occurred that would reasonably be expected to materially adversely affect the issuance of a favorable determination letter or otherwise materially adversely affect such qualification). No ERISA Event has occurred other than as would not, individually or in the aggregate, have a Material Adverse Effect.

- (b) There exists no Unfunded Pension Liability with respect to any Plan that would have a Material Adverse Effect.
- (c) To the knowledge of the Borrower, no Multiemployer Plan is insolvent or in reorganization under Title IV of ERISA. None of the Borrower or any ERISA Affiliate has incurred any material liability with respect to a complete or partial withdrawal from any Multiemployer Plan, and, if each of the Borrower and each ERISA Affiliate were to withdraw in a complete withdrawal from any Multiemployer Plan as of the date this assurance is given or deemed given, the aggregate withdrawal liability that would be incurred would not reasonably be expected to result in a Material Adverse Effect.
- (d) To the knowledge of the Borrower, there are no actions, suits or claims pending against or involving a Plan (other than routine claims for benefits) or, which would reasonably be expected to be asserted successfully against any Plan and, if so asserted successfully, would reasonably be expected either singly or in the aggregate to have a Material Adverse Effect
- (e) The Borrower, and to the knowledge of the Borrower, any ERISA Affiliate have made all material contributions to or under each Plan and Multiemployer Plan required by law within the applicable time limits prescribed thereby, the terms of such Plan or Multiemployer Plan, respectively, or any contract or agreement requiring contributions to a Plan or Multiemployer Plan save where any failure to comply, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect.
- (f) Except as would not individually or in the aggregate have a Material Adverse Effect, (x) neither the Borrower nor any ERISA Affiliate has ceased operations at a facility so as to become subject to the provisions of Section 4062(e) of ERISA or withdrawn as a "substantial employer" (as defined in Section 4001(a)(2) of ERISA) from a Plan so as to become subject to the provisions of Section 4063 of ERISA, and (y) neither the Borrower nor any ERISA Affiliate has any liability under Section 4069 or 4212(c) of ERISA.]

#### 8.11 [Reserved].

- 8.12 <u>Insurance</u>. The properties of the Borrower and its Domestic Subsidiaries are insured with financially sound and reputable insurance companies not Affiliated with the Borrower, in such amounts, with such deductibles and covering such risks as are customarily (as determined in good faith by an Authorized Officer of the Borrower) carried by companies engaged in similar business and owning similar properties in localities where the Borrower or the applicable Subsidiary operates.
- 8.13 <u>Subsidiaries</u>. The Borrower has no Subsidiaries other than those Subsidiaries listed on <u>Schedule 8.13</u>. <u>Schedule 8.13</u> sets forth the correct percentage ownership (direct and indirect) of the Borrower in each class of capital stock or other Equity Interests of each of its Subsidiaries and also identifies the direct owner thereof.
- 8.14 <u>Compliance with Statutes, etc.</u> Each of the Borrower and its Domestic Subsidiaries is in compliance with all applicable statutes, regulations and orders of, and all applicable restrictions imposed by, all Governmental Authorities in respect of the conduct of its

business and the ownership of its property, except such non-compliances as could not reasonably be expected to have a Material Adverse Effect.

## 8.15 <u>Investment Company Act; Margin Regulations.</u>

- (a) Neither the Borrower nor any of its Subsidiaries is an "investment company" or a company "controlled" by an "investment company," within the meaning of the Investment Company Act of 1940, as amended.
- (b) None of the Credit Parties is engaged in any will not engage in, principally or as one of its important activities, in the business of purchasing or carrying margin stock (within the meaning of Regulation U issued by the FRB), or extending credit for the purpose of purchasing or carrying margin stock. No part of the proceeds of the Loans will be used to purchase or carry any such margin stock or to extend credit to others for the purpose of purchasing or carrying any such margin stock or for any purpose that violates the provisions of Regulation T, U or X of the FRB.
- [Environmental Matters. Except as set forth on Schedule 5.5 of the Asset Purchase Agreement or as would not reasonably be expected to have a Material Adverse Effect: (a) each of the Borrower and its Domestic Subsidiaries has complied during the previous three (3) years and is in compliance with all applicable Environmental Laws and has obtained and is in compliance with the terms of all Environmental Permits required for the business operations, properties or facilities of the Borrower or any of its Domestic Subsidiaries (and all such Environmental Permits are in full force and effect, free from breach, and the Transaction will not adversely affect them); (b) there are no Environmental Claims pending or to the knowledge of the Borrower, threatened, against the Borrower or any of its Domestic Subsidiaries; (c) no Lien, other than a Permitted Lien, has been recorded or to the knowledge of the Borrower, threatened under any Environmental Law with respect to any Real Property owned by the Borrower or any of its Domestic Subsidiaries; (d) neither the Borrower nor any of its Domestic Subsidiaries has treated, recycled, stored, disposed of, arranged for or permitted the disposal of, transported, handled, or Released any substance, including Hazardous Materials, or owned or operated any property or facility (and no such property or facility is contaminated by any such substance) in a manner that has given or would give rise to any damages or liabilities under any Environmental Law; (e) neither the Borrower nor any of its Domestic Subsidiaries has contractually assumed or accepted responsibility for any liability of any other Person under any Environmental Law; and (e) there are no facts, circumstances, conditions or occurrences with respect to the past or present business, operations, properties or facilities of the Borrower or any of its Domestic Subsidiaries, that could reasonably be expected to give rise to any Environmental Claim against the Borrower or its Domestic Subsidiaries or any liability of the Borrower or its Domestic Subsidiaries under any Environmental Law.]
- 8.17 [Employment and Labor Relations. Other than proceedings commenced in the Chapter 11 Cases or related thereto, neither the Borrower nor any of its Domestic Subsidiaries is engaged in any unfair labor practice that could reasonably be expected, either individually or in the aggregate, to have a Material Adverse Effect. Other than proceedings commenced in the Chapter 11 Cases or related thereto, there is (i) no unfair labor practice complaint pending against the Borrower or any of its Domestic Subsidiaries or, to the knowledge of the Borrower,

threatened against any of them, before the National Labor Relations Board or similar agency or entity governing labor relations of any Domestic Subsidiary, and no grievance or arbitration proceeding arising out of or under any collective bargaining agreement is so pending against the Borrower and its Domestic Subsidiaries or, to the knowledge of the Borrower, threatened against any of them, (ii) no strike, labor dispute, slowdown or stoppage pending against the Borrower or any of its Domestic Subsidiaries or, to the knowledge of the Borrower, threatened against the Borrower or any of its Domestic Subsidiaries, (iii) no union representation question exists with respect to the employees of the Borrower or any of its Domestic Subsidiaries, (iv) no equal employment opportunity charges or other claims of employment discrimination are pending or, to the Borrower's knowledge, threatened against the Borrower or any of its Domestic Subsidiaries and (v) no wage and hour department investigation has been made of the Borrower or any of its Domestic Subsidiaries, except (with respect to any matter specified in clauses (i) to (v) above, either individually or in the aggregate) such as could not reasonably be expected to have a Material Adverse Effect.]

- Intellectual Property, etc. Each of the Borrower and its Domestic Subsidiaries owns or has the right to use all the patents, trademarks, domain names, service marks, trade names, copyrights, inventions, trade secrets, proprietary information and know-how of any type, whether or not written (including, but not limited to, rights in computer programs and databases) and formulas, or rights with respect to the foregoing, (collectively, the "IP Rights") that are reasonably necessary for the operation of their respective businesses without any known conflict with the IP Rights of any other Person, except to the extent any rights of others which, or the failure to so own or have which, as the case may be, directly or indirectly, could not reasonably be expected to have a Material Adverse Effect. To the best knowledge of the Borrower, no slogan or other advertising device, product, process, method, substance, part or other material now employed, or now contemplated to be employed, by Borrower or any of its Domestic Subsidiaries infringes upon any valid rights held by any other Person, except to the extent such infringement, individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect. No claim or litigation regarding any of the foregoing is pending or, to the best knowledge of the Borrower, threatened, which, either individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.
- 8.19 Representations and Warranties in Asset Purchase Agreement. [On the Effective Date and the date of each Withdrawal and funding of Loans, all representations and warranties made by Borrower and its Subsidiaries party thereto in the Asset Purchase Agreement shall be true and correct in all respects with the same effect as though such representations and warranties had been made on the date of such Credit Event (it being understood and agreed that any representation or warranty which by its terms is made as of a specified date shall be required to be true and correct in all respects only as of such specified date).]
- 8.20 <u>Foreign Assets Control Regulations, Etc.</u> Neither of the advance of the Loans nor the use of the proceeds of any thereof will violate the Trading With the Enemy Act (50 U.S.C. § 1 et seq., as amended) (the "<u>Trading With the Enemy Act</u>") or any of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) (the "<u>Foreign Assets Control Regulations</u>") or any enabling legislation or executive order relating thereto (which for the avoidance of doubt shall include, but shall not be limited to (a) Executive Order 13224 of September 21, 2001 Blocking Property and Prohibiting

Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (66 Fed. Reg. 49079 (2001)) (the "Executive Order") and (b) the Patriot Act. Furthermore, none of the Borrower or its Affiliates (a) is or will become a "blocked person" as described in the Executive Order, the Trading With the Enemy Act or the Foreign Assets Control Regulations or (b) engages or will engage in any dealings or transactions, or be otherwise associated, with any such "blocked person" or in any manner violative of any such order. Each Credit Party is in compliance, in all material respects, with the Patriot Act. No part of the proceeds of the Loans will be used by the Credit Parties, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

## 8.21 [Reserved].

- 8.22 <u>Permits; Licenses.</u> Each Credit Party has, and is in compliance with, all permits, licenses, authorizations, approvals, entitlements and accreditations required for such Person lawfully to own, lease, manage or operate, or to acquire, each business currently owned, leased, managed or operated, or to be acquired, by such Person, which, if not obtained, would reasonably be expected to have a Material Adverse Effect. Other than the pendency of the Chapter 11 Cases, no condition exists or event has occurred which, in itself or with the giving of notice or lapse of time or both, would result in the suspension, revocation, impairment, forfeiture or non-renewal of any such permit, license, authorization, approval, entitlement or accreditation, and there is no claim that any thereof is not in full force and effect, except, to the extent any such condition, event or claim would not reasonably be expected to have a Material Adverse Effect.
- 8.23 <u>Bank Accounts and Securities Accounts.</u> [Schedule 5.17] to the Asset Purchase Agreement sets forth a complete and accurate list as of the Effective Date of all deposit, checking and other bank accounts, all securities and other accounts maintained with any broker dealer and all other similar accounts maintained by each Credit Party, together with a description thereof including, without limitation, the bank or broker dealer at which such account is maintained, the account number and the purpose thereof.]

# 8.24 [Reserved].

## 8.25 Reorganization Matters; Administrative Priority; Lien Priority.

- (a) Proper notice of (x) the motion seeking approval of the Credit Documents and the Final DIP Order and (y) the hearing for the approval of the Final DIP Order has been given, provided that the Borrower shall give, on a timely basis as specified in the Final DIP Order, all notices required to be given to all parties specified in the Final DIP Order.
- (b) After the entry of the Final DIP Order and pursuant to and to the extent permitted or provided in the Final DIP Order, the Obligations will constitute allowed superpriority administrative expense claims in the Chapter 11 Cases having priority over all administrative expense claims and unsecured claims against the Borrower and other Credit Parties now existing or hereafter arising, of any kind whatsoever, including, without limitation,

all administrative expense claims of the kind specified in sections 326, 330, 331, 503(b), 506(c), 507(a), 507(b), 726 or any other provision of the Bankruptcy Code or otherwise, as provided under section 364(c)(1) of the Bankruptcy Code, subject in all respects, to the Carve-Out.

- (c) After the entry of the Final DIP Order and pursuant to and to the extent provided in the Final DIP Order, the Obligations will be secured by a valid and perfected first priority Lien on all of the Collateral, subject in all respects, to the Carve-Out.
- (d) The Final DIP Order (with respect to the period on and after the entry of the Final DIP Order) is in full force and effect and has not been reversed, stayed, vacated, terminated, modified or amended without the prior written consent of the Administrative Agent and the Commitment Parties, each in their sole discretion.
- (e) The Approved Budget, the Capital Expenditure Budget and all projected consolidated balance sheets, income statements and cash flow statements of the Borrower and its Subsidiaries delivered to the Administrative Agent in connection herewith were prepared in good faith on the basis of the assumptions stated therein, which assumptions were believed in good faith by the Borrower to be fair in light of the conditions existing at the time of delivery of such report or projections (it being understood that any projections or estimates made in the items described in this subsection (e) are not to be viewed as facts and are subject to significant uncertainties and contingencies, that no assurance can be given that any such projections or estimates will be realized, that actual results may differ from projected results and such differences may be material).

SECTION 9 <u>Affirmative Covenants</u>. The Borrower hereby covenants and agrees that, until the Total Commitments have been terminated and Loans, Fees and all other Obligations (other than contingent obligations for which no claim has been asserted) incurred hereunder are paid in full in cash:

#### 9.01 Information Covenants.

#### (a) [Reserved].

(b) Quarterly Financial Statements. The Borrower will furnish to the Administrative Agent for distribution to Lenders, as soon as available, but in any event within 45 days after the close of each of the first three quarterly accounting periods in each Fiscal Year of the Borrower, (i) the consolidated and consolidating balance sheet of the Borrower and its Subsidiaries as at the end of such quarterly accounting period and the related consolidated and consolidating statements of income and retained earnings and statement of cash flows for such quarterly accounting period and for the elapsed portion of the Fiscal Year ended with the last day of such quarterly accounting period, in each case setting forth comparative figures for the corresponding quarterly accounting period in the prior Fiscal Year, all in reasonable detail and certified by an Authorized Officer of the Borrower that they fairly present in all material respects in accordance with GAAP the financial condition of the Borrower and its Subsidiaries as of the dates indicated and the results of their operations for the periods indicated, subject to normal year-end audit adjustments and the absence of footnotes, and (ii) management's discussion and

analysis of the important operational and financial developments during such quarterly accounting period.

#### (c) [Reserved].

- (d) <u>Management Letters</u>. The Borrower will furnish to the Administrative Agent for distribution to Lenders, promptly after the Borrower's or any of its Subsidiaries' receipt thereof, a copy of any "management letter" received from its certified public accountants and management's response thereto; <u>provided</u> that such delivery only shall be required to the extent allowed by the relevant certified public accountant's policy and practice.
- (e) <u>Variance Report</u>. The Borrower will furnish to the Administrative Agent and to the Commitment Parties and the Lenders who have signed a non-disclosure agreement:
  - (i) As soon as available but in any event within five (5) Business Days after the end of each Budget Testing Period (with the first variance report due on February 19, 2016), a variance report, showing comparisons of actual amounts for each line item against the budgeted amounts for such line item in the Approved Budget for the preceding two-week period and on a cumulative basis for the immediately preceding Budget Testing Period accompanied by a narrative explanation. For purposes of the first Budget Testing Period, such variance report shall compare cumulative actuals and budgeted amounts from the first week of the Approved Budget through the week ending February 13, 2016.
  - (ii) As soon as available but in any event within five (5) Business Days after the end of each Capital Expenditure Testing Period (with the first variance report due on February 5, 2016), a variance report, showing comparisons of actual amounts for each line item for the month then ended against the budgeted amounts for such line item in the Capital Expenditure Budget for such month then ended (as well as on a cumulative basis from January 1, 2016 through the end of such Capital Expenditure Testing Period), in each case calculated on GAAP basis, accompanied by a narrative explanation.
- Agent for distribution to Lenders, at the time of the delivery of the financial statements provided for in Sections 9.01(b), a compliance certificate from an Authorized Officer of the Borrower in the form of Exhibit D certifying on behalf of the Borrower that, to such Authorized Officer's knowledge after due inquiry, no Default or Event of Default has occurred and is continuing or, if any Default or Event of Default has occurred and is continuing, specifying the nature and extent thereof, which certificate shall (i) certify that there have been no changes to the Schedules to this Credit Agreement and (ii) set forth in reasonable detail the calculations required to establish compliance with the covenant set forth in Section 10.10(b).
- (g) <u>Notice of Default, Litigation and Material Adverse Effect</u>. The Borrower will furnish to the Administrative Agent for distribution to Lenders, promptly, and in any event within three (3) Business Days after any Authorized Officer of the Borrower or any of its Subsidiaries obtains knowledge thereof, notice of (i) the occurrence of any event which constitutes a Default or an Event of Default, (ii) of any matter that has resulted or could

reasonably be expected to result in a Material Adverse Effect, including (x) the breach or nonperformance of, or any default under, a Contractual Obligation of the Borrower or any Domestic Subsidiary, (y) any dispute, litigation, investigation, proceeding or suspension between the Borrower or any Domestic Subsidiary and any Governmental Authority, or (z) the commencement of, or any material adverse development in, any litigation or governmental investigation or proceeding pending against the Borrower or any of its Domestic Subsidiaries (including pursuant to any applicable Environmental Laws), (iii) of the occurrence of any ERISA Event, (iv) of any material change in accounting policies or financial reporting practices by any Credit Party or any Domestic Subsidiary thereof and (v) any breach of, or failure to perform by Borrower or any of its Domestic Subsidiaries of their agreements, covenants, representations or warranties contained in the Asset Purchase Agreement or in the Sale Order. Each notice pursuant to Section 9.01(g) shall be accompanied by a statement of an Authorized Officer of the Borrower setting forth details of the occurrence referred to therein and stating what action the Borrower has taken and proposes to take with respect thereto. Each notice pursuant to Section 9.01(g) shall describe with particularity any and all provision of this Agreement and any other Credit Document that have been breached.

- (h) Other Reports and Filings. The Borrower will furnish to the Administrative Agent for distribution to Lenders, promptly after the filing or delivery thereof, copies of all financial information, proxy materials and reports, if any, which the Borrower or any of its Domestic Subsidiaries shall publicly file with the Securities and Exchange Commission or any successor thereto (the "SEC") or deliver to holders (or any trustee, agent or other representative therefor) or any of its material Indebtedness pursuant to the terms of the documentation governing the same; and
- (i) <u>Environmental Matters</u>. The Borrower will furnish to the Administrative Agent for distribution to Lenders, promptly after any Authorized Officer of the Borrower or any of its Domestic Subsidiaries obtains actual knowledge thereof, notice of the following environmental matters to the extent that such environmental matters, either individually or when aggregated with all other such environmental matters, could reasonably be expected to have a Material Adverse Effect:
  - (i) any pending or threatened Environmental Claim against the Borrower or any of its Domestic Subsidiaries or any Real Property owned, leased or operated by the Borrower or any of its Domestic Subsidiaries;
  - (ii) any condition or occurrence on or arising from any Real Property owned, leased or operated by the Borrower or any of its Domestic Subsidiaries that (a) results in noncompliance by the Borrower or any of its Domestic Subsidiaries with any applicable Environmental Law or Environmental Permit or (b) could reasonably be expected to result in an Environmental Claim against the Borrower or any of its Domestic Subsidiaries or any such Real Property;
  - (iii) any condition or occurrence on any Real Property owned, leased or operated by the Borrower or any of its Domestic Subsidiaries that could reasonably be expected to cause such Real Property to be subject to any restrictions on the ownership, lease, occupancy, use or transferability by the

Borrower or any of its Domestic Subsidiaries of such Real Property under any Environmental Law;

- (iv) any actual or alleged noncompliance with Environmental Laws or Environmental Permits; or
- (v) any removal, remediation, or corrective action to the extent required by any Environmental Law or any Governmental Authority in response to the Release or threatened Release of any Hazardous Material on any Real Property owned, leased or operated by the Borrower or any of its Domestic Subsidiaries.

All such notices shall describe in reasonable detail the nature of the claim, investigation, condition, occurrence or removal or remedial action and the Borrower's or such Subsidiary's response thereto.

(j) Other Information. The Borrower will furnish to the Administrative Agent for distribution to Lenders, promptly such other information or documents (financial or otherwise) with respect to the business, financial or corporate affairs of the Borrower or any of its Subsidiaries as any Agent or any Lender (through the Administrative Agent) may reasonably request from time to time.

Documents required to be delivered pursuant to Section 9.01(b) or (h) (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date (i) on which the Borrower posts such documents, or provides a link thereto on the Borrower's website on the Internet; or (ii) on which such documents are posted on the Borrower's behalf on an Internet or intranet website, if any, to which each Lender and the Administrative Agent have access (whether a commercial, third-party website or whether sponsored by the Administrative Agent); provided that: (i) the Borrower shall deliver paper copies of such documents to the Administrative Agent or any Lender that requests the Borrower to deliver such paper copies until a written request to cease delivering paper copies is given by the Administrative Agent or such Lender and (ii) the Borrower shall notify (which may be by facsimile or electronic mail) the Administrative Agent and each Lender of the posting of any such documents and provide to the Administrative Agent by electronic mail electronic versions (i.e., soft copies) of such documents.

Borrower hereby acknowledges that (a) the Administrative Agent will make available to the Lenders materials and/or information provided by or on behalf of the Borrower hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks or another similar electronic system (the "Platform") and (b) certain of the Lenders (each, a "Public Lender") may have personnel who do not wish to receive material non-public information with respect to the Borrower or any of its Affiliates, or the respective securities of any of the foregoing, and who may be engaged in investment and other market-related activities with respect to such Persons' securities. So long as the Borrower has not clearly and conspicuously marked "PUBLIC" (which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof), all such Borrower Materials shall be deemed to be non-public information with respect to the Borrower or any of its Affiliates. By marking Borrower

Materials "<u>PUBLIC</u>," (i) the Borrower shall be deemed to have authorized the Administrative Agent and the Lenders to treat such Borrower Materials as not containing any material non-public information (although it may be sensitive and proprietary) with respect to such Credit Party or its securities for purposes of United States Federal and state securities laws, (ii) all Borrower Materials marked "<u>PUBLIC</u>" are permitted to be made available through a portion of the Platform designated "<u>Public Side Information</u>" and (iii) the Administrative Agent shall be entitled to treat the Borrower Materials that are not marked "<u>PUBLIC</u>" as being suitable only for posting on a portion of the Platform not designated "<u>Public Side Information</u>."

#### 9.02 Books, Records and Inspections.

The Borrower will, and will cause each of its Domestic Subsidiaries to, keep proper books of record and accounts in which full, true and correct entries in conformity with GAAP and all requirements of law shall be made of all dealings and transactions in relation to its business and activities. The Borrower will, and will cause each of its Domestic Subsidiaries to, permit officers and designated representatives of the Administrative Agent or any Lender to visit and inspect, under guidance of officers of the Borrower or such Subsidiary, any of the properties of the Borrower or such Subsidiary, and to examine the books of account of the Borrower or such Subsidiary and discuss the affairs, finances and accounts of the Borrower or such Subsidiary with, and be advised as to the same by, its and their officers and independent accountants, all upon reasonable prior notice and at such reasonable times and intervals and to such reasonable extent as the Administrative Agent (at the direction of the Required Lenders) or any such Lender may reasonably request (and subject, in the case of any such meetings or advice from such independent accountants, to such accountants' customary policies and procedures); provided that, excluding any such visits and inspections during the continuance of an Event of Default (a) only the Administrative Agent on behalf of and at the direction of the Required Lenders may exercise rights of the Administrative Agent and the Lenders under this Section 9.02 and (b) the Administrative Agent shall not exercise such rights more than twice in any calendar year.

#### 9.03 Maintenance of Property; Insurance.

- (a) The Borrower will, and will cause each of its Domestic Subsidiaries to, (i) keep all property necessary to the business of the Borrower and its Domestic Subsidiaries in good working order and condition, ordinary wear and tear excepted and subject to the occurrence of casualty events, (ii) maintain with financially sound and reputable insurance companies insurance on all such property and against all such risks as is consistent and in accordance with industry practice for companies similarly situated owning similar properties and engaged in similar businesses as the Borrower and its Domestic Subsidiaries, and (iii) furnish to the Administrative Agent, upon its request therefor (at the direction of the Required Lenders), full information as to the insurance carried. The provisions of this Section 9.03 shall be deemed supplemental to, but not duplicative of, the provisions of any Security Documents that require the maintenance of insurance.
- (b) If at any time any improvement on a parcel of Real Property is located in an area identified as a special flood hazard area by the Federal Emergency Management Agency or any successor thereto or other applicable agency, the Borrower will, and will cause each of its

Domestic Subsidiaries to, at all times keep and maintain flood insurance in an amount no less than the amount sufficient to comply with the rules and regulations promulgated under the National Flood Insurance Act of 1968 and Flood Disaster Protection Act of 1973, each as amended from time to time.

- (c) The Borrower will, and will cause each of its Domestic Subsidiaries to, at all times keep its property insured in favor of the Collateral Agent, and all policies or certificates (or certified copies thereof) with respect to such insurance (and any other insurance maintained by the Borrower and/or such Subsidiaries) shall be endorsed to the Collateral Agent's satisfaction for the benefit of the Collateral Agent (including, without limitation, by naming the Collateral Agent as loss payee and/or additional insured).
- 9.04 <u>Existence</u>; <u>Franchises</u>. The Borrower will, and will cause each of its Domestic Subsidiaries to, do or cause to be done, all things necessary to preserve and keep in full force and effect (i) its existence and (ii) its material rights, franchises, licenses, permits, copyrights, trademarks and patents; <u>provided</u>, <u>however</u>, that nothing in this <u>Section 9.04</u> shall prevent Asset Sales and other transactions by the Borrower or any of its Domestic Subsidiaries in accordance with <u>Section 10.02</u> and <u>Section 10.07</u>.
- 9.05 <u>Compliance with Statutes, etc.</u> The Borrower will, and will cause each of its Domestic Subsidiaries to, comply with all applicable statutes, regulations and orders of, and all applicable restrictions imposed by, all Governmental Authorities in respect of the conduct of its business and the ownership of its property, except such non-compliances as could not, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

## 9.06 Compliance with Environmental Laws.

Except as could not reasonably be expected to have a Material Adverse Effect, (i) the Borrower will comply, and will cause each of its Domestic Subsidiaries to comply, with all Environmental Laws and Environmental Permits applicable to or required in respect of the conduct of its business or operations or by the ownership, lease or use of any Real Property now or hereafter owned, leased or operated by the Borrower or any of its Domestic Subsidiaries and (ii) to the extent required by Environmental Law, will pay or cause to be paid all costs and expenses incurred in connection with such compliance, and will keep or cause to be kept all such Real Property free and clear of any Liens, other than Permitted Liens, imposed pursuant to such Environmental Laws. Except as could not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, neither the Borrower nor any of its Domestic Subsidiaries will generate, use, treat, store, Release or dispose of, or permit the generation, use, treatment, storage, Release or disposal of Hazardous Materials on any Real Property now or hereafter owned, leased or operated by the Borrower or any of its Domestic Subsidiaries, or transport or permit the transportation of Hazardous Materials to or from any such Real Property, except for Hazardous Materials generated, used, treated, stored, Released or disposed of at any such Real Properties, or transported, in compliance with all applicable Environmental Laws and as required in connection with the normal operation, use and maintenance of the business or operations of the Borrower or any of its Domestic Subsidiaries;

- (i) After the receipt by the Administrative Agent or any Lender of any notice of the type described in Section 9.01(h), (ii) at any time that Borrower or any of its Domestic Subsidiaries is not in compliance with Section 9.06(a) or (iii) in the event that any Agent (at the direction of the Required Lenders) or the Lenders have exercised any of the remedies pursuant to Section 11, the Borrower will (in each case) provide, at the expense of the Borrower and as is reasonably requested by the Administrative Agent (at the direction of the Required Lenders), an environmental site assessment report concerning any relevant Real Property owned, leased or operated by the Borrower or any of its Domestic Subsidiaries, prepared by an environmental consulting firm reasonably approved by the Administrative Agent (at the direction of the Required Lenders), indicating the presence or absence of Hazardous Materials and the potential cost of any required removal or remedial action in connection with such Hazardous Materials on such Real Property. If the Borrower fails to provide the same within 30 days after such request was made, the Administrative Agent (at the direction of the Required Lenders) may order the same, the reasonable cost of which shall be borne by the Borrower, and the Borrower shall and hereby does grant to the Administrative Agent and the Lenders and their respective agents reasonable access to such Real Property, and specifically grant the Administrative Agent and the Lenders an irrevocable non-exclusive license, subject to the rights of tenants, to undertake (in the case of the Administrative Agent, at the direction of the Required Lenders) such an assessment at any reasonable time upon reasonable notice to the Borrower, provided that such access and work shall not unreasonably interfere with normal operations of the Borrower or any of them, all at expense of the Borrower.
- 9.07 <u>ERISA</u>. Promptly and in any event within 30 days after the Borrower knows that any ERISA Event has occurred that would reasonably be expected to result in material liability to the Borrower, the Borrower shall supply to the Administrative Agent (in sufficient copies for all Lenders, if the Administrative Agent at the direction of the Required Lenders so requests) a certificate of an authorized officer of the Borrower describing such ERISA Event and the action, if any, proposed to be taken with respect to such ERISA Event and a copy of any notice filed with the PBGC or the IRS pertaining to such ERISA Event and any notices received by the Borrower or ERISA Affiliate from the PBGC or any other governmental agency with respect thereto
- 9.08 <u>End of Fiscal Years; Fiscal Quarters</u>. Borrower will cause (i) its and each of its Domestic Subsidiaries' fiscal years to end on December 31 of each calendar year and (ii) its and each of its Domestic Subsidiaries' fiscal quarters to end on the last day of each period described in the definition of "Fiscal Quarter".
- 9.09 Payment of Taxes. Borrower will pay and discharge, and will cause each of its Domestic Subsidiaries to pay and discharge all taxes, assessments and governmental charges or levies imposed upon it, and all lawful claims which, if unpaid, might become a Lien or charge upon any properties of the Borrower or any of its Domestic Subsidiaries not otherwise permitted under Section 10.01(a); provided that neither Borrower nor any of its Domestic Subsidiaries shall be required to pay any such tax, assessment, charge, levy or claim (i) which is being contested in good faith and by proper proceedings if it has maintained adequate reserves with respect thereto in accordance with GAAP or (ii) which is stayed by the filing of the Chapter 11 Cases.

- Use of Proceeds. Subject to Section 10.10 of this Agreement, the proceeds of any Loans shall be used (a) in accordance with the terms of the Approved Budget (subject to permitted variances) and only for the following purposes: (i) the payment of prepetition amounts (including prepetition payments to certain critical vendors identified by the Borrower) to the extent authorized by the Bankruptcy Court pursuant to first day orders and other orders entered in the Chapter 11 Cases prior to the Effective Date or with the consent of the Commitment Parties, and (ii) to the extent set forth in the Approved Budget (subject to permitted variances) and in accordance with the terms of this Agreement and the Final DIP Order, for the payment of working capital and other general corporate needs of the Borrower and the Guarantors in the ordinary course of business, and (b) for the payment of other costs and expenses of administering the Chapter 11 Cases (including payments benefiting from the Carve-Out), including allowed professional fees subject to the terms and conditions set forth in this Agreement (and including fees incurred prior to the Effective Date) and the Final DIP Order (for the avoidance of doubt, no proceeds of the Loans may be used to make any adequate protection payments set forth in paragraph 11(a) of the Cash Collateral Order). Notwithstanding the foregoing, the use of the proceeds of the Loans, the Carve-Out and the Collateral shall be subject to further restrictions consistent with those in the Final DIP Order, including with respect to limitations on investigating and challenging claims of the Prepetition Secured Parties and limitations on paying fees and expenses of official committees. No proceeds of any Loans or the Collateral may be used to initiate, prepare or prosecute proceedings or actions on account of any claims and defenses against the Prepetition Secured Parties, the Agent or the Lenders.
- For the avoidance of doubt and notwithstanding any other provision of this (b) Agreement, other than the Investigation Budget (as defined in the Final DIP Order) (which may be used solely for the purposes authorized in the Final DIP Order), no Loans, Collateral, Prepetition Collateral, including Cash Collateral, or any portion of the Carve-Out, may be used directly or indirectly by any Debtor, any official committee appointed in the Chapter 11 Cases, including the Creditors' Committee (as defined in the Final DIP Order) and the Retiree Committee (as defined in the Final DIP Order), or any trustee appointed in the Chapter 11 Cases or any successor case, including any chapter 7 case, or any other person, party or entity to (i) investigate, object, contest, or raise any defense to the validity, perfection, priority, extent, or enforceability of the Prepetition Obligations or the Obligations and/or the liens, claims or rights granted under the Cash Collateral Order, the Final DIP Order, the Credit Documents or the Prepetition Debt Documents, or take any action purporting to do any of the foregoing; (ii) investigate, assert or prosecute any Claims and Defenses (as defined in the Final DIP Order) against the Prepetition Secured Parties, the Agent or the Lenders or their respective predecessorsin-interest, agents, affiliates, representatives, attorneys, or advisors or any action purporting to do the foregoing in respect of the Prepetition Obligations, the Prepetition Liens, the Obligations, the DIP Claims and Liens, and/or the Adequate Protection Obligations, Adequate Protection Liens and Prepetition Secured Parties' Superpriority Claims (as defined in the Final DIP Order) granted under the Cash Collateral Order or the Final DIP Order, as applicable; (iii) prevent, hinder, or otherwise delay the Prepetition Secured Parties', the Agent's or the Lenders', as applicable, enforcement or realization on the Prepetition Obligations, Prepetition Collateral, Obligations, Collateral, Cash Collateral, and the liens, claims, and rights granted to such parties granted under the Cash Collateral Order and the Final DIP Order, in accordance with the Prepetition Debt Documents, the Credit Documents, the Cash Collateral Order or the Final DIP Order, as applicable; (iv) seek to modify any of the rights and remedies granted to the Prepetition Secured

Parties, the Agent or the Lenders granted under the Cash Collateral Order, the Final DIP Order (other than with the consents contemplated thereunder), the Prepetition Debt Documents or the Credit Documents, as applicable; (v) apply to the Bankruptcy Court for authority to approve superpriority claims or grant liens (other than as permitted by the Final DIP Order) or security interests in the Collateral or any portion thereof that are senior to, or on parity with, the DIP Claims and Liens, Adequate Protection Liens, and Prepetition Secured Parties' Superpriority Claims, or Prepetition Liens, unless all Obligations, Prepetition Obligations, Adequate Protection Obligations, and claims granted to the Agent, Lenders or Prepetition Secured Parties granted under the Cash Collateral Order and the Final DIP Order, as applicable, have been refinanced or paid in full in cash (including the cash collateralization of any letters of credit) or otherwise agreed to in writing by the Required Lenders or the Steering Committee and the Agent, as applicable, each in its sole discretion; or (vi) seek to pay any amount on account of any claims arising prior to the Petition Date unless such payments have been authorized to be paid prior to the Effective date or are agreed to in writing by the Required Lenders in their sole discretion or are otherwise included in the Approved Budget.

## 9.11 Additional Security; Further Assurances; etc.

- The Borrower will cause, and will cause each of the other Credit Parties to (a) cause, each of its Wholly-Owned Domestic Subsidiaries formed or acquired (or which first becomes such a Wholly-Owned Domestic Subsidiary) after the Effective Date to become a Credit Party (and, if requested by the Administrative Agent (at the direction of the Required Lenders) to execute and deliver any applicable deposit account control agreements, securities account control agreements, intellectual property security agreement, intellectual property agreement supplements and all other appropriate Security Documents, in each case, in form and substance satisfactory to the Required Lenders in their sole discretion within ten (10) days (or such longer time period if agreed to by the Administrative Agent at the direction of the Required Lenders) after the formation or acquisition thereof or after the first date upon which the respective Subsidiary of such Person becomes a Wholly-Owned Domestic Subsidiary. Upon execution and delivery of a joinder to this Agreement, each such Person (i) shall become a Guarantor hereunder and thereupon shall have all of the rights, benefits, duties, and obligations in such capacity under the Credit Documents and (ii) will grant Liens to the Collateral Agent, for the benefit of the Secured Creditors, in any property of such Credit Party which constitutes Collateral pursuant to Security Documents satisfactory to the Required Lender in their sole discretion, if the Final DIP Order does not already create a valid and perfected Lien on such assets. In addition, each new Wholly-Owned Subsidiary that is required to execute any Credit Document shall execute and deliver, or cause to be executed and delivered, all other relevant documentation of the type described in Section 6 as such new Subsidiary would have had to deliver if such new Subsidiary were a Credit Party on the Effective Date, including the filing of the UCC financing statements (or their equivalent). The Borrower shall deliver to the Administrative Agent prompt written notice if it or any other Credit Party acquires any Real Property on which any improvement is located that is subject to federal flood insurance laws and requirements.
- (b) The Borrower will, and will cause each other Credit Party to, grant to the Collateral Agent for the benefit of the Secured Creditors security interests in such assets (including, if requested by the Administrative Agent at the direction of the Required Lenders,

mortgages on any Real Property) and of the Borrower and such other Credit Party as are not covered by the original Security Documents (collectively, the "Additional Security Documents").

- (c) The Borrower will, and will cause each of the other Credit Parties to, at the expense of the Borrower, make, execute, endorse, acknowledge, file and/or deliver to the Collateral Agent from time to time such vouchers, invoices, schedules, confirmatory assignments, conveyances, financing statements, transfer endorsements, powers of attorney, certificates, Real Property surveys, reports, landlord waivers, bailee agreements, control agreements and other assurances or instruments and take such further steps relating to the Collateral covered by any of the Security Documents as the Collateral Agent (at the direction of the Required Lenders) may reasonably require. Furthermore, the Borrower will, and will cause the other Credit Parties that are Subsidiaries of the Borrower to, deliver to the Collateral Agent such opinions of counsel, title insurance and other related documents as may be reasonably requested by the Administrative Agent (at the direction of the Required Lenders) to assure that this Section 9.11 has been complied with.
- (d) If the Administrative Agent or the Required Lenders reasonably determine that they are required by law or regulation to have appraisals prepared in respect of any Real Property of the Borrower and the other Credit Parties constituting Collateral, the Borrower will, at its own expense, provide to the Administrative Agent appraisals which satisfy the applicable requirements of the Real Estate Appraisal Reform Amendments of the Financial Institution Reform, Recovery and Enforcement Act of 1989, as amended, and which shall otherwise be in form and substance satisfactory to the Administrative Agent in its sole discretion.
- (e) To the extent the Required Lenders request any deposit account control agreement, securities account control agreement or any intellectual property security agreement, in each case, in form and substance satisfactory to the Administrative Agent in its sole discretion, then the Borrower shall cause all such actions to be taken as promptly as practicable after the Effective Date (at the direction of the Required Lenders).
- (f) The Borrower agrees that each action required by clauses (b) through (e) of this <u>Section 9.11</u> shall be completed as soon as possible, but in no event later than fifteen (15) days after such action is requested to be taken by the Administrative Agent (at the direction of the Required Lenders) or the Required Lenders (as such date may be extended at the sole discretion of the Administration Agent acting at the direction of the Required Lenders); <u>provided</u> that, in no event will Borrower or any of its Subsidiaries be required to take any action, other than using its commercially reasonable efforts, to obtain consents from third parties with respect to its compliance with this <u>Section 9.11</u>.
- (g) As soon as practicable, and in any event not later than five (5) Business Days after the request is made by the Required Lenders, each Credit Party shall execute and deliver to the Administrative Agent notices of the Final DIP Order, abstracts of the Final DIP Order or any other document reasonably requested by the Administrative Agent or the Required Lenders in connection with recording or registering any security interests or Liens against any Real Property of the Credit Parties.

- 9.12 <u>Pleadings</u>. The Borrower will furnish the counsel to the Administrative Agent and counsel to the Commitment Parties no later than two Business Days prior to filing or distribution, with copies of all pleadings, motions, applications, judicial information, financial information and other documents to be filed by or on behalf of any Credit Party with the Bankruptcy Court or the Chapter 11 Cases, or to be distributed by or on behalf of any Credit Party to any official committee appointed in the Chapter 11 Cases (except that with respect to emergency pleadings, motions or other filings for which, despite such Credit Party's commercially reasonable efforts, two Business Days' notice is impracticable, Borrower shall be required to furnish the same no later than concurrently with such filing or distribution thereof, as applicable).
- 9.13 <u>Bi-Weekly Calls</u>. At the request of the Required Lenders, the Borrower will hold bi-weekly calls, at which counsel or any other advisor to the Commitment Parties are allowed to participate to discuss (i) the status of and updates to the Borrower's business, and (ii) and address issues or matters that are in scope or nature consistent with the issues or matters addressed on past bi-weekly calls (if any).
- 9.14 <u>Cooperation</u>. Borrower shall and shall cause its Domestic Subsidiaries to, at Borrower and its Domestic Subsidiaries' sole expense, use commercially reasonable efforts to provide, at the request of the Buyer, all cooperation in connection with Buyer's arrangement of debt financing to be entered into on or around the APA Closing Date. Such cooperation by the Borrower shall include, at the request of the Buyer, (a) providing to the lenders of the Buyer financial and other information in the Borrower's possession to consummate such debt financing, (b) making the Borrower's and its Subsidiaries' officers available to assist lenders of the Buyer (including by participating in a reasonable number of meetings, presentations, due diligence sessions, sessions with rating agencies and reasonably cooperating with the marketing efforts of the Buyer and lenders of the Buyer), (c) assisting with the preparation of materials for syndication documents (including bank information memoranda and lender and investor presentations), business projections and similar documents required in connection with such financing and other documents required in connection with obligation such financing, (d) cooperating reasonably with the due diligence by lenders of the Buyer and (e) otherwise reasonably cooperating in connection with the consummation of such financing.
- 9.15 <u>Cash Management</u>. The Credit Parties shall maintain their cash management system approved by the Bankruptcy Court; <u>provided</u>, that the Borrower shall only withdraw or transfer from its accounts amounts necessary to fund expenses of the Credit Parties which do not violate the budget covenant set forth in <u>Section 10.10</u>. Any material changes from such cash management system must be acceptable to the Required Lenders.

## SECTION 10 Negative Covenants.

The Borrower hereby covenants and agrees that, until the Total Commitments have been terminated and the Loans, Fees and all other Obligations (other than contingent obligations for which no claim has been asserted) incurred hereunder, are paid in full:

10.01 <u>Liens</u>. The Borrower will not, and will not permit any of its Domestic Subsidiaries to, create, incur, assume or suffer to exist any Lien upon or with respect to any

property or assets (real or personal, tangible or intangible) of the Borrower or any of its Domestic Subsidiaries, whether now owned or hereafter acquired, or sell any such property or assets subject to an understanding or agreement, contingent or otherwise, to repurchase such property or assets (including sales of accounts receivable with recourse to the Borrower or any of its Domestic Subsidiaries), or assign any right to receive income or permit the filing of any financing statement under the UCC or any other similar notice of Lien under any similar recording or notice statute; <u>provided</u> that the provisions of this <u>Section 10.01</u> shall not prevent the creation, incurrence, assumption or existence of the following (Liens described below are herein referred to as "<u>Permitted Liens</u>"):

- (a) Liens for taxes, assessments or governmental charges or levies which are (i) not yet due, (ii) being contested in good faith and by appropriate proceedings for which adequate reserves have been established to the extent required by and in accordance with GAAP or (iii) stayed by the filing of the Chapter 11 Cases;
- (b) carriers', warehousemen's, materialmen's and mechanics' liens and other similar Liens arising in the ordinary course of business, which are (i) not overdue for a period of more than 45 days or which are being contested in good faith and by appropriate proceedings diligently conducted, if adequate reserves with respect thereto are maintained on the books of the applicable Person to the extent required with GAAP or (ii) stayed by the filing of the Chapter 11 Cases;
  - (c) Liens in existence on the Petition Date;
- (d) Liens created or permitted by or pursuant to this Agreement, Cash Collateral Order, the Final DIP Order or the Security Documents;
- (e) (a) licenses, sublicenses, leases or subleases granted by the Borrower or any of its Subsidiaries to other Persons, which do not (x) interfere in any material respect with the business of Borrower or any of its Subsidiaries or (y) secure any Indebtedness for borrowed money, and (b) the rights reserved or vested in any Person by the terms of any license, sublicense, lease or sublease held by the Borrower or any of its Subsidiaries or by a statutory provision, to terminate any such license, sublicense, lease or sublease, or to require annual or periodic payments as a condition to the continuance thereof;
- (f) Liens upon assets of the Borrower or any of its Subsidiaries subject to Capitalized Lease Obligations to the extent such Capitalized Lease Obligations are permitted by Section 10.04(d), provided that (x) such Liens only serve to secure the payment of Indebtedness arising under such Capitalized Lease Obligation and (y) the Lien encumbering the asset giving rise to the Capitalized Lease Obligation does not encumber any other asset of the Borrower or any Subsidiary of the Borrower;
- (g) easements, rights-of-way, restrictions, zoning and other similar restrictions, encroachments and other similar charges or encumbrances, and minor title deficiencies, in each case not securing Indebtedness and not materially interfering with the conduct of the business of the Borrower or any of its Subsidiaries;

- (h) Liens arising from precautionary UCC financing statement filings (or similar filings under other applicable Law) regarding operating leases entered into in the ordinary course of business and permitted under this Agreement;
- (i) statutory and common law landlords' liens under leases to which the Borrower or any of its Subsidiaries is a party;
- (j) Liens (other than Liens imposed under ERISA) incurred in the ordinary course of business in connection with workers' compensation claims, environmental matters, unemployment insurance and social security and other employment benefits and Liens securing the performance of bids, tenders, leases and contracts in the ordinary course of business, statutory obligations, surety bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business (exclusive of obligations in respect of the payment for borrowed money);
- (k) Liens arising out of any conditional sale, title retention, consignment or other similar arrangements for the sale of goods entered into by the Borrower or any of its Subsidiaries in the ordinary course of business to the extent such Liens do not attach to any assets other than the goods subject to such arrangements;
- (l) Liens (x) incurred in the ordinary course of business in connection with the purchase or shipping of goods or assets (or the related assets and proceeds thereof), which Liens are in favor of the seller or shipper of such goods or assets and only attach to such goods or assets, and (y) in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods;
- (m) bankers' Liens, rights of setoff and other similar Liens existing solely with respect to cash and Cash Equivalents on deposit in one or more bank, custodian, investment, customs and other accounts maintained by the Borrower or any Subsidiary, in each case granted in the ordinary course of business in favor of the bank or banks with which such accounts are maintained, securing amounts owing to such bank or banks with respect to cash management and operating account arrangements;
- (n) Liens granted in the ordinary course of business on the unearned portion of insurance premiums securing the financing of insurance premiums to the extent the financing is permitted under Section 10.04; and
  - (o) Liens incurred in respect of Indebtedness permitted under Section 10.04(k).

Notwithstanding anything to the contrary herein, (a) the designation in any Credit Document of any Lien as a "Permitted Lien" is not, and shall not be deemed to be, an acknowledgment by any Secured Creditor that that any such Permitted Lien has priority over the Secured Creditors' Liens securing the Obligations, and (b) any reference in any Credit Document to "subject to Permitted Liens" or "other than Permitted Liens" shall not be construed to be a subordination or postponement of any Lien of any Secured Creditor to any holder of a Permitted Lien, nor shall such reference elevate the priority of any Permitted Lien above the level it would otherwise have under applicable law.

- 10.02 <u>Fundamental Changes</u> Other than as contemplated by the Sale Motion, the Borrower will not, and will not permit any of its Domestic Subsidiaries to wind up, liquidate or dissolve its affairs or enter into any transaction of merger, amalgamation or consolidation, convey, sell, lease or otherwise dispose of all or substantially all of its assets (whether now owned or hereafter acquired) to or in favor of any Person, except that, subject to the Commitment Parties' consent in their sole discretion:
  - (a) any Subsidiary of the Borrower that is a Credit Party may be merged, amalgamated or consolidated with or into the Borrower, <u>provided</u> that (a) no Default or Event of Default shall have occurred, be continuing or result therefrom, (b) both before and after giving effect to such transaction, the Borrower is in pro forma compliance with the financial covenants set forth in <u>Section 10.10</u>, (c) all of the transactions contemplated by this <u>Section 10.02(a)</u>, and the terms, conditions and documentation thereof shall be in form and substance satisfactory to the Administrative Agent in its sole discretion, (d) at least 10 Business Days' prior written notice of such transaction is given by the Borrower to the Administrative Agent, (e) any security interests granted to the Collateral Agent for the benefit of the Secured Creditors pursuant to the Security Documents and the Final DIP Order in the assets of such Subsidiary shall remain in full force and effect and perfected (to at least the same extent as in effect immediately prior to such merger, amalgamation or consolidation) and all actions required to maintain said perfected status have been taken and (f) the Borrower shall be the continuing or surviving corporation;
  - (b) any Subsidiary of the Borrower that is a Credit Party may merge, amalgamate or consolidate with and into any Wholly-Owned Domestic Subsidiary of the Borrower which is a Guarantor, so long as any security interests granted to the Collateral Agent for the benefit of the Secured Creditors pursuant to the Security Documents and the Final DIP Order in the assets of such Subsidiary shall remain in full force and effect and perfected (to at least the same extent as in effect immediately prior to such merger, consolidation, dissolution or liquidation) and all actions required to maintain said perfected status have been taken;
  - (c) any Subsidiary of the Borrower that is not a Credit Party may be merged, consolidated or amalgamated with and into, or transfer any of its assets to, any Wholly-Owned Subsidiary of the Borrower which is not a Guarantor; and
  - (d) any Subsidiary of the Borrower that is a Credit Party may Dispose of all or substantially all of its assets (upon voluntary liquidation or otherwise) to the Borrower or another Wholly-Owned Domestic Subsidiary of the Borrower which is a Guarantor, and any Subsidiary of the Borrower which is not a Credit Party may Dispose of all or substantially all of its assets (upon voluntary liquidation or otherwise) to another Subsidiary of the Borrower which is not a Credit Party.
- 10.03 <u>Restricted Payments</u>. The Borrower will not, and will not permit any of its Domestic Subsidiaries to, authorize, declare or make, directly or indirectly, any Dividends, or incur any obligation (contingent or otherwise) to do so, except that each Domestic Subsidiary of the Borrower may pay cash Dividends to the Borrower or to another Domestic Subsidiary of the Borrower (and, in the case of a Dividend by a non-wholly owned Domestic Subsidiary to the

Borrower or to another Domestic Subsidiary of the Borrower and to each other owner of Equity Interests of such Domestic Subsidiary based on their relative ownership interests of the relevant class of Equity Interests).

- 10.04 <u>Indebtedness</u>. The Borrower will not, and will not permit any of its Domestic Subsidiaries to, contract, create, incur, assume or suffer to exist any Indebtedness, except:
  - (a) Indebtedness incurred pursuant to this Agreement, the Final DIP Order or the other Credit Documents;
    - (b) Indebtedness outstanding on the Petition Date;
  - (c) Indebtedness of the Borrower and the other Credit Parties under Hedging Agreements entered into in the ordinary course of business and consistent with past practices and providing protection to the Borrower and its Domestic Subsidiaries against fluctuations in currency values or commodity prices in connection with the Borrower's or such Domestic Subsidiaries' operations, in each case, so long as the entering into of such Hedging Agreements are bona fide hedging activities and are not for speculative purposes;
  - (d) Indebtedness of the Borrower and its Domestic Subsidiaries evidenced by Capitalized Lease Obligations and purchase money Indebtedness, <u>provided</u> that in no event shall the sum of the aggregate principal amount of all Capitalized Lease Obligations and purchase money Indebtedness entered into after the Effective Date permitted by this clause (d) exceed \$100,000;
  - (e) Indebtedness constituting Intercompany Loans to the extent permitted by Section 10.05(g);
  - (f) Indebtedness consisting of unsecured guaranties by the Borrower and the Subsidiaries of the Borrower that are Guarantors of each other's Indebtedness and lease and other contractual obligations permitted under this Agreement;
  - (g) Indebtedness in respect of overdraft facilities, employee credit card programs, netting services, automated clearinghouse arrangements and other cash management and similar arrangements in the ordinary course of business and consistent with past practice;
  - (h) Indebtedness incurred in the ordinary course of business consistent with past practice in respect of obligations of the Borrower or any of its Domestic Subsidiaries to pay the deferred purchase price of goods or services or progress payments in connection with such goods and services; provided that such obligations are incurred in connection with open accounts extended by suppliers on customary trade terms in the ordinary course of business and not in connection with the borrowing of money or Hedging Agreements;
  - (i) Indebtedness of the Borrower or any Domestic Subsidiary consisting of (i) obligations to pay insurance premiums or (ii) take or pay obligations contained in supply agreements, in each case arising in the ordinary course of business and consistent

with past practice, and not in connection with the borrowing of money or Hedging Agreements;

- (j) Indebtedness representing deferred compensation to employees of the Borrower (or any direct or indirect parent thereof) and its Domestic Subsidiaries incurred in the ordinary course of business and consistent with past practice; and
  - (k) Indebtedness in respect of any Approved Collateralized Obligations.
- 10.05 <u>Investments</u>. The Borrower will not, and will not permit any of its Domestic Subsidiaries to, directly or indirectly, lend money or credit or make advances to any Person, or purchase or acquire any stock, obligations or securities of, or any other Equity Interest in, or make any capital contribution to, any other Person, or purchase or own a futures contract or otherwise become liable for the purchase or sale of currency or other commodities at a future date in the nature of a futures contract, or hold any cash or Cash Equivalents (each of the foregoing an "<u>Investment</u>" and, collectively, "<u>Investments</u>"), except that the following shall be permitted:
  - (a) the Borrower and its Subsidiaries may acquire and hold accounts receivables owing to any of them, if created or acquired in the ordinary course of business and payable or dischargeable in accordance with customary trade terms of the Borrower or such Subsidiary;
  - (b) the Borrower and its Subsidiaries may acquire and hold cash and Cash Equivalents;
  - (c) the Borrower and its Subsidiaries may hold the Investments held by them on the Petition Date; <u>provided</u> that any additional Investments made with respect thereto shall be permitted only if permitted under the other provisions of this <u>Section 10.05</u>;
  - (d) the Borrower and its Subsidiaries may acquire and own investments (including debt obligations) received in connection with the bankruptcy or reorganization of suppliers and customers and in good faith settlement of delinquent obligations of, and other disputes with, customers and suppliers arising in the ordinary course of business;
  - (e) the Borrower and its Subsidiaries may make loans and advances to their officers, directors and employees for moving, relocation and travel expenses and other similar expenditures, in each case in the ordinary course of business and consistent with past practice in an aggregate amount not to exceed \$50,000 at any time (determined without regard to any write-downs or write-offs of such loans and advances);
  - (f) the Borrower and the other Credit Parties may enter into Hedging Agreements to the extent permitted by <u>Section 10.04(c)</u>;
  - (g) (i) any Credit Party may make intercompany loans and advances to any other Credit Party, (ii) any Subsidiary which is not a Credit Party may make intercompany loans and advances to any Credit Party and (iii) any Subsidiary which is not a Credit Party may make intercompany loans and advances to any other Subsidiary

which is not a Credit Party (such intercompany loans and advances referred to in preceding clauses (i) through (iii), collectively, the "Intercompany Loans"), provided, that (v) no Default or an Event of Default then exists or would result therefrom, (w) each Intercompany Loan shall be evidenced by an Intercompany Note, (x) each such Intercompany Note owned or held by a Credit Party shall be pledged to the Collateral Agent as Collateral, (y) each Intercompany Loan made by any Subsidiary of the Borrower that is not a Credit Party to a Credit Party shall be subject to the subordination provisions contained in the respective Intercompany Note and (z) any Intercompany Loans made to any Guarantor pursuant to this clause (g) shall cease to be permitted by this clause (g) if such Guarantor ceases to constitute a Guarantor that is a Wholly-Owned Domestic Subsidiary;

## (h) [reserved];

- (i) the Borrower and its Subsidiaries may own the Equity Interests of their respective Subsidiaries created or acquired in accordance with the terms of this Agreement (so long as all amounts invested in such Subsidiaries are independently justified under another provision of this Section 10.05);
- (j) the Borrower and its Subsidiaries may receive and hold promissory notes and other non-cash consideration received in connection with any asset sale permitted by Section 10.07;
- (k) the Borrower and its Subsidiaries may make advances in the form of a prepayment of expenses to vendors, suppliers and trade creditors consistent with their past practices, so long as such expenses were incurred in the ordinary course of business of the Borrower or such Subsidiary and consistent with the Approved Budget; and
- (1) the Borrower and its Subsidiaries may make payments for the benefit of Black Warrior Methane Corp. and Black Warrior Transmission Corp., subject in all respects to the Cash Collateral Order.
- 10.06 <u>Transactions with Affiliates</u>. The Borrower will not, and will not permit any of its Domestic Subsidiaries to, enter into any transaction or series of related transactions with any Affiliate of the Borrower or any of its Domestic Subsidiaries, other than (i) transactions among Borrower and any Wholly Owned Domestic Subsidiary that is a Credit Party and (ii) transactions among the Subsidiaries of the Borrower which are not Credit Parties, except that the following in any event shall be permitted:
  - (a) Dividends may be paid to the extent provided in <u>Section 10.03</u>;
  - (b) customary fees, indemnities and reimbursements may be paid to non-officer directors of the Borrower and the other Credit Parties;
  - (c) the Borrower and its Subsidiaries may enter into, and may make payments under, employment agreements, severance agreements, employee benefits plans, stock purchase plans, stock option plans, indemnification provisions and other similar compensatory arrangements with officers, employees and directors of the Borrower and

its Domestic Subsidiaries in the ordinary course of business and consistent with past practice; <u>provided</u>, <u>however</u>, no funds of any Credit Party shall be used to make any of the foregoing payments on behalf of any Subsidiary that is not a Credit Party;

- (d) any sale of equipment expressly permitted under the Asset Purchase Agreement; and
  - (e) Investments permitted under Section 10.05(l).
- 10.07 <u>Asset Sales</u>. The Borrower will not, and will not permit any of its Domestic Subsidiaries to effectuate any Asset Sale, except as contemplated by the Sale Motion, or subject to the Commitment Parties' consent in their sole discretion:
  - (a) the Borrower and its Subsidiaries may liquidate or otherwise dispose of obsolete surplus or worn-out property (including IP Rights no longer material to the business) in the ordinary course of business and consistent with past practice;
  - (b) sale of inventory in the ordinary course of business and consistent with past practice;
    - (c) sale of property by any Credit Party to another Credit Party;
  - (d) sale of property by any Subsidiary that is not a Credit Party to another Subsidiary that is not a Credit Party,
  - (e) each of the Borrower and its Subsidiaries may sell or discount, in each case, without recourse, in the ordinary course of business and consistent with past practice, overdue accounts receivable arising in the ordinary course of business, but only in connection with the compromise or collection thereof and not as part of any financing transaction;
  - (f) any leasing or subleasing, or any non-exclusive licensing or sublicensing, of real, personal or intellectual property in the ordinary course of business and consistent with past practice;
  - (g) the Borrower and its Subsidiaries may liquidate or otherwise dispose of Cash Equivalents in the ordinary course of business, in each case, for cash at fair market value;
    - (h) any Asset Sale permitted by Section 10.02;
  - (i) any sale of equipment expressly permitted under the Asset Purchase Agreement; and
  - (j) any single Asset Sale (or series of related Asset Sales) not to exceed \$25,000, <u>provided</u> that the fair market value of all Asset Sales made pursuant to this clause (j) shall not exceed \$500,000 in the aggregate;

<u>provided</u>, <u>however</u>, that any sale of assets pursuant to this <u>Sections 10.07</u> shall be for fair market value, as determined in good faith by the Borrower, other than Asset Sales set forth in clauses (a), (c), (d), and (e) of this <u>Section 10.07</u> so long as such Asset Sales are in the ordinary course of business and consistent with past practices.

- 10.08 Limitation on Certain Restrictions on Subsidiaries. The Borrower will not, and will not permit any of its Domestic Subsidiaries to, directly or indirectly, create or otherwise cause or suffer to exist or become effective any encumbrance or restriction on the ability of any such Subsidiary to (a) make Dividends or make any other distributions on its capital stock or any other Equity Interest or participation in its profits owned by the Borrower or any of its Subsidiaries, (b) make loans or advances to the Borrower or any of its Subsidiaries or (c) create, incur, assume or suffer to exist any Lien on properties or assets of the Borrower or any of its Subsidiaries, except for such encumbrances or restrictions existing under or by reason of (i) applicable law, (ii) this Agreement, the other Credit Documents or the Asset Purchase Agreement, (iii) customary provisions restricting subletting or assignment of any lease governing any leasehold interest of the Borrower or any of its Subsidiaries, (iv) customary provisions restricting assignment of any licensing agreement (in which Borrower or any of its Subsidiaries is the licensee) or other contract entered into by the Borrower or any of its Subsidiaries in the ordinary course of business, and (v) restrictions in existence as of the Effective Date.
- 10.09 <u>Business</u>; etc. Borrower will not, and will not permit any of its Domestic Subsidiaries to, engage directly or indirectly in any business other than the businesses engaged in by the Borrower and its Subsidiaries as of the Effective Date and reasonable extensions thereof and businesses ancillary or complimentary thereto.
- 10.10 <u>Financial Covenants</u>. Borrower will not, and will not permit any of its Domestic Subsidiaries to:

#### (a) Budget Variance.

- (i) permit the aggregate amount of the actual net cash flow of the Debtors of the type set forth in the line item "Cumulative Net Cash Flow" on the Approved Budget for any Budget Testing Period to be lower than the budgeted amount set forth on the line item "Cumulative Net Cash Flow" on the Approved Budget for such Budget Testing Period by more than \$10,000,000, provided, that "Cumulative Net Cash Flow" shall not include (w) fees and expenses of the Steering Committee Advisors, the Consultants, and the professionals retained by the Commitment Parties, and fees and expenses of professionals retained by the Administrative Agent, the Prepetition Administrative Agent, the Prepetition Trustee and the Debtors, (x) Cash Collateral provided with respect to Approved Collateralized Obligations, (y) key employee retention payments approved by both the Required Lenders and the Bankruptcy Court and (z) any other amounts excluded from "Cumulative Net Cash Flow" in the Cash Collateral Order;
- (ii) permit the aggregate amount of the actual disbursements of the Debtors of the type set forth in the line item "Cumulative Disbursements" on the

Approved Budget for any Budget Testing Period to be greater than the budgeted amount set forth on the line item "Cumulative Disbursements" on the Approved Budget for such Budget Testing Period by more than the greater of (x) \$5,000,000 and (y) 5.0% of such budgeted amount; provided, that "Cumulative Disbursement" shall not include (w) fees and expenses of the Steering Committee Advisors, the Consultants, and the professionals retained by the Commitment Parties, and fees and expenses of professionals retained by the Administrative Agent, the Prepetition Administrative Agent, the Prepetition Trustee and the Debtors, (x) Cash Collateral provided with respect to Approved Collateralized Obligations, (y) key employee retention payments approved by both the Required Lenders and the Bankruptcy Court and (z) any other amounts excluded from "Cumulative Disbursements" or "Cumulative Net Cash Flow" in the Cash Collateral Order.

- (b) <u>Capital Expenditures</u>. allow the cumulative amount of the actual capital expenditures, as calculated on a GAAP basis, for any Capital Expenditure Testing Period to exceed the budgeted amount set forth in the Capital Expenditure Budget for such Capital Expenditure Testing Period, also as calculated on a GAAP basis, by more than \$1,500,000.
- 10.11 <u>Amendments of Organization Documents</u>. Borrower will not, and will not permit any of its Domestic Subsidiaries to, directly or indirectly, amend, terminate or waive (or permit any amendment, termination or waiver of) the Organization Documents of the Borrower or any of its Subsidiaries, without the prior written consent of the Commitment Parties in their sole discretion.
- 10.12 <u>Chapter 11 Cases</u>. Borrower will not, and will not permit any of its Domestic Subsidiaries to, directly or indirectly:
- (a) Except for the Carve-Out, incur, create, assume, suffer to exist or permit any other superpriority claim which is *pari passu* with or senior to the claims of the Administrative Agent, the Collateral Agent and the Lenders against the Credit Parties or the Adequate Protection Liens or the Prepetition Secured Parties' Superpriority Claims, if any, except as set forth in, or permitted by, the Final DIP Order;
- (b) Without limiting subsection (c) below, make or permit to be made any change to the Final DIP Order without the consent of the Required Lenders in their sole discretion; or
- (c) File an application for the approval of any superpriority claim or any Lien in any of the Chapter 11 Cases that is *pari passu* with or senior to the Administrative Agent's, the Collateral Agent's and the Lenders' superpriority claims or Liens securing the Obligations, except as permitted under the Final DIP Order.
- 10.13 <u>Prepayments, Etc. of Indebtedness</u>. Borrower will not, and will not permit any of its Domestic Subsidiaries to, directly or indirectly:
- (a) Prepay redeem, purchase, defease or otherwise satisfy prior to the scheduled maturity thereof in any manner, or make any payment in violation of any subordination terms of, any Indebtedness that is contractually subordinated to the Obligations; or

- (b) Make (i) any prepetition "critical vendor" payments or other payments on account of any creditor's prepetition unsecured claim, (ii) payments on account of claims or expenses arising under Section 503(b)(9) of the Bankruptcy Code, (iii) payments under any management incentive plan or on account of claims or expenses arising under Section 503(c) of the Bankruptcy Code, and (iv) payments on account of any prepetition Indebtedness, except in each case, in amounts and on terms and conditions authorized by order of the Bankruptcy Court on or prior to the Effective Date or with the consent of the Commitment Parties in their sole discretion.
- 10.14 <u>Sale Leasebacks</u>. Borrower will not, and will not permit any of its Domestic Subsidiaries to, engage, directly or indirectly, in any sale leaseback transactions involving any of the assets of the Borrower or its Subsidiaries.
- 10.15 <u>Anti-Terrorism Laws</u>. Borrower will not, and will not permit any of its Subsidiaries to, directly or indirectly:
- (a) Conduct any business or engage in any transaction or dealing with any Blocked Person, including the making or receiving any contribution of funds, goods or services to or for the benefit of any "blocked person" as described in the Executive Order, the Trading With the Enemy Act or the Foreign Assets Control Regulations;
- (b) Deal in or otherwise engage in any transaction relating to, any property or interests in property blocked pursuant to Foreign Assets Control Regulations, Trading With the Enemy Act or the Executive Order; or
- (c) Engage in or conspire to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in Foreign Assets Control Regulations, Trading With the Enemy Act, the Executive Order or the Patriots Act.
- 10.16 <u>Intercompany Limitation</u>. Notwithstanding anything to the contrary herein, Borrower will not, and will not permit any Guarantor to, send, transfer or otherwise provide any funds, assets or property (either in the form of Dividends, Investments, guarantees of obligations, repayments of intercompany liabilities (including interest thereon), Asset Sales, provisions of services or otherwise) to any Subsidiary of the Borrower that is not a Guarantor.
- SECTION 11 Events of Default. Upon the occurrence of any of the following specified events (each, an "Event of Default"):
- 11.01 <u>Payments</u>. The Borrower or any Credit Party shall default in the payment when due of any principal or interest on any Loan or any Fees or any other amounts owing hereunder or under any other Credit Document; or
- 11.02 <u>Representations, etc.</u> Any representation, warranty or statement made or deemed made by any Credit Party in or in connection with any Credit Document or the borrowings hereunder, or any representation, warranty or statement or information contained in any report, certificate, financial statement or other instrument furnished by a Credit Party in connection with or pursuant to any Credit Document, shall have been false or misleading in any material respect

on the date as of which made, deemed made or furnished (or in the case of any representation or warranty that is qualified as to "materiality" or "Material Adverse Effect", shall prove to have been false or misleading in any respect when so made, deemed made or furnished); or

- 11.03 <u>Covenants</u>. The Borrower or any of its Subsidiaries shall (i) default in the due performance or observance by it of any term, covenant or agreement contained in <u>Sections 9.01(e)</u>, 9.01(g)(i), (ii) or (v), 9.04(i), 9.08, 9.10, 9.11, 9.12 or <u>Section 10</u> or (ii) default in the due performance or observance by it of any other term, covenant or agreement contained in this Agreement or any other Credit Document (other than those set forth in <u>Sections 11.01</u> and <u>11.02</u>) and such default shall continue unremedied for a period of three (3) days; or
- 11.04 <u>Default Under Other Agreements</u>. (i) The Borrower or any of its Subsidiaries shall (x) default in any payment when due (subject to any applicable grace period) which is not stayed by the filing of the Chapter 11 Cases, whether by acceleration or otherwise of any Indebtedness (other than the Obligations) of the Borrower or any of its Subsidiaries or (y) default in the observance or performance of any agreement or condition relating to any Indebtedness (other than the Obligations) which is not stayed by the filing of the Chapter 11 Cases or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event shall occur or condition exist, the effect of which is to accelerate the maturity of such Indebtedness, or (ii) any Indebtedness (other than the Obligations), the payment of which is not stayed by the filing of the Chapter 11 Cases shall be declared to be (or shall become) due and payable, or required to be prepaid, other than by a regularly scheduled required prepayment, prior to the stated maturity thereof, <u>provided</u> that it shall not be a Default or an Event of Default under this <u>Section 11.04</u> unless the aggregate principal amount of all Indebtedness as described in preceding clauses (i) and (ii) is at least \$500,000.
- 11.05 <u>Bankruptcy</u>, <u>etc.</u> Other than the Canadian Entities, any Subsidiary of the Borrower that is not a Debtor commences any proceeding under any bankruptcy or insolvency proceeding, or there is commenced against such Subsidiary any such proceeding which remains undismissed for a period of 30 days after the filing thereof, or any such Subsidiary is adjudicated insolvent or bankrupt; or any order of relief or other order approving any such case or proceeding is entered; or any such Subsidiary makes a general assignment for the benefit of creditors; or any Company action is taken by the Borrower or any of its Subsidiaries for the purpose of effecting any of the foregoing; or

## 11.06 [ERISA.

- (a) One or more ERISA Events shall have occurred, or
- (b) there is or arises any potential withdrawal liability under Section 4201 of ERISA, if the Borrower or the ERISA Affiliates were to withdraw completely from any and all Multiemployer Plans;

and the liability of the Borrower and the ERISA Affiliates contemplated by the foregoing clauses (a) and (b), either individually or in the aggregate, has had or would be reasonably expected to have, a Material Adverse Effect;] or

- 11.07 <u>Security Documents</u>. Any of the Security Documents or the Final DIP Order shall cease to be in full force and effect, or shall cease to give the Collateral Agent for the benefit of the Secured Creditors the Liens, rights, powers and privileges purported to be created thereby (including, without limitation, a perfected security interest in, and Lien on, all of the Collateral, in favor of the Collateral Agent, superior to and prior to the rights of all third Persons (except as permitted by <u>Section 10.01</u>), and subject to no other Liens (except as permitted by <u>Section 10.01</u>); or
- 11.08 <u>Guaranties</u>. Any Guaranty or any provision thereof set forth in Section 13 shall cease to be in full force or effect as to any Guarantor (except as a result of a release of any Guarantor in accordance with Section 14.11), or any Guarantor or any Person acting for or on behalf of such Guarantor shall deny or disaffirm such Guarantor's obligations under this Agreement or any other Credit Document; or
- 11.09 <u>Judgments</u>. One or more judgments or decrees shall be entered against the Borrower or any Subsidiary of the Borrower which is not stayed by the filing of the Chapter 11 Cases or otherwise involving in the aggregate for the Borrower and its Subsidiaries a liability (not paid or to the extent not covered by a reputable and solvent insurance company) and such judgments and decrees either shall be final and non-appealable or shall not be vacated, discharged or stayed or bonded pending appeal for any period of 30 consecutive days, and the aggregate amount of all such judgments equals or exceeds \$500,000; or
  - 11.10 Change of Control. A Change of Control shall occur;
- 11.11 <u>Sale of all or Substantially all Assets</u>. Other than as contemplated by the Sale Motion, there occurs a sale of all or substantially all assets of the Borrower and its Subsidiaries;

#### 11.12 Chapter 11 Cases.

- (a) (i) Any Chapter 11 Case shall be dismissed or converted to a case under chapter 7 of the Bankruptcy Code or any Debtor shall file a motion or other pleading seeking the dismissal or conversion of any Chapter 11 Case pursuant to Bankruptcy Code section 1112 or otherwise; or (ii) a trustee, responsible officer, or an examiner (other than a fee examiner) pursuant to Bankruptcy Code section 1104 shall be appointed or elected, as applicable, in any Chapter 11 Case, any Debtor shall apply for, consent to, or acquiesce in, any such appointment, or the Bankruptcy Court shall enter an order providing for such appointment;
- (b) Any Debtor shall file a motion or application for the approval of any superpriority claim, lien or security interest in the Chapter 11 Cases (other than the Carve-Out) that is *pari passu* with, or senior to claims, liens and interests of the Administrative Agent, the Collateral Agent and the Lenders;
- (c) Any Debtor creates or incurs, or the Bankruptcy Court enters an order granting, any claim, lien or security interest that is *pari passu* with, or senior to, any of the liens, claims or security interests granted to the Lenders, the Administrative Agent and the Collateral Agent;

- (d) Without the consent of the Required Lenders in their sole discretion, an order shall be entered reversing, staying, vacating, extending, supplementing, modifying or amending the Final DIP Order, or any Debtor shall file an application, motion or other pleading, or shall consent to the filing of such application, motion or other pleading seeking the foregoing;
- (e) Without the consent of the Required Lenders in their sole discretion, the date any provision of the Final DIP Order for any reason ceases to be valid and binding or any Debtor so asserts in any pleading filed in any court;
- (f) The Debtors, subject to permitted variances and <u>Section 10.10</u> of this Agreement, shall not use the Loans and Collateral in a manner consistent with the Approved Budget and otherwise comply in any respect with any provision of the Final DIP Order;
- (g) the Bankruptcy Court shall enter an order or orders granting relief from the automatic stay to the holder or holders of any security interest to proceed against, including foreclosure (or the granting of a deed in lieu of foreclosure or the like) on any of the Debtor's assets (other than in respect of insurance proceeds or with respect to assets having a fair market value of less than \$1,000,000);
- (h) Following the entry of the Final DIP Order, an order of the Bankruptcy Court (or any other court of competent jurisdiction) shall be entered against any Lender regarding the Loans that has an adverse effect on such Lender's rights and remedies hereunder; or any Debtor shall file a motion or other pleading or shall consent to a motion or other pleading filed by any other Person seeking the foregoing;
- (i) Debtor's use or support of any portion of the Loans or Collateral to challenge the validity, perfection, priority, extent or enforceability of the Loans or the prepetition obligation or the Liens on the assets of the Debtors securing the Loans or the prepetition obligations;
- (j) Any Debtor's support of any investigation or assertion of any claims against any Lender or any Agent;
- (k) Filing by any Debtor of any motion or proceeding, or the entry of any order by the Bankruptcy Court, which could have material impairment of the Lender's rights under the Credit Documents;
- (l) An order shall be entered reversing, staying, vacating, terminating, modifying or amending the Sale Order, in each case without the prior written consent of the Required Lenders in their sole discretion, or any Debtor shall file a motion or other pleading or shall consent to a motion or other pleading filed by any other Person seeking the foregoing;
- (m) Any breach of, or failure to perform by Debtors of any of their agreements, covenants, representations or warranties contained in the Asset Purchase Agreement or in

the Sale Order, which continues unremedied or uncured beyond any applicable notice or grace period set forth therein;

- (n) The termination of the Asset Purchase Agreement pursuant to Section 11 thereof prior to the APA Closing Date;
- (o) Payment of or granting adequate protection with respect to prepetition claims other than as approved by the Bankruptcy Court on or prior to the Effective Date or provided for in the Final DIP Order; or any Debtor shall file a motion or other pleading or shall consent to a motion or other pleading filed by any other Person seeking the foregoing;
- (p) A final non-appealable order of the Bankruptcy Court shall be entered that provides for the recovery from any portions of the Collateral of any costs or expenses of preserving or disposing of such Collateral under section 506(c) of the Bankruptcy Code; or any Credit Party shall bring a motion in the Chapter 11 Cases seeking, or otherwise consent to, authority from the Bankruptcy Court (i) to recover from any portions of the Collateral any costs or expenses of preserving or disposing of such Collateral under section 506(c) of the Bankruptcy Code or (ii) to effect any other action or actions (x) materially adverse to the Administrative Agent, the Collateral Agent or the Lenders or their rights and remedies hereunder or their interest in the Collateral or (y) inconsistent with the Credit Documents in any material respect;
- (q) The commencement of any adversary proceeding, contested matter or other action by any Credit Party or any other party either asserting any claims and defenses or otherwise against any of the Prepetition Secured Parties (as defined in the Final DIP Order) under the Prepetition Credit Agreement with respect to the obligations of any Credit Party thereunder or the liens granted to such agent to secure the obligations under the Prepetition Credit Agreement, except as may be permitted under the Final DIP Order; or
  - (r) The occurrence of any "Event of Default" under the Final DIP Order;

then, except as otherwise provided in the Final DIP Order, including any applicable notice period, if any Event of Default occurs, the Administrative Agent (at the direction of the Required Lenders) and the Lenders, may, exercise any remedies available to them hereunder, the Final DIP Order and applicable non-bankruptcy law, including, taking the following actions (without further order of, or application or motion to, the Bankruptcy Court (except as otherwise expressly provided in the Final DIP Order) or any other court, and without interference from any Credit Party or any other party in interest, at the same or different times), without prejudice to the rights of the Administrative Agent, the Collateral Agent or any Lender to enforce its claims against any Credit Party:

- (i) declare the Total Commitment terminated, whereupon all Commitments of each Lender shall forthwith terminate immediately;
- (ii) declare the principal of and any accrued interest in respect of all Loans and all Obligations owing hereunder and thereunder to be, whereupon the same shall

become, forthwith due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby waived by each Credit Party;

- (iii) after providing five (5) Business Days' notice in accordance with the Final DIP Order, enforce the guaranty and all of the Liens and security interests created pursuant to the Credit Documents and the Final DIP Order, and all remedies under applicable law (including, but not limited to, the Bankruptcy Code and the UCC), subject only to satisfaction of any notice requirement set forth in the Final DIP Order; and/or
- (iv) after providing five (5) Business Days' notice in accordance with the Final DIP Order, set-off any amounts held as Cash Collateral (including in any deposit accounts or securities accounts subject to a Deposit Account Control Agreement or a Securities Account Control Agreement).
- 11.13 <u>Application of Proceeds</u>. Subject to the Final DIP Order, including the payment of the Carve-Out, after the exercise of remedies by the Agent (at the direction of the Required Lenders) or any Lender pursuant to <u>Section 11</u>, any amounts received on account of the Obligations shall be applied by the Administrative Agent in the following order:
- (a) <u>First</u>, to the payment in full in cash of all costs and expenses, fees, commissions and taxes of such sale, collection or other realization including compensation to the Collateral Agent, the Administrative Agent or their respective agents, trustees and counsels, and all expenses, liabilities and advances made or incurred by the Collateral Agent or the Administrative Agent, as applicable in connection therewith and all other amounts for which the Collateral Agent or the Administrative Agent, as applicable, is entitled to indemnification and/or reimbursement, in each case, pursuant to the provisions of any Credit Document;
- (b) <u>Second</u>, to payment in full in cash of all expenses, liabilities and advances made or incurred by the Secured Creditors (including, without limitation, the reasonable fees and disbursements of outside counsel, special counsel, financial advisor and other advisors or consultants retained by each of the Administrative Agent, the Collateral Agent and the Commitment Parties as set forth in <u>Section 14.01</u>) and all other amounts for which the Secured Creditors are entitled to indemnification and/or reimbursement, in each case, pursuant to the provisions of any Credit Document;
- (c) <u>Third</u>, without duplication of amounts applied pursuant to clause (a) and (b) above, to the indefeasible payment in full in cash, *pro rata*, of interest and other amounts constituting Obligations (other than principal), any fees and premiums (if any), in each case, equally and ratably in accordance with the respective amounts thereof then due and owing;
- (d) <u>Fourth</u>, to the indefeasible payment in full in cash of principal amount of the Obligations, in each case, equally and ratably in accordance with the respective amounts thereof then due and owing; and
- (e) <u>Last</u>, the balance, if any, after all of the Obligations have been indefeasibly paid in full, to the Borrower or as otherwise required by Law.

In the event that any such proceeds are insufficient to pay in full the items described in clauses (a) through (e) of this <u>Section 11.13</u>, the Credit Parties shall remain liable, jointly and severally, for any deficiency.

## SECTION 12 The Administrative Agent.

- 12.01 <u>Appointment</u>. The Lenders hereby irrevocably designate and appoint Citibank, N.A., as Administrative Agent (for purposes of this <u>Section 12</u> and <u>Section 14.01</u>, the term "Administrative Agent" also shall include Citibank, N.A., in its capacity as Collateral Agent pursuant to the Security Documents) to act as specified herein and in the other Credit Documents. Each Lender hereby irrevocably authorizes the Administrative Agent to take such action on its behalf under the provisions of this Agreement, the other Credit Documents and any other instruments and agreements referred to herein or therein and to exercise such powers and to perform such duties hereunder and thereunder as are specifically delegated to or required of the Administrative Agent by the terms hereof and thereof and such other powers as are reasonably incidental thereto. The Administrative Agent may perform any of its respective duties hereunder by or through its officers, directors, agents, employees or affiliates.
- 12.02 Nature of Duties. The Administrative Agent shall not have any duties or responsibilities except those expressly set forth in this Agreement and in the other Credit Documents. The Lenders hereby acknowledge that the Administrative Agent shall not be under any duty to take any discretionary or optional action permitted to be taken by it pursuant to or in connection with this Agreement or any other Credit Document unless it shall be instructed in writing to do so by the Required Lenders. Neither the Administrative Agent nor any of its Related Parties shall be liable for any action taken or omitted by it or them hereunder or under any other Credit Document or in connection herewith or therewith, unless caused by its or their gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable decision). The duties of the Administrative Agent shall be mechanical and administrative in nature; the Administrative Agent shall not have by reason of this Agreement or any other Credit Document a fiduciary relationship in respect of any Lender; and nothing in this Agreement or in any other Credit Document, expressed or implied, is intended to or shall be so construed as to impose upon the Administrative Agent any obligations in respect of this Agreement or any other Credit Document except as expressly set forth herein or therein.
- 12.03 Lack of Reliance on the Administrative Agent. Independently and without reliance upon the Administrative Agent, each Lender, to the extent it deems appropriate, has made and shall continue to make (i) its own independent investigation of the financial condition and affairs of the Borrower and its Subsidiaries in connection with the making and the continuance of the Loans and the taking or not taking of any action in connection herewith and (ii) its own appraisal of the creditworthiness of the Borrower and its Subsidiaries and, except as expressly provided in this Agreement, the Administrative Agent shall not have any duty or responsibility, either initially or on a continuing basis, to provide any Lender with any credit or other information with respect thereto, whether coming into its possession before the making of the Loans or at any time or times thereafter. The Administrative Agent shall not be responsible to any Lender for any recitals, statements, information, representations or warranties herein or in any document, certificate or other writing delivered in connection herewith or for the execution, effectiveness, genuineness, validity, enforceability, perfection, collectability, priority or

sufficiency of this Agreement or any other Credit Document or the financial condition of the Borrower or any of its Subsidiaries or be required to make any inquiry concerning either the performance or observance of any of the terms, provisions or conditions of this Agreement or any other Credit Document, or the financial condition of the Borrower or any of its Subsidiaries or the existence or possible existence of any Default or Event of Default.

- 12.04 <u>Certain Rights of the Administrative Agent</u>. If the Administrative Agent requests instructions from the Required Lenders with respect to any act or action (including failure to act) in connection with this Agreement or any other Credit Document, the Administrative Agent shall be entitled to refrain from such act or taking such action unless and until the Administrative Agent shall have received instructions from the Required Lenders; and the Administrative Agent shall not incur liability to any Lender by reason of so refraining. Without limiting the foregoing, neither any Lender nor the holder of any Note shall have any right of action whatsoever against the Administrative Agent as a result of the Administrative Agent acting or refraining from acting hereunder or under any other Credit Document in accordance with the instructions of the Required Lenders.
- 12.05 <u>Reliance</u>. The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any note, writing, resolution, notice, statement, certificate, telex, teletype or telecopier message, cablegram, radiogram, order or other document or telephone message signed, sent or made by any Person that the Administrative Agent believed to be the proper Person, and, with respect to all legal matters pertaining to this Agreement and any other Credit Document and its duties hereunder and thereunder, upon advice of counsel selected by the Administrative Agent.
- 12.06 <u>Indemnification</u>. To the extent the Agent or any affiliate thereof or any of their respective Related Parties (collectively, the "Agent Indemnified Persons") is not reimbursed and indemnified by the Borrower, the Lenders will reimburse and indemnify such Agent Indemnified Persons in proportion to their respective percentage of the aggregate outstanding amount of the Loans and Commitments held by all Lenders and their Affiliates on such date (determined as if there were no Defaulting Lenders) for and against any and all liabilities, obligations, losses, damages, penalties, claims, actions, judgments, costs, expenses or disbursements of whatsoever kind or nature which may be imposed on, asserted against or incurred by any Agent Indemnified Person in performing its duties hereunder or under any other Credit Document or in any way relating to or arising out of this Agreement or any other Credit Document; provided that no Lender shall be liable for any portion of such liabilities, obligations, losses, damages, penalties, claims, actions, judgments, suits, costs, expenses or disbursements resulting from such Agent Indemnified Person's gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable decision).

## 12.07 Resignation by the Administrative Agent.

(a) The Administrative Agent may resign from the performance of all its respective functions and duties hereunder and/or under the other Credit Documents at any time by giving 15 Business Days' prior written notice to the Lenders and, unless a Default or an Event of Default under Section 11.05 then exists, the Borrower. Such resignation shall take effect upon

the appointment of a successor Administrative Agent pursuant to clauses (b) and (c) below or as otherwise provided below.

- (b) Upon any such notice of resignation by the Administrative Agent, the Required Lenders shall appoint a successor Administrative Agent hereunder or thereunder who shall be a commercial bank or trust company reasonably acceptable to the Borrower, which acceptance shall not be unreasonably withheld, delayed or conditioned (provided that the Borrower's approval shall not be required if an Event of Default then exists).
- (c) If a successor Administrative Agent shall not have been so appointed within such 15 Business Day period, the Administrative Agent, with the consent of the Borrower (which consent shall not be unreasonably withheld, delayed or conditioned, provided that the Borrower's consent shall not be required if an Event of Default then exists), shall then appoint a successor Administrative Agent who shall serve as Administrative Agent hereunder or thereunder until such time, if any, as the Required Lenders appoint a successor Administrative Agent as provided above.
- (d) If no successor Administrative Agent has been appointed pursuant to clause (b) or (c) above by the 20th Business Day after the date such notice of resignation was given by the Administrative Agent, the Administrative Agent's resignation shall become effective and the Required Lenders shall thereafter perform all the duties of the Administrative Agent hereunder and/or under any other Credit Document until such time, if any, as the Required Lenders appoint a successor Administrative Agent as provided above.
- (e) Upon a resignation of the Administrative Agent pursuant to this Section 12.07, the Administrative Agent shall remain indemnified to the extent provided in this Agreement and the other Credit Documents and the provisions of Section 12.06 and Section 14.01 (and the analogous provisions of the other Credit Documents) shall continue in effect for the benefit of the Administrative Agent for all of its actions and inactions while serving as the Administrative Agent. Upon its acceptance of any appointment as a successor Administrative Agent hereunder, such successor Administrative Agent shall succeed to and become vested with all the rights, powers, privileges and duties of the retiring Administrative Agent, and the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Credit Documents.

#### 12.08 Collateral Matters.

(a) Each Lender authorizes and directs the Collateral Agent to enter into the Security Documents for the benefit of the Lenders and the other Secured Creditors. In addition, each Lender, for the benefit of all parties to this Agreement, authorizes and directs the Collateral Agent to enter into any amendments to the Security Documents that may be deemed by the Required Lenders to be necessary in connection therewith for the benefit of the Lenders and the Secured Creditors. Each Lender hereby agrees that, except as otherwise set forth herein, any action taken by the Required Lenders in accordance with the provisions of this Agreement or the Security Documents, and the exercise by the Required Lenders of the powers set forth herein or therein, together with such other powers as are reasonably incidental thereto, shall be authorized and binding upon all of the Lenders. The Collateral Agent is hereby authorized on behalf of all

of the Lenders, without the necessity of any notice to or further consent from any Lender, from time to time prior to an Event of Default, to take any action with respect to any Collateral or Security Documents which may be necessary to perfect and maintain perfected the security interest in and liens upon the Collateral granted pursuant to the Security Documents.

- (b) The Lenders hereby authorize the Collateral Agent, at its option and in its discretion (at the direction of the Required Lenders), to release any Lien granted to or held by the Collateral Agent upon any Collateral (i) upon termination of the Commitments and payment and satisfaction in full of all of the Obligations (other than indemnification obligations for which no claim has been asserted) at any time arising under or in respect of this Agreement or the Credit Documents or the transactions contemplated hereby or thereby, (ii) constituting property being sold or otherwise disposed of (to Persons other than the Borrower and its Subsidiaries) upon the sale or other disposition thereof in compliance with Section 10.02, (iii) if approved, authorized or ratified in writing by the Required Lenders (or all of the Lenders hereunder, to the extent required by Section 14.11) or (iv) as otherwise may be expressly provided in the relevant Security Documents or the Sale Order. Upon request by the Administrative Agent at any time, the Lenders will confirm in writing the Collateral Agent's authority to release particular types or items of Collateral pursuant to this Section 12.08; provided that any Lender that does not respond to such request within fifteen (15) days of it being made by the Collateral Agent shall have deemed to have confirmed the Collateral Agent's authority to release the Collateral.
- (c) Anything contained in any of the Credit Documents to the contrary notwithstanding, the Borrower, the Agents and each Secured Creditor hereby agree that in the event of a foreclosure by the Collateral Agent or on any of the Collateral pursuant to a public or private sale or other disposition, the Collateral Agent (at the direction of the Required Lenders) or any Lender may be the purchaser or licensor of any or all of such Collateral at any such sale or other disposition, and the Collateral Agent, as agent for, and representative of, the Secured Creditors (but not any Lender or Lenders in its or their respective individual capacities unless Required Lenders shall otherwise agree in writing) shall be entitled (at the direction of the Required Lender), for the purpose of bidding and making settlement or payment of the purchase price for all or any portion of the Collateral sold at any such public sale, to use and apply any of the Obligations as a credit on account of the purchase price for any collateral payable by the Collateral Agent at such sale or other disposition.
- 12.09 <u>Delivery of Information</u>. The Administrative Agent shall not be required to deliver to any Lender originals or copies of any documents, instruments, notices, communications or other information received by the Administrative Agent from any Credit Party, any Subsidiary, the Required Lenders, any Lender or any other Person under or in connection with this Agreement or any other Credit Document except (i) as specifically provided in this Agreement or any other Credit Document and (ii) as specifically and reasonably requested from time to time in writing by any Lender with respect to a specific document, instrument, notice or other written communication received by and in the possession of the Administrative Agent at the time of receipt of such request and then only in accordance with such specific request.

## SECTION 13 Guaranty.

## 13.01 Guaranty.

- (a) Each of the Guarantors hereby, jointly and severally, unconditionally and irrevocably, guarantees to the Administrative Agent, for the ratable benefit of the Secured Creditors, and to the Secured Creditors the prompt and complete payment and performance when due and payable (whether at the stated maturity, by acceleration or otherwise) of all Obligations of the Borrowers and each other Credit Party.
- (b) Each Guarantor and each Secured Creditor (by its acceptance of the benefits of this Agreement) hereby confirms that it is its intention that the guaranty made by the Guarantors not constitute a fraudulent transfer or conveyance for purposes of the Bankruptcy Code, the Uniform Fraudulent Conveyance Act or any similar Federal or state law. To effectuate the foregoing intention, each Guarantor and each Secured Creditor (by its acceptance of the benefits of this Agreement) hereby irrevocably agrees that the Obligations guaranteed by such Guarantor shall be limited to such amount as will, after giving effect to such maximum amount and all other (contingent or otherwise) liabilities of such Guarantor that are relevant under such laws, not constitute a fraudulent transfer or conveyance for purposes of such laws.
- (c) Each Guarantor agrees that the Obligations guaranteed by it hereunder may at any time and from time to time exceed the amount of the liability of such Guarantor hereunder without impairing the guaranty contained in this <u>Section 13</u> or affecting the rights and remedies of the Administrative Agent or any other Secured Creditor hereunder.
- (d) No payment made by the Borrower, any of the Guarantors or any other Person or received or collected by the Administrative Agent or any other Secured Creditor from the Borrower, any of the Guarantors or any other Person by virtue of any action or proceeding or any set-off or appropriation or application at any time or from time to time in reduction of or payment of any of the Obligations shall be deemed to modify, reduce, release or otherwise affect the liability of any Guarantor hereunder which shall, notwithstanding any such payment (other than any payment made by such Guarantor in respect of the Obligations or any payment received or collected from such Guarantor in respect of any of the Obligations), remain liable for the Obligations guaranteed by it hereunder up to the maximum liability of such Guarantor hereunder until the earlier to occur of (i) the Maturity Date or (ii) the release of such Guarantor from this Agreement in accordance with the express provisions of Section 14.11 hereof.
- 13.02 Amendments, etc. with respect to the Obligations. To the maximum extent permitted by law, each Guarantor shall remain obligated hereunder notwithstanding that, without any reservation of rights against any Guarantor and without notice to or further assent by any Guarantor, any demand for payment of any of the Obligations made by the Administrative Agent or any other Secured Creditor may be rescinded by the Administrative Agent or such other Secured Creditor and any of the Obligations continued, and the Obligations, or the liability of any other Person upon or for any part thereof, or any collateral security or guaranty therefor or right of offset with respect thereto, may, from time to time, in whole or in part, be renewed, extended, amended, modified, accelerated, compromised, subordinated, waived, surrendered or released by the Administrative Agent or any other Secured Creditor, and this Agreement and the

other Credit Documents and any other documents executed and delivered in connection therewith may be amended, waived, modified, supplemented or terminated, in whole or in part, in accordance with their respective terms, as the Administrative agent (at the direction of the Required Lenders) may deem advisable from time to time, and any collateral security, guaranty or right of offset at any time held by the Administrative agent or any other Secured Creditor for the payment of any of the Obligations may be sold, exchanged, waived, surrendered or released. Neither the Administrative agent nor any other Secured Creditor shall have any obligation to protect, secure, perfect or insure any Lien at any time held by it as security for any of the Obligations or for the guaranty contained in this Section 13 or any property subject thereto, except to the extent required by applicable law.

13.03 Guaranty Absolute and Unconditional. Each Guarantor waives, to the maximum extent permitted by applicable law, any and all notice of the creation, renewal, extension or accrual of any of the Obligations and notice of or proof of reliance by the Administrative agent or any other Secured Creditor upon the guaranty contained in this Section 13 or acceptance of the guaranty contained in this Section 13; each of the Obligations, and any obligation contained therein, shall conclusively be deemed to have been created, contracted or incurred, or renewed, extended, amended or waived, in reliance upon the guaranty contained in this Section 13; and all dealings between the Borrowers and any of the other Credit Parties, on the one hand, and the Administrative agent and the other Secured Creditors, on the other hand, likewise shall be conclusively presumed to have been had or consummated in reliance upon the guaranty contained in this Section 13. Except as expressly required under the Final DIP Order, each Guarantor waives, to the maximum extent permitted by applicable law, diligence, presentment, protest, demand for payment and notice of default or nonpayment to or upon any of the Borrower or any of the other Credit Parties with respect to any of the Obligations. Each Guarantor understands and agrees, to the extent permitted by law, that the guaranty contained in this Section 13 shall be construed as a continuing, absolute and unconditional guaranty of payment and not of collection. Each Guarantor hereby waives, to the maximum extent permitted by applicable law, any and all defenses that it may have arising out of or in connection with any and all of the following: (a) the validity or enforceability of this Agreement or any other Credit Document, any of the Obligations or any other collateral security therefor or guaranty or right of offset with respect thereto at any time or from time to time held by the Administrative agent or any other Secured Creditor, (b) any defense, set-off or counterclaim (other than a defense of payment or performance) that may at any time be available to or be asserted by any Borrower against the Administrative agent or any other Secured Creditor, (c) any change in the time, place, manner or place of payment or any amendment, waiver or increase in any of the Obligations in accordance with the terms of the documentation evidencing the same, (d) any exchange, taking, or release of Collateral, (e) any change in the structure or existence of any of the Borrower or any of the other Credit Parties (except as provided in, or permitted by, this Agreement), (f) any application of Collateral to any of the Obligations, (g) any law, regulation or order of any jurisdiction, or any other event, affecting any term of any Obligation or the rights of the Administrative agent or any other Secured Creditor with respect thereto, including, without limitation: (i) the application of any such law, regulation, decree or order, including any prior approval, which would prevent the exchange of any currency (other than Dollars) for Dollars or the remittance of funds outside of such jurisdiction or the unavailability of Dollars in any legal exchange market in such jurisdiction in accordance with normal commercial practice, (ii) a declaration of banking moratorium or any suspension of payments by banks in such jurisdiction

or the imposition by such jurisdiction or any Governmental Authority thereof of any moratorium on, the required rescheduling or restructuring of, or required approval of payments on, any indebtedness in such jurisdiction, (iii) any expropriation, confiscation, nationalization or requisition by such country or any Governmental Authority that directly or indirectly deprives the Borrower or any other Credit Party of any assets or their use, or of the ability to operate its business or a material part thereof, or (iv) any war (whether or not declared), insurrection, revolution, hostile act, civil strife or similar events occurring in such jurisdiction which has the same effect as the events described in clause (i), (ii) or (iii) above (in each of the cases contemplated in clauses (i) through (iv) above, to the extent occurring or existing on or at any time after the date of this Agreement), or (h) any other circumstance whatsoever (other than payment in full in cash of the Obligations (other than inchoate indemnity obligations) guaranteed by it hereunder) (with or without notice to or knowledge of the Borrower or any other Credit Party) that constitutes, or might be construed to constitute, an equitable or legal discharge of the Borrower or any other Credit Party for its Obligations, or of such Guarantor under the guaranty contained in this Section 13, in bankruptcy or in any other instance. When making any demand hereunder or otherwise pursuing its rights and remedies hereunder against any Guarantor, the Administrative Agent or any other Secured Creditor may, but shall be under no obligation to, make a similar demand on or otherwise pursue such rights and remedies as it may have against the Borrower, any other Guarantor or any other Person or against any collateral security or guaranty for the Obligations guaranteed by such Guarantor hereunder or any right of offset with respect thereto, and any failure by the Administrative Agent or any other Secured Creditor to make any such demand, to pursue such other rights or remedies or to collect any payments from the Borrower, any other Guarantor or any other Person or to realize upon any such collateral security or guaranty or to exercise any such right of offset, or any release of the Borrower, any other Guarantor or any other Person or any such collateral security, guaranty or right of offset, shall not relieve any Guarantor of any obligation or liability hereunder, and shall not impair or affect the rights and remedies, whether express, implied or available as a matter of law, of the Administrative Agent or any other Secured Creditor against any Guarantor. For the purposes hereof "demand" shall include the commencement and continuance of any legal proceedings.

- 13.04 <u>Reinstatement</u>. The Guaranty of any Guarantor contained in this <u>Section 13</u> shall continue to be effective, or be reinstated, as the case may be, if at any time payment, or any part thereof, of any of the Obligations guaranteed by such Guarantor hereunder is rescinded or must otherwise be restored or returned by the Administrative Agent or any other Secured Creditor upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of the Borrower or any other Credit Party, or upon or as a result of the appointment of a receiver, intervenor or conservator of, or trustee or similar officer for, the Borrower or any other Credit Party or any substantial part of its property, or otherwise, all as though such payments had not been made.
- 13.05 <u>Payments</u>. Each Guarantor hereby agrees that payments hereunder will be paid to the Administrative Agent, for the benefit of the Secured Creditors, without set-off, counterclaim or other defense and on the same basis as payments are made by the Borrowers under <u>Sections</u> 5.03 and 5.04 of this Agreement.
- 13.06 <u>Information</u>. Each Guarantor assumes all responsibility for being and keeping itself informed of the Borrower's and each other Credit Party's and their Subsidiaries' financial condition and assets and of all other circumstances bearing upon the risk of nonpayment of the

Obligations and the nature, scope and extent of the risks that such Guarantor assumes and incurs hereunder and agrees that none of the Administrative Agent or the other Secured Creditors will have any duty to advise such Guarantor of information known to it or any of them regarding such circumstances or risks.

- 13.07 <u>Indemnity and Subrogation</u>. In addition to all rights of indemnity and subrogation as the Guarantors may have under applicable law (subject to <u>Section 13.09</u>), the Borrower agrees that (a) in the event a payment of any Obligation shall be made by any Guarantor under this Agreement, the Borrower shall indemnify such Guarantor for the full amount of such payment and such Guarantor shall be subrogated to the rights of the Person to whom such payment shall have been made to the extent of such payment and (b) in the event any assets of any Guarantor shall be sold pursuant to this Agreement or any other Credit Document to satisfy in whole or in part any Obligation owed to any Secured Creditor, the Borrower shall indemnify such Guarantor in an amount equal to the fair value of the assets so sold.
- 13.08 <u>Contribution and Subrogation</u>. Each Guarantor agrees (subject to <u>Section 13.09</u>) that, to the extent that a Guarantor shall have paid more than its proportionate share (based, to the maximum extent permitted by law, on the respective Adjusted Net Worth of the Guarantors on the date the respective payment is made) of any payment made hereunder, such Guarantor shall be entitled to seek and receive contribution from and against any other Guarantor hereunder that has not paid its proportionate share of such payment. Each Guarantor's right of contribution shall be subject to the terms and conditions of Section 13.09. Notwithstanding anything to the contrary contained above, any Guarantor that is released from this Agreement (and its guarantees contained herein) in accordance with the express provisions of Section 14.11 shall thereafter have no contribution obligations, or rights, pursuant to this Section 13.08, and at the time of any such release, the contribution rights and obligations of the remaining Guarantors shall be recalculated on the respective date of release (as otherwise provided herein) based on the payments made hereunder by the remaining Guarantors. The provisions of this Section 13.08 shall in no respect limit the obligations and liabilities of any Guarantor to the Administrative Agent and the other Secured Creditors, and each Guarantor shall remain liable to the Administrative Agent and the other Secured Creditors for the full amount guaranteed by such Guarantor hereunder.
- 13.09 <u>Subordination</u>. Notwithstanding any provision in this Agreement to the contrary, all rights of the Guarantors under <u>Sections 13.07</u> and <u>13.08</u> and all other rights of indemnity, contribution or subrogation under applicable law or otherwise shall be fully subordinated to the indefeasible payment in full in cash of the Obligations, and no Credit Party shall be entitled to be subrogated to any of the rights of the Administrative Agent or any other Secured Creditor against the Borrower or any other Credit Party or any collateral security or guaranty or right of offset held by the Administrative Agent or any other Secured Creditor for the payment of any of the Obligations, nor shall any Credit Party seek or be entitled to seek any contribution or reimbursement from the Borrower or any other Credit Party in respect of payments made by such Credit Party hereunder (or paid with proceeds of collateral of such Credit Party hereunder), until all amounts owing to the Administrative Agent and the other Secured Creditors on account of the Obligations are paid in full in cash and the Total Commitment has been terminated. If any amount shall be paid to any Credit Party on account of such contribution or subrogation rights at any time when all of the Obligations shall not have been paid in full in cash or any of the

Commitments shall remain in effect, such amount shall be held by such Credit Party in trust for the Administrative Agent and the other Secured Creditors, segregated from other funds of such Credit Party, and shall, forthwith upon receipt by such Credit Party, be turned over to the Administrative Agent in the exact form received by such Credit Party (duly indorsed by such Credit Party to the Administrative Agent, if required), to be held as collateral security for all of the Obligations (whether matured or unmatured) of, or guaranteed by, such Credit Party and/or then or at any time thereafter may be applied against any Obligations, whether matured or unmatured, in such order as the Administrative Agent may determine.

## SECTION 14 Miscellaneous.

#### 14.01 Payment of Expenses; Indemnifications etc.

- The Borrower hereby agrees, whether or not the transactions herein contemplated are consummated, to pay (i) all reasonable out-of-pocket costs and expenses of the Administrative Agent, the Collateral Agent and the Commitment Parties (including, without limitation, the reasonable fees and disbursements of outside counsel, special counsel, financial advisor and other advisors or consultants retained by each of the Administrative Agent, the Collateral Agent and the Commitment Parties) (but excluding the fees and expenses of any outside counsel or financial advisor retained by any individual Commitment Party) incurred on, prior to or after the Effective Date, in connection with (x) the Chapter 11 Cases or any other insolvency or bankruptcy proceedings of the Debtors, the preparation, execution, delivery and administration of this Agreement and the other Credit Documents and the documents and instruments referred to herein and therein and (y) the enforcement of this Agreement and the other Credit Documents and the documents and instruments referred to herein and therein or in connection with any refinancing or restructuring of the credit arrangements or any "work-out", and (ii) any fees of the Administrative Agent charged in connection with any "seasoning" of the Facility. Notwithstanding the foregoing, in no event shall (i) the Administrative Agent and (ii) the Lenders, in each case, be entitled to the reimbursement of costs and expenses of more than one primary counsel, one local Alabama counsel and other special counsel and advisors, as needed.
- (b) The Borrower shall pay and hold the Administrative Agent, the Collateral Agent, the Commitment Parties and each of the Lenders harmless from and against any and all present and future stamp, excise and other similar documentary taxes with respect to the foregoing matters and save the Administrative Agent, the Collateral Agent and each of the Lenders harmless from and against any and all liabilities with respect to or resulting from any delay or omission (other than to the extent attributable to the Administrative Agent, the Collateral Agent or such Lender, as applicable, as a result of the gross negligence or willful misconduct of such Person (as determined by a court of competent jurisdiction in a final and non-appealable decision)) to pay such taxes.
- (c) The Borrower shall indemnify the Administrative Agent, the Collateral Agent, each Commitment Party, each Lender, each Prepetition First Lien Secured Party and each of their respective Related Parties (each, an "Indemnified Person") from, and hold each of them harmless against, any and all liabilities, obligations (including removal or remedial actions), losses, damages (including, without limitation, consequential damages), penalties, claims,

actions, judgments, suits, costs, expenses and disbursements (including documented fees, disbursements and other charges of one primary counsel and one local counsel for each relevant jurisdiction to such Indemnified Persons (unless there is an actual or perceived conflict of interest or the availability of different claims or defenses in which case each such Person may retain its own counsel)) incurred by, imposed on or assessed against any of them as a result of, or arising out of, or in any way related to, or by reason of, (a) any investigation, litigation or other proceeding (whether or not such Indemnified Person is a party thereto and whether or not such investigation, litigation or other proceeding is brought by or on behalf of any Credit Party) related to the entering into and/or performance of this Agreement, any other Credit Document, the Final DIP Order or the proceeds of any Loans hereunder or the consummation of the Transaction or any other transactions contemplated herein or in any other Credit Document or the exercise of any of their rights or remedies provided herein or in the other Credit Documents, or (b) the actual or alleged presence of Hazardous Materials in the air, surface water or groundwater or on the surface or subsurface of any Real Property at any time owned, leased or operated by the Borrower or any of its Subsidiaries, the generation, storage, transportation, handling or disposal of Hazardous Materials by the Borrower or any of its Subsidiaries at any location, whether or not owned, leased or operated by the Borrower or any of its Subsidiaries, the non-compliance by the Borrower or any of its Subsidiaries with any Environmental Law (including applicable permits thereunder), or any Environmental Claim asserted against the Borrower, any of its Subsidiaries or any Real Property at any time owned, leased or operated by the Borrower or any of its Subsidiaries, provided that no Credit Party shall have any obligation hereunder to any Indemnified Person with respect to indemnified liabilities to the extent it has been determined by a final non-appealable judgment of a court of competent jurisdiction to have resulted solely from the gross negligence or willful misconduct of, or material breach of the obligations under this Agreement by, such Indemnified Person. To the extent that the undertaking to indemnify, pay or hold harmless the Administrative Agent, the Collateral Agent, any Lender or any other Indemnified Person set forth in the preceding sentence may be unenforceable because it is violative of any law or public policy, the Borrower shall make the maximum contribution to the payment and satisfaction of each of the indemnified liabilities which is permissible under applicable law.

- (d) To the fullest extent permitted by applicable law, the Borrower shall not assert, and hereby waives, any claim against any Indemnified Person, on any theory of liability, for special, indirect, consequential, punitive or incidental damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Loan or the use of the proceeds thereof.
- (e) No Indemnified Person shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Credit Documents or the transactions contemplated hereby or thereby, except to the extent the liability of such Indemnified Person results solely from such Indemnified Person's gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable decision).

- (f) All amounts due under this <u>Section 14.01</u> shall be payable to the Administrative Agent for its own benefit or for the benefit of the Lenders or the other Indemnified Persons, as applicable, not later than five (5) Business Days after demand therefor.
- 14.02 Right of Setoff. In addition to any rights now or hereafter granted under applicable law or otherwise, and not by way of limitation of any such rights, upon the occurrence and during the continuance of an Event of Default, but subject to any notice provisions in the Final DIP Order, each Agent (at the direction of the Required Lenders) and each Lender is hereby authorized at any time or from time to time, without presentment, demand, protest or other notice of any kind to any Credit Party or to any other Person, but subject to any notice provisions in the Final DIP Order, any such notice being hereby expressly waived, to set off and to appropriate and apply any and all deposits (general or special, time or demand, provision or final) and any other Indebtedness at any time held or owing by such Agent or such Lender (including, without limitation, by branches and agencies of such Agent or such Lender wherever located) to or for the credit or the account of the Borrower or any of its Subsidiaries against and on account of the Obligations and liabilities of the Credit Parties to such Agent or such Lender under this Agreement or under any of the other Credit Documents, including, without limitation, all interests in Obligations purchased by such Lender pursuant to Section 14.04(b), and all other claims of any nature or description arising out of or connected with this Agreement or any other Credit Document, irrespective of whether or not such Agent or such Lender shall have made any demand hereunder and although said Obligations, liabilities or claims, or any of them, shall be contingent or unmatured. The rights of each Lender and their respective Affiliates under this Section 14.02 are in addition to other rights and remedies (including other rights of set-off) that such Lender or their respective Affiliates may have. Each Lender agrees to notify the Borrower and the Administrative Agent promptly after any such set-off and application, provided that the failure to give such notice shall not affect the validity of such set-off and application.

#### 14.03 Notices.

Except as otherwise expressly provided herein, all notices and other (a) communications provided for hereunder shall be in writing (including telecopier or other electronic cable communication) and mailed, telecopied or electronic transmission or delivered: if to any Credit Party, at the address specified opposite its signature below or in the other relevant Credit Documents; if to any Lender, at its address specified on Schedule 1.01(b); if to any Agent, at the Notice Office with a copy to Weil, Gotshal & Manges LLP, 767 Fifth Avenue, New York, NY 10153 (Attn: [•]; Fax: +1 (212) 310-8007; Email: [•]); and if to the Commitment Parties, to Akin Gump Strauss Hauer & Feld LLP, One Bryant Park, New York, NY 10036 (Attn: Ira Dizengoff and Kristine Manoukian) and 1333 New Hampshire Avenue, N.W., Washington, DC 20036 (Attn: James Savin) or, as to any Credit Party or any Agent, at such other address as shall be designated by such party in a written notice to the other parties hereto and, as to each Lender, at such other address as shall be designated by such Lender in a written notice to the Borrower and the Administrative Agent. All such notices and communications shall, when mailed, telecopied, e-mailed or sent by overnight courier, be effective when deposited in the mails, delivered to the overnight courier or sent by telecopier or electronic transmission, except that notices and communications to any Agent and the Borrower shall not be effective until received by any Agent or the Borrower, as the case may be.

(b) Notices and other communications to the Lenders hereunder may be delivered or furnished by electronic communications (including by posting such notices or communications on the Platform) pursuant to procedures approved by the Administrative Agent; provided that the foregoing shall not apply to notices pursuant to Section 2 unless otherwise agreed by the Administrative Agent, the Collateral Agent and the applicable Lender. Each of the Administrative Agent and the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it; provided that approval of such procedures may be limited to particular notices or communications.

# 14.04 Benefit of Agreement; Assignments; Participations.

(a) Successors and Assigns Generally. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the respective successors and assigns of the parties hereto; provided, however, that the Borrower may not assign or otherwise transfer any of its rights, obligations or interest hereunder without the prior written consent of the Administrative Agent (acting at the direction of the Required Lenders) and the Lenders and, provided further, that, although any Lender may, without consent or notice, grant participations to Eligible Transferees in its rights hereunder, such Lender shall remain a "Lender" for all purposes hereunder (and may not transfer or assign all or any portion of its Commitments hereunder except as provided in Sections 2.13 and 13.04) and the participant shall not constitute a "Lender" hereunder and, provided further, that no Lender shall transfer or grant any participation under which the participant shall have rights to approve any amendment to or waiver of this Agreement or any other Credit Document except to the extent such amendment or waiver would (i) extend the final scheduled maturity of any Loan in which such participant is participating, or reduce the rate or extend the time of payment of interest or Fees thereon (except in connection with a waiver of applicability of any post-default increase in interest rates) or reduce the principal amount thereof (it being understood that any amendment or modification to the financial definitions in this Agreement or to Section 14.07(a) shall not constitute a reduction in the rate of interest or Fees payable hereunder), or increase the amount of the participant's participation over the amount thereof then in effect (it being understood that a waiver of any Default or Event of Default or a mandatory prepayment of the Loans shall not constitute a change in the terms of such participation, and that an increase in any Commitment (or the available portion thereof) or Loan shall be permitted without the consent of any participant if the participant's participation is not increased as a result thereof), (ii) consent to the assignment or transfer by any Borrower of any of their respective rights and obligations under this Agreement or (iii) release all or substantially all of (x) the Collateral under all the Security Documents (except as expressly provided in the Credit Documents or the Sale Motion) or (y) the Guaranty under this Agreement supporting the Loans in which such participant is participating. In the case of any such participation, the participant shall not have any rights under this Agreement or any of the other Credit Documents (the participant's rights against such Lender in respect of such participation to be those set forth in the agreement executed by such Lender in favor of the participant relating thereto) and all amounts payable by the Borrower hereunder shall be determined as if such Lender had not sold such participation. To the extent that a participation would at the time of participation directly cause increased costs under Section 5.05 from those being charged by the respective Lender prior to such participation, then the Borrower shall not be obliged to pay such increased costs.

- Notwithstanding the foregoing, any Lender (or any Lender together with one or more other Lenders) may (x) assign all or a portion of its Commitments and related outstanding Obligations hereunder to any existing Lender or Affiliate of a Lender or Approved Fund, provided, that no such assignment may be made to any such Person that is, or would at such time constitute, a Defaulting Lender or any of its Subsidiaries or (y) assign all, or if less than all, a portion equal to at least \$100,000 in the aggregate for the assigning Lender or assigning Lenders, of such Commitments and related outstanding Obligations hereunder to one or more Eligible Transferees, each of which assignees shall become a party to this Agreement as a Lender by execution of an Assignment and Assumption Agreement (it being understood that assignments may be done on a non-pro-rata basis between Loans and unfunded Commitments), provided that (i) at such time, Schedule 1.01(a) shall be deemed modified to reflect the Commitments and/or outstanding Loans, as the case may be, of such new Lender and of the existing Lenders, (ii) upon the surrender of the relevant Notes (if any) by the assigning Lender, new Notes will be issued, at the Borrower's sole expense, to such new Lender and to the assigning Lender upon the request of such new Lender or assigning Lender, such new Notes to be in conformity with the requirements of Section 2.05 (with appropriate modifications) to the extent needed to reflect the revised Commitments and/or outstanding Loans, as the case may be, (iii) the consent of the Administrative Agent shall be required in connection with any such assignment pursuant to clause (y) above (such consent not to be unreasonably withheld, delayed or conditioned); and (iv) the Administrative Agent shall receive at the time of each such assignment, from the assigning or assignee Lender, the payment of a non-refundable assignment fee of \$3,500 (provided that only one such fee shall be payable in the case of one or more concurrent assignments by or to investment funds managed or advised by the same investment advisor or an affiliated investment advisor). To the extent of any assignment pursuant to this Section 14.04(b), the assigning Lender shall be relieved of its obligations hereunder with respect to its assigned Commitments and outstanding Loans. At the time of each assignment pursuant to this Section 14.04(b) to a Person which is not already a Lender hereunder, the respective assignee Lender shall, to the extent legally entitled to do so, provide to the Borrower the appropriate Internal Revenue Service Forms (and, if applicable, a Tax Compliance Certificate) and such other documentation described in Section 5.05(b). To the extent that an assignment of all or any portion of a Lender's Commitments and related outstanding Obligations pursuant to this Section 14.04(b) would, at the time of such assignment, result in increased costs under Section 5.05 from those being charged by the respective assigning Lender prior to such assignment, then the Borrower shall not be obligated to pay such increased costs (although the Borrower, in accordance with and pursuant to the other provisions of this Agreement, shall be obligated to pay any other increased costs of the type described above resulting from changes after the date of the respective assignment). Notwithstanding anything to the contrary herein, no assignment shall be made by any Lender to the Borrower or any of the Borrower's Affiliates or Subsidiaries, or to a natural person.
- (c) <u>Register</u>. The Borrower hereby designates the Administrative Agent to serve as its agent, solely for purposes of this <u>Section 14.04</u>, to maintain a register (the "<u>Register</u>") on which it will record the Loans made by each of the Lenders and each repayment in respect of the principal amount of the Loans of each Lender. Failure to make any such recordation, or any error in such recordation, shall not affect the Borrower's obligations in respect of such Loans, except under Section 5.05(a). With respect to any Lender, the transfer of the rights to the principal of, and interest on, any Loan shall not be effective until such transfer is

recorded on the Register maintained by the Administrative Agent with respect to ownership of such Loans and prior to such recordation all amounts owing to the transferor with respect to such Loans shall remain owing to the transferor. The registration of assignment or transfer of all or part of any Loans shall be recorded by the Administrative Agent on the Register upon and only upon the acceptance by the Administrative Agent of a properly executed and delivered Assignment and Assumption Agreement pursuant to Section 14.04(b). The Administrative Agent shall allow the Borrower or any Lender to inspect the Register at any time upon reasonable prior notice to the extent such inspection is reasonably determined by the Borrower or any Lender to be necessary to establish the relevant Loans or other obligations are in registered form under Section 5f.103-1(c) of the U.S. Treasury Regulations. Without duplication with Section 14.01, the Borrower agrees to indemnify the Administrative Agent from and against any and all losses, claims, damages and liabilities of whatsoever nature which may be imposed on, asserted against or incurred by the Administrative Agent in performing its duties under this Section 14.04(c).

Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts (and related interest amounts) of each Participant's interest in the Loans or other obligations under this Agreement (the "Participant Register"). The entries in the Participant Register shall be conclusive, absent manifest error, and each Person whose name is recorded in the Participant Register shall be treated as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. No Lender shall have any obligation to disclose all or any portion of any Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any Loans or its other obligations under any Credit Document) to any Person except to the extent that such disclosure is necessary to establish that such Loan is in registered from under Section 5f.103-1(c) of the U.S. Treasury Regulations.

- (d) Nothing in this Agreement shall prevent or prohibit any Lender from pledging its Loans hereunder to a Federal Reserve Bank or any central bank in support of borrowings made by such Lender from such Federal Reserve Bank or central bank, any Lender which is a fund may pledge all or any portion of its Loans to its trustee or to a collateral agent providing credit or credit support to such Lender in support of its obligations to such trustee, such collateral agent or a holder of such obligations, as the case may be. No pledge pursuant to this clause (d) shall release the transferor Lender from any of its obligations hereunder.
- (e) Any Lender which assigns all of its Commitments and/or Loans hereunder in accordance with Section 14.04(b) shall cease to constitute a "Lender" hereunder, except with respect to indemnification provisions under this Agreement (including, without limitation, Sections 5.05, 12.06, 14.01 and 14.06), which shall survive as to such assigning Lender.
- 14.05 <u>No Waiver; Remedies Cumulative</u>. No failure or delay on the part of the Administrative Agent, the Collateral Agent or any Lender in exercising any right, power or privilege hereunder or under any other Credit Document and no course of dealing between the Borrower or any other Credit Party and the Administrative Agent, the Collateral Agent or any Lender shall operate as a waiver thereof; nor shall any single or partial exercise of any right, power or privilege hereunder or under any other Credit Document preclude any other or further

exercise thereof or the exercise of any other right, power or privilege hereunder or thereunder. The rights, powers and remedies herein or in any other Credit Document expressly provided are cumulative and not exclusive of any rights, powers or remedies which the Administrative Agent, the Collateral Agent or any Lender would otherwise have. No notice to or demand on any Credit Party in any case shall entitle any Credit Party to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of the Administrative Agent, the Collateral Agent or any Lender to any other or further action in any circumstances without notice or demand.

#### 14.06 Payments Pro Rata.

- (a) Except as otherwise provided in this Agreement, the Administrative Agent agrees that promptly after its receipt of each payment from or on behalf of the Borrower in respect of any Obligations hereunder, the Administrative Agent shall distribute such payment to the Lenders entitled thereto (other than any Lender that has consented in writing to waive its *pro rata* share of any such payment) *pro rata* based upon their respective shares, if any, of the Obligations with respect to which such payment was received.
- (b) Each of the Lenders agrees that, if it should receive any amount hereunder (whether by voluntary payment, by realization upon security, by the exercise of the right of setoff or banker's lien, by counterclaim or cross action, by the enforcement of any right under the Credit Documents, or otherwise), which is applicable to the payment of the principal of, or interest on, the Loans, of a sum which with respect to the related sum or sums received by other Lenders is in a greater proportion than the total of such Obligation then owed and due to such Lender bears to the total of such Obligation then owed and due to all of the Lenders immediately prior to such receipt, then such Lender receiving such excess payment shall purchase for cash without recourse or warranty from the other Lenders an interest in the Obligations of the respective Credit Party to such Lenders in such amount as shall result in a proportional participation by all the Lenders in such amount; provided that (i) if all or any portion of such excess amount is thereafter recovered from such Lenders, such purchase shall be rescinded and the purchase price restored to the extent of such recovery, but without interest and (ii) the provisions of this paragraph (b) shall not be construed to apply to any payment made by the Borrower pursuant to and in accordance with the express terms of this Agreement or any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans or participations pursuant to Section 14.04.
- (c) Notwithstanding anything to the contrary contained herein, the provisions of the preceding Sections 13.06(a) and (b) shall be subject to the express provisions of this Agreement which require, or permit, differing payments to be made to Non-Defaulting Lenders as opposed to Defaulting Lenders.

#### 14.07 <u>Calculations</u>; <u>Computations</u>.

(a) The financial statements to be furnished to the Lenders pursuant hereto shall be made and prepared in accordance with GAAP, consistently applied throughout the periods involved (except as set forth in the notes thereto or as otherwise disclosed in writing by the Borrower to the Lenders); <u>provided</u> that, (i) notwithstanding anything to the contrary

contained herein, all such financial statements shall be prepared, and all financial covenants contained herein or in any other Credit Document shall be calculated, in each case, without giving effect to any election under FASB ASC 825 (or any similar accounting principle) permitting a Person to value its financial liabilities at the fair value thereof and (ii) to the extent expressly provided herein, certain calculations shall be made on a pro forma basis.

(b) All computations of interest and other Fees hereunder shall be made on the basis of a year of 360 days for the actual number of days.

# 14.08 <u>GOVERNING LAW; SUBMISSION TO JURISDICTION; VENUE; WAIVER</u> OF JURY TRIAL.

- (a) THIS AGREEMENT AND THE OTHER CREDIT DOCUMENTS AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER AND THEREUNDER SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH AND BE GOVERNED BY THE LAWS OF THE STATE OF NEW YORK, EXCEPT AS GOVERNED BY THE BANKRUPTCY CODE.
- THE BORROWER IRREVOCABLY AND UNCONDITIONALLY (b) AGREES THAT IT WILL NOT COMMENCE ANY ACTION, LITIGATION OR PROCEEDING OF ANY KIND OR DESCRIPTION, WHETHER IN LAW OR EQUITY, WHETHER IN CONTRACT OR IN TORT OR OTHERWISE, AGAINST ANY AGENT, ANY LENDER OR ANY RELATED PARTY OF THE FOREGOING IN ANY WAY RELATING TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT TRANSACTIONS RELATING HERETO OR THERETO, IN ANY FORUM OTHER THAN THE BANKRUPTCY COURT. THE BORROWER HEREBY IRREVOCABLY ACCEPTS RESPECT OF ITS FOR ITSELF AND IN PROPERTY. **GENERALLY** UNCONDITIONALLY, THE EXCLUSIVE JURISDICTION OF THE BANKRUPTCY COURT. THE BORROWER HEREBY FURTHER IRREVOCABLY WAIVES ANY CLAIM THAT THE BANKRUPTCY COURT LACK PERSONAL JURISDICTION OVER THE BORROWER, AND AGREES NOT TO PLEAD OR CLAIM, IN ANY LEGAL ACTION PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT BROUGHT IN THE BANKRUPTCY COURT, THAT SUCH COURTS LACK PERSONAL JURISDICTION OVER THE BORROWER. NOTHING IN THIS AGREEMENT OR IN ANY OTHER CREDIT DOCUMENT SHALL AFFECT ANY RIGHT THAT ANY AGENT OR ANY LENDER MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT, INCLUDING WITH RESPECT TO THE COLLATERAL, AGAINST ANY CREDIT PARTY OR ITS PROPERTIES IN THE COURTS OF ANY JURISDICTION.
- (c) THE BORROWER FURTHER IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS OUT OF THE BANKRUPTCY COURT IN ANY SUCH ACTION OR PROCEEDING BY THE MAILING OF COPIES THEREOF IN THE MANNER PROVIDED FOR NOTICES IN <u>SECTION 14.03</u>. THE BORROWER HEREBY IRREVOCABLY WAIVES ANY OBJECTION TO SUCH SERVICE OF PROCESS AND FURTHER IRREVOCABLY WAIVES AND AGREES NOT TO PLEAD OR CLAIM IN ANY ACTION OR PROCEEDING COMMENCED HEREUNDER OR UNDER ANY OTHER

CREDIT DOCUMENT THAT SERVICE OF PROCESS WAS IN ANY WAY INVALID OR INEFFECTIVE. NOTHING HEREIN SHALL AFFECT THE RIGHT OF ANY AGENT OR ANY LENDER TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR TO COMMENCE LEGAL PROCEEDINGS OR OTHERWISE PROCEED AGAINST THE BORROWER IN ANY OTHER JURISDICTION.

- (d) THE BORROWER HEREBY IRREVOCABLY WAIVES ANY OBJECTION WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF THE BANKRUPTCY COURT OR PROCEEDINGS ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT BROUGHT IN THE BANKRUPTCY COURT REFERRED TO IN CLAUSE (A) ABOVE AND HEREBY FURTHER IRREVOCABLY WAIVES AND AGREES NOT TO PLEAD OR CLAIM IN SUCH COURT THAT ANY SUCH ACTION OR PROCEEDING BROUGHT IN SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.
- (e) EACH OF THE PARTIES TO THIS AGREEMENT HEREBY IRREVOCABLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE OTHER CREDIT DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND OTHER CREDIT DOCUMENTS BY, AMONG OTHER THINGS, THE WAIVERS AND CERTIFICATIONS IN THIS SECTION.
- 14.09 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts and by the different parties hereto on separate counterparts (including by facsimile or other electronic transmission), each of which when so executed and delivered shall be an original, but all of which shall together constitute one and the same instrument. A set of counterparts executed by all the parties hereto shall be lodged with the Borrower and the Administrative Agent.
- 14.10 <u>Headings Descriptive</u>. The headings of the several sections and subsections of this Agreement are inserted for convenience only and shall not in any way affect the meaning or construction of any provision of this Agreement.

## 14.11 Amendment or Waiver; etc.

(a) Neither this Agreement nor any other Credit Document nor any terms hereof or thereof may be changed, waived, discharged or terminated unless such change, waiver, discharge or termination is in writing signed by the respective Credit Parties party hereto or thereto and the Required Lenders (although additional parties may be added to (and annexes may be modified to reflect such additions), provided that no such change, waiver, discharge or

termination shall, without the consent of each Lender (other than, except with respect to following clause (i), a Defaulting Lender):

- (i) (x) extend the final scheduled maturity of any Loan, or (y) postpone the scheduled maturity of any Loan or extend any time for any payment (excluding mandatory prepayments) of any interest, Fees or other amounts due to the Lenders (or any of them) under this Agreement or under any other Credit Document without the written consent of each Lender entitled to such payment;
- (ii) waive, reduce or forgive the principal of, or the rate of interest specified herein on, any Loan or any Fees or other amounts payable hereunder or under any other Credit Document without the written consent of each Lender entitled to such amount, provided, however, that only the consent of the Required Lenders shall be necessary to amend the definition of "Default Rate" or to waive any obligation of the Borrower to pay interest at the Default Rate;
- (iii) release all or substantially all of (x) the Collateral (except as expressly provided or permitted in the Credit Documents or the Sale Motion) or (y) the guarantees under this Agreement;
- (iv) amend, modify or waive any provision of this <u>Section 14.11</u> or the definition of "<u>Required Lenders</u>" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder without the written consent of each Lender adversely affected thereby;
- (v) consent to the assignment or transfer by the Borrower of any of its rights and obligations under this Agreement;
- (vi) extend or increase the Commitment of any Lender (or reinstate any Commitment terminated pursuant to <u>Section to 11</u>) without the written consent of such Lender;
- (vii) change <u>Section 11.13</u> or <u>Section 14.06</u> in a manner that would alter the pro rata sharing of payments required thereby without the written consent of each Lender or in any manner that materially and adversely affects the Lenders without the written consent of each Lender adversely affected thereby;
- (viii) subordinate any of the Obligations or any Lien created by this Agreement or any other Credit Document (except as permitted in this Agreement and the Credit Documents) without the consent of each Lender; or
- (ix) modify, waive, release or subordinate the superpriority claim status of the Obligations (except as permitted in this Agreement and the Credit Documents) without the consent of each Lender.
- (b) Notwithstanding anything to the contrary contained in this <u>Section 14.11</u>, (x) Security Documents (including any Additional Security Documents) and related documents

executed by Subsidiaries in connection with this Agreement may be in a form reasonably determined by the Agent and may be amended, supplemented and waived with the consent of the Agent and the Borrower without the need to obtain the consent of any other Person if such amendment, supplement or waiver is delivered in order (i) to comply with local Law or advice of local counsel, (ii) to cure ambiguities, omissions, mistakes or defects or (iii) to cause such Security Document or other document to be consistent with this Agreement and the other Credit Documents, (y) if following the Effective Date, the Administrative Agent and any Credit Party shall have jointly identified an ambiguity, inconsistency, obvious error or any error or omission of a technical or immaterial nature, in each case, in any provision of the Credit Documents (other than the Security Documents), then the Administrative Agent and the Credit Parties shall be permitted to amend such provision and such amendment shall become effective without any further action or consent of any other party to any Credit Documents if the same is not objected to in writing by the Required Lenders within five (5) Business Days following receipt of notice thereof and (z) no credit bidding with respect to the Obligations hereunder shall be permitted without the consent of each affected Lender.

## 14.12 Survival.

- (a) All indemnities set forth herein including, without limitation, in Sections 5.05, 12.06 and 14.01 shall survive the execution, delivery and termination of this Agreement and the making and repayment of the Obligations. The Indemnified Persons shall be intended third party beneficiaries of all indemnities set forth herein.
- (b) To the extent any payment by or on behalf of the Borrower is made to any Agent or any Lender, or any Agent or any Lender exercises its rights of set-off, and such payment or the proceeds of such set-off or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by such Agent or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, then to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such set-off had not occurred.
- (c) All representations and warranties made hereunder and in any other Credit Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by each Agent and each Lender, regardless of any investigation made by any Agent or any Lender on their behalf and notwithstanding that any Agent or any Lender may have had notice or knowledge of any Default at the time of any Loan, and shall continue to in full force and effect as long as any Loan or any other Obligation hereunder shall remain unpaid or unsatisfied.
- 14.13 <u>Domicile of Loans</u>. Each Lender may transfer and carry its Loans at, to or for the account of any office, Subsidiary or Affiliate of such Lender. Notwithstanding anything to the contrary contained herein, to the extent that a transfer of Loans pursuant to this <u>Section 14.13</u> would, at the time of such transfer, result in increased costs under <u>Section 5.05</u> from those being charged by the respective Lender prior to such transfer, then the Borrower shall not be obligated

to pay such increased costs (although the Borrower shall be obligated to pay any other increased costs of the type described above resulting from changes after the date of the respective transfer).

### 14.14 [Reserved].

## 14.15 <u>Confidentiality</u>.

- Subject to the provisions of clause (b) of this Section 14.15, each Lender (a) agrees that it will use its reasonable efforts not to disclose without the prior consent of the Borrower (other than to its employees, auditors, advisors or counsel or to another Lender if such Lender or such Lender's holding or parent company in its sole discretion determines that any such party should have access to such information, provided such Persons shall be subject to the provisions of this Section 14.15 to the same extent as such Lender) any information with respect to the Borrower or any of its Subsidiaries which is now or in the future furnished pursuant to this Agreement or any other Credit Document, provided that any Lender may disclose any such information (i) as has become generally available to the public other than by virtue of a breach of this Section 14.15(a) by the respective Lender, (ii) as may be required or appropriate in any report, statement or testimony submitted to any municipal, state or Federal regulatory body or self-regulatory body having or claiming to have jurisdiction over such Lender or to the Federal Reserve Board or the Federal Deposit Insurance Corporation or similar organizations (whether in the United States or elsewhere) or their successors or in connection with pledges or assignments in favor of any Federal Reserve Bank or central bank permitted under Section 14.04(d), (iii) as may be required or appropriate in respect to any summons or subpoena or in connection with any litigation, (iv) in order to comply with any law, order, regulation or ruling applicable to such Lender, (v) to the Administrative Agent or the Collateral Agent, (vi) to any direct or indirect contractual counterparty in any swap, hedge or similar agreement (or to any such contractual counterparty's professional advisor), so long as such contractual counterparty (or such professional advisor) agrees to be bound by the provisions of this Section 14.15 and (vii) to any prospective or actual transferee or participant in connection with any contemplated transfer or participation of any Loan or interest therein by such Lender, provided that such prospective transferee agrees to be bound by the confidentiality provisions contained in this Section 14.15.
- (b) The Borrower hereby acknowledges and agrees that each Lender may share with any of its Affiliates, and such Affiliates may share with such Lender, any information related to the Borrower or any of its Subsidiaries (including, without limitation, any non-public customer information regarding the creditworthiness of the Borrower and its Subsidiaries), provided such Persons shall be subject to the provisions of this <u>Section 14.15</u> to the same extent as such Lender.
- 14.16 Patriot Act. Each Lender subject to the USA PATRIOT Improvement and Reauthorization Act, Pub. L. 109-177 (signed into law March 9, 2009) (as amended from time to time, the "Act") and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies the Borrower that pursuant to the requirements of the Act, it is required to obtain, verify and record information that identifies each Credit Party and other information that will allow such Lender or the Administrative Agent, as applicable, to identify each Credit Party in accordance with the Act. The Borrower shall, promptly following a reasonable request by the Administrative Agent or any Lender, provide all documentation and other information that the

Administrative or such Lender reasonably requests in order to comply with its ongoing obligations under applicable "know your customer" and anti-money laundering rules and regulations, including the Act.

- 14.17 <u>Interest Rate Limitation</u>. Notwithstanding anything to the contrary contained in any Credit Document, the interest paid or agreed to be paid under the Credit Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable law (the "<u>Maximum Rate</u>"). If any Agent or any Lender shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the principal of the Loans or, if it exceeds such unpaid principal, refunded to the Borrower. In determining whether the interest contracted for, charged, or received by an Administrative Agent or a Lender exceeds the Maximum Rate, such Person may, to the extent permitted by applicable law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.
- 14.18 <u>Waiver of Immunities</u>. If any Credit Party has or hereafter may acquire any immunity (sovereign or otherwise) from any legal action, suit or proceeding, from jurisdiction of any court or from set-off or any legal process (whether service or notice, attachment prior to judgment attachment in aid of execution of judgment, execution of judgment or otherwise) with respect to itself or any of its property, such Credit Party hereby (to the fullest extent permitted by applicable law) irrevocably waives and agrees not to plead or claim such immunity in respect of its obligation under this Agreement and each other Credit Documents. Each Credit Party agrees that the foregoing waivers shall be effective to the fullest extent permitted under the Foreign Sovereign Immunities Act of 1976 of the United States of America, as amended from time to time, and are intended to be irrevocable and not subject to withdrawal for purposes of such Act.
- 14.19 No Advisory or No Fiduciary Duty. Each Agent, the Lenders and their Affiliates (collectively, solely for purposes of this Section 14.19, the "Lender Parties") may have economic interests that conflict with those of the Borrower, its stockholders and/or its Affiliates. The Borrower agrees that nothing in this Agreement or other Credit Documents or otherwise will be deemed to create an advisory, fiduciary or agency relationship or fiduciary or other implied duty between any Lender Party, on the one hand, and the Borrower, its stockholders or its Affiliates, The Borrower acknowledges and agrees, and acknowledges its Affiliates' understanding, that (i) the transactions contemplated by this Agreement and the other Credit Documents (including the exercise of rights and remedies hereunder and thereunder) are arm'slength commercial transactions between the Lender Parties, on the one hand, and the Borrower, on the other, and (ii) in connection therewith and with the process leading thereto, (x) no Lender Parties have assumed an advisory or fiduciary responsibility in favor of the Borrower, its stockholders or its Affiliates with respect to the transactions contemplated hereby (or the exercise of rights or remedies with respect thereto) or the process leading thereto (irrespective of whether any Lender Parties have advised, are currently advising or will advise the Borrower, its stockholders or its Affiliates on other matters) or any other obligation to the Borrower except the obligations expressly set forth in this Agreement and other Credit Documents and (y) each Lender Party is acting solely as principal and not as the agent or fiduciary of the Borrower, its management, stockholders, creditors or any other Person. The Borrower acknowledges and

agrees that it has consulted its own legal and financial advisors to the extent it deemed appropriate and that it is responsible for making its own independent judgment with respect to such transactions and the process leading thereto. The Borrower agrees that it will not claim that any Lender Party has rendered advisory services of any nature or respect, or owes a fiduciary or similar duty to the Borrower, in connection with such transaction or the process leading thereto.

- 14.20 <u>Required Lender Instructions</u>. Notwithstanding anything herein to the contrary, for purposes of any notice, instruction or direction required or requested from the Required Lenders in this Agreement (including, but not limited to, the letter of direction set forth in <u>Section 7.09</u>), any written notice, instruction or direction (including, without limitation, any notice, instruction or direction delivered by email) from Akin Gump Strauss Hauer & Feld LLP, as counsel to the Commitment Parties, shall be deemed a notice, instruction or direction delivered by the Required Lenders, as applicable.
- 14.21 <u>Final DIP Order</u>. In the event of any conflict between, or inconsistencies among, the Final DIP Order, on the one hand, and this Agreement or any other Credit Document, on the other hand, the Final DIP Order shall control to the extent of such conflict or inconsistency.

\* \* \*

execute and deliver this Agreement as of the date first above written.				
Address:				
	WALTER ENERGY, INC.			
	By: Name: Title:			

IN WITNESS WHEREOF, the parties hereto have caused their duly authorized officers to

	dministrative Agent
By:	
	Name:
	Title:

-	ME OF INSTITUTION] ender
By:	
3	Name:
	Title:

## Exhibit B

**Approved Budget** 

#### Walter Energy, Inc. and Domestic Subsidiaries

DIP BUDGET

(Amounts in USD)													
		1	2	3	4	5	6	7	8	9	10	11	Total
	Actual / Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	for
	Week Ending	1/23/16	1/30/16	2/6/16	2/13/16	2/20/16	2/27/16	3/5/16	3/12/16	3/19/16	3/26/16	4/2/16	11 Weeks
TOTAL NET CASH FLOW													
Receipts													
Sales/AR Receipts		\$ 1,214,142	\$ 16,695,315	\$ 7,062,054	\$ 2,631,837	\$ 17,814,480	\$ 7,019,668	\$ 9,638,096	\$ 10,522,590	\$ 480,106	\$ 8,489,162	\$ 418,033	\$ 81,985,484
Other		93,500	900,000	15,000	35,000	415,000	625,500	265,000	35,000	47,500	693,000	260,000	3,384,500
Total Cash Receipts	=	1,307,642	17,595,315	7,077,054	2,666,837	18,229,480	7,645,168	9,903,096	10,557,590	527,606	9,182,162	678,033	85,369,984
Disbursements													
Payroll, Benefits & Pension		(2,152,874)	(5,647,429)	(3,469,460)	(6,059,358)	(2,760,038)	(6,696,455)	(2,607,310)	(6,833,727)	(2,582,310)	(3,634,964)	(4,752,333)	(47,196,256)
Leases, Taxes, Utilities, Fuel, Insurance		(921,140)	(4,699,271)	(554,000)	(3,207,852)	(1,038,700)	(1,727,194)	(402,500)	(2,174,041)	(1,039,810)	(970,309)	(803,500)	(17,538,316)
Freight & Royalties		(4,122,716)	(2,829,123)	(2,712,823)	(2,664,026)	(4,802,658)	(2,731,221)	(1,262,000)	(1,368,772)	(2,925,579)	(1,563,662)	(1,242,000)	(28,224,579)
Other Expenditures		(3,776,431)	(5,691,951)	(6,924,951)	(5,824,951)	(5,490,912)	(4,734,954)	(3,246,191)	(3,479,191)	(2,909,191)	(3,314,179)	(3,432,178)	(48,825,081)
Total Disbursements <sup>1</sup>	=	(10,973,161)	(18,867,773)	(13,661,233)	(17,756,186)	(14,092,307)	(15,889,823)	(7,518,001)	(13,855,731)	(9,456,890)	(9,483,114)	(10,230,011)	(141,784,232)
TOTAL NET CASH FLOW <sup>2</sup>	=	\$ (9,665,519)	\$ (1,272,458)	\$ (6,584,179)	\$ (15,089,349)	\$ 4,137,173	\$ (8,244,655)	\$ 2,385,095	\$ (3,298,141)	\$ (8,929,284)	\$ (300,952)	\$ (9,551,978)	\$ (56,414,247)
Cash Collateral Budget Metrics													
Consideration Technic										•			
<u>Cumulative Totals</u>		(40.072.464)	(20.040.024)	(42 502 467)	(64.250.254)	(75.250.664)	(04.240.404)	(00.750.405)	(442.644.246)	(422.074.406)	(424 554 224)	(4.44.704.222)	
Total Disbursements		(10,973,161)	(29,840,934)	(43,502,167)	(61,258,354)	(75,350,661)	(91,240,484)	(98,758,485)	(112,614,216)	(122,071,106)	(131,554,221)	(141,784,232)	
Total Net Cash Flow		(9,665,519)	(10,937,977)	(17,522,156)	(32,611,505)	(28,474,332)	(36,718,987)	(34,333,892)	(37,632,033)	(46,561,318)	(46,862,269)	(56,414,247)	

#### Notes:

<sup>&</sup>lt;sup>1</sup> Although permitted pursuant to the terms of the Cash Collateral Order, the figures above exclude (i) adequate protection payments, (ii) fees and expenses of professionals retained outside the ordinary course of business, (iii) the Debtors' use of the Cash Collateral to collateral to collateral any new, replaced or renewed letters of credit, surety bonds or workers' compensation obligations in each case, that have been consented to by the Steering Committee in its sole discretion and (iv) key employee retention payments approved by both the Steering Committee and the Court that are approved to be paid pursuant to the Approved Budget and the Cash Collateral Order.

<sup>&</sup>lt;sup>2</sup> Budget assumes continuation of ordinary course transactions between the Debtors and Non-Debtor Affiliates Black Warrior Methane and Black Warrior Transmission.

# Exhibit C

**Capital Expenditure Budget** 

# Capital Expenditure Budget

	Jan	Feb	March
No. 4	-	-	_
No. 7	3,351,000	3,801,000	3,082,000
Maple Underground	40,000	40,000	40,000
WBWB	-	150,000	-
Walter Coke	700,000	700,000	700,000
BWM	150,667	150,667	150,667
Total	4,241,667	4,841,667	3,972,667

## **Notice Recipients**

District/Off: 1126–2 User: Itumlin Date Created: 1/28/2016

Case: 15–02741–TOM11 Form ID: pdf000 Total: 246

#### Recipients submitted to the BNC (Bankruptcy Noticing Center) without an address:

cr Delaware Trust Company, as Indenture Trustee

aty Lisa Beckerman

TOTAL: 2

```
Recipients of Notice of Electronic Filing:
                                pdarby@babc.com
            Patrick Darby
aty
                                zareba.adrian@pbgc.gov
            Adrian Zareba
            Adrienne K Walker
aty
                                     awalker@mintz.com
                              ecfpleadings@kccllc.com
            Albert Kass
aty
            Amber M. Whillock
                                     awhillock@starneslaw.com
aty
            Arthur Lee Tucker
                                    leetucker@leetucker-law.com
aty
            Benjamin Shaw Goldman
                                          bgoldman@handarendall.com
aty
aty
            Bill D Bensinger
                                  bdbensinger@csattorneys.com
                                  bwalding@waldinglaw.com
            Brian R Walding
aty
            C Taylor Crockett
                                   taylor@taylorcrockett.com
aty
                                     csteege@jenner.com
            Catherine L. Steege
aty
            Cathleen C Moore
                                    ccmoore@babc.com
aty
                                           melissa@mosespc.com
            Charles Howard Moses, III
aty
aty
            Clark R Hammond
                                    chammond@wallacejordan.com
                                        ebrazeal@joneswalker.com
            Clyde Ellis Brazeal, III
aty
aty
            D Christopher Carson
                                       ccarson@burr.com
aty
            Daniel Pasky
                               dpasky@mcglinchey.com
            Daniel D Sparks
aty
                                  ddsparks@csattorneys.com
            Daniel D Sparks
                                  ddsparks@csattorneys.com
aty
aty
            David B. Anderson
                                     dbanderson@andersonweidner.com
aty
            David Lewis Selby, II
                                       dselby@baileyglasser.com
                                  dsm@spain-gillon.com
            David S. Maxey
aty
aty
            Edward E. May
                                 bankruptcy@maylegalgroup.com
                                    ed.ragland@usdoj.gov
            Edward Q Ragland
aty
            Edwin Bryan Nichols
                                       bnichols@waldinglaw.com
aty
                               epruitt@bakerdonelson.com
            Eric L. Pruitt
aty
            Eric T Ray
                             eray@balch.com
aty
            Frank A. Anderson
                                    anderson.frank@pbgc.gov
aty
                                         fmg@spain-gillon.com
aty
            Frederick Mott Garfield
            George N. Davies
                                   gdavies@qcwdr.com
aty
            Ginger D Cockrell
                                    GINGERCOCKRELL@COMCAST.NET
aty
                                        gconnor@qcwdr.com
aty
            Glen Marshall Connor
aty
            Grady Milton McCarthy
                                         milton.mccarthy@asmc.alabama.gov
            Gregory Michael Taube
                                         greg.taube@nelsonmullins.com
aty
                               idizengoff@akingump.com
            Ira Dizengoff
aty
aty
            J Leland Murphree
                                    Lmurphree@maynardcooper.com
aty
            James Savin
                              jsavin@akingump.com
                                     jbailey@babc.com
            James Blake Bailey
aty
                                     JamesH@pm-j.com
            James G Henderson
aty
aty
                                 jwhite@bakerdonelson.com
            James H White
aty
            Jason Wayne Bobo
                                     jwb@cabaniss.com
            Jay R. Bender
                                jbender@babc.com
aty
            Jayna Partain Lamar
aty
                                     jlamar@maynardcooper.com
aty
            Jennifer Brooke Kimble
                                         jkimble@rumberger.com
            Jesse S Vogtle, Jr
                                   jvogtle@balch.com
aty
            John W. Mills
                                john.mills@btlaw.com
aty
            Joy Beth Smith
                                 joybeth@maxpopejr.com
aty
            Karl John Fingerhood
                                       karl.fingerhood@usdoj.gov
aty
            Kelley Askew Gillikin
                                        kelley.gillikin@revenue.alabama.gov
aty
aty
            Kenneth Joe Wilson, Jr
                                        kjwilson@wardwilsonlaw.com
aty
                                     kmanoukian@akingump.com
            Kristine Manoukian
aty
            Kristofor D Sodergren
                                        bknotice@rcslaw.com
                                  lapeterson@folev.com
            Lars A. Peterson
aty
                                    eisenberg.leah@arentfox.com
aty
            Leah M. Eisenberg
aty
            Lee R. Benton
                                lbenton@bcattys.com
aty
            Lindan J. Hill
                                lhill@gattorney.com
                                   mhebbeln@foley.com
            Mark F. Hebbeln
aty
            Mark P. Williams
                                   mpwilliams@nwkt.com
aty
aty
            Marty L. Brimmage, Jr.
                                        mbrimmage@akingump.com
                                    mfranklin@najjar.com
            Marvin E. Franklin
aty
            Matthew M Cahill
                                    mcahill@bakerdonelson.com
aty
            Max C. Pope, Jr
                                  max@maxpopejr.com
aty
            Melissa M. Root
                                  mroot@jenner.com
aty
```

aty Michael A Fritz, Sr aty Michael B Odom aty Michael E Bybee bankruptcy@fritzlawalabama.c modom@rumberger.com mbybee1@bellsouth.net	
aty Michael E Bybee mbybeel@bellsouth.net	
aty Michael Leo Hall mhall@burr.com	
aty Norman Matt Stockman nstockman@handarendall.	com
aty Patricia Chen patricia.chen@ropesgray.com	
aty Patrick O'Neal Gray pgray@sullivangraylaw.com	
aty Polly Misra polly.misra@nlrb.gov	
aty R. Scott Williams swilliams@rumberger.com	
aty Randolph M Fowler rfowler@pjgf.com	
aty Richard Patrick Carmody richard.carmody@arlaw.	com
aty Robert A Morgan rmorgan@rosenharwood.com	
aty Robert A Morgan rmorgan@rosenharwood.com	
aty Robert Karl Ozols rozols@maynardcooper.com	
aty Robert Moore Weaver weaver@qcwdr.com	
aty S Scott Hickman scotthickmanlaw@gmail.com	
aty Samuel Maples sam@mtandj.com	
aty Samuel Stephens sstephens@bcattys.com	
aty Shelley Bush Marmon samarmon@cjmlaw.com	
aty Stephen B Porterfield sporterfield@sirote.com	
aty Steven J. Shaw sshaw@sjslawfirm.com	
aty Susan Reid Sherrill-Beard sherrill-beards@sec.go	
aty Thomas Benjamin Humphries thumphries@sirote.co	om
aty Walter F McArdle wfm@spain-gillon.com	
aty William W Kannel wkannel@mintz.com	

TOTAL: 88

Recipients	submitted to the BNC (Bankruptcy Noticing Center):
db •	Walter Energy, Inc., et al. 3000 Riverchase Galleria Suite 1700 Birmingham, AL 35244–2359
cr	WHH Real Estate, LLC c/o Daniel D. Sparks 505 20th Street North Suite 1800 Birmingham, AL 35203
cr	Cowin & Company, Inc. c/o Daniel D. Sparks 505 20th Street North Suite 1800 Birmingham, AL 35203
cr	Nelson Brothers, LLC c/o Daniel D. Sparks 505 20th Street North Suite 1800 Birmingham, AL 35203
ba	J. Thomas Corbett Bankruptcy Administrator 1800 5th Avenue North Birmingham, AL 35203
cr	United Mine Workers of America c/o Sharon L. Levine Lowenstein Sandler, LLP 65 Livingston Avenue & 6 Becker Farm Rd Roseland, NJ 07068
intp	Steering Committee c/o Akin Gump Strauss Hauer & Feld LLP One Bryant Park Bank of America Tower New York, NY 10036–6745
intp	Wilmington Trust, National Association Corporate Capital Markets 50 South Sixth Street Ste 1290 Minneapolis, MN 55402
intp	Scott Greissman White & Case LLP 1155 Avenue of the Americas New York, NY 10036
cr	Alabama State Port Authority c/o Benjamin S. Goldman, Esquire 2001 Park Place North Suite 1200 Birmingham, AL 35203
cr	Thompson Tractor Co., Inc. c/o Benjamin S. Goldman, Esquire 2001 Park Place North Suite 1200 Birmingham, AL 35203 UNITED STATES
cr	Parker Towing Company, Inc. c/o Benjamin S. Goldman, Esquire 2001 Park Place North Suite 1200 Birmingham, AL 35203 UNITED STATES
cr	RGGS Land & Minerals, LTD., L.P. c/o Robert A. morgan ROSN HARWOOD, kPA 2200 Jack Warner Parkway, Suite 200 P. O. Box 2727 Tuscaloosa, AL 35403–2727
cr	Birmingham Rail & Locomotive, Co., Inc. Lindan J. Hill 600 University Park Place Suite 100 Birmingham, AL 35209
cr	Arch Insurance Company c/o C. Ellis Brazeal III Jones Walker LLP 1819 5th Avenue North Suite 1100 Birmingham, AL 35203
cr	Aspen American Insurance Company c/o C. Ellis Brazeal III Jones Walker LLP 1819 5th Avenue North Suite 1100 Birmingham, AL 35203
op	Kurtzman Carson Consultants LLC Attn: James Le 2335 Alaska Ave. El Segundo, CA 90245
cr	Shook and Fletcher Supply Company, Inc. c/o Stephen B. Porterfield Sirote & Permutt, P.C. 2311 Highland Avenue S. Birmingham, AL 35205
cr	G. R. Harsh Sr., Real Estate Holdings, LLC c/o Milton Harsh 110 Malaga Avenue Homewood, AL 35209
intp	Janine LaDouceur 264 Commerce Street Hawthorne, NY 10532
cr	Hager Oil Company, Inc. c/o Marvin E. Franklin Najjar Denaburg, P.C. 2125 Morris Avenue Birmingham, AL 35116
cr	S.E. Belcher, Jr. Private Foundation No. 3 c/o Jesse S. Vogtle, Jr. PO Box 306 Birmingham, AL 35201
cr	CONSOLIDATED PIPE & SUPPLY CO., INC. c/o Marvin E. Franklin Najjar Denaburg, P.C. 2125 Morris Avenue Birmingham, AL 35203
cr	Pension Benefit Guaranty Corporation 1200 K St., NW Washington, DC 20005
cr	Automotive Rentals, Inc. c/o McGlinchey Stafford 10407 Centurion Pkwy. N. Suite 200 Jacksonville, FL 32256
cr	Jefferson County Department of Health and/or Mark E. Wilson, MD 1400 Sixth Avenue South Birmingham, AL 35233
cr	Wesley West Minerals, Ltd. c/o Robert A. Morgan ROSEN HARWOOD, PA 2200 Jack Warner Parkway, Suite 200 PO Box 2727 Tusclaoosa, AL 35403–2727

intp U.S. Securities and Exchange Commission Atlanta Regional Office 950 East Paces Ferry Road, Suite 900 Atlanta, GA 30326–1382 George M. Phillippi 4 Office Park Circle, Suite 313 Birmingham,, AL 35223 cr cr Appalachian Power Company d/b/a American Electric Power c/o Eric T. Ray, Esq. Post Office Box Birmingham, ÂL 35201 Ramsay McCormack Land Co. Inc. intp c/o Lee R. Benton Benton & Centeno, LLP 2019 3rd Avenue Birmingham, AL 35203 Dominion Resources Black Warrior Trust by and through its Trustee, Southwest Bank Benton Benton & Centeno, LLP 2019 3rd Avenue North Birmingha intp c/o Lee R. Birmingham, AL 35203 Balch & Bingham LLP PO Box 306 Comerica Bank Birmingham, AL 35201 cr NATIONAL LABOR RELATIONS BOARD Region 10 Birmingham Resident Office 1130 22nd St S. cr **Suite 3400** BIRMINGHAM, AL 35205 JEFFERSON cr Frontier Enterprises Balch & Bingham LLP PO Box 306 Birmingham,, AL 35201 Mayer Electric Supply Co., Inc. 3405 4th Avenue S Attn: Mark J. Horn Birmingham, AL 35222 crcm Delaware Trust Company, as Indenture Trustee 2711 Centerville crcm Attn: Sandra E. Horwitz Wilmington, DE 19808 Road UMWA 1974 Pension Plan and Trust crcm Attn: David W. Allen 2121 K Street, N.W. Washington, DC 20037 crcm UMB Bank National Association Attn: Mark Flannagan 1010 Grand Blvd. Kansas City, MO 64106 United Steelworkers Attn: David R. Jury 60 Boulevard of the Allies, Room 807 Pittsburgh, PA crcm 15222 crcm Hager Oil Company, Inc. Attn: Philip C. Grace P O Box 1429 Jasper, AL 35502-1429 Attn: Grant Crandall 18354 Quantico Gateway Drive, Suite United Mine Workers of America crcm Triangle, VA 22172 Carroll Engineering Co. Attn: Greg Wolfe 227 Industrial Park Dr Harlan, KY 40831 crcm Consolidated Pipe & Supply Co., Inc. Attn: Chris Harper 1205 Hilltop Parkway crcm Birmingham, AL 35124 Michael Earl Carney 51140 Highway 13 Eldridge, AL 35554 cr Caterpillar Financial Services Corporation Baker, Donelson, Bearman, Caldwell & Berkowitz, cr Birmingham, AL 35203 Suite 1400 420 20th Street North Sandvik Mining and Construction USA, LLC 201 17th Street NW Suite 1700 Atlanta, GA 30363 cr Rachel L Webber ROSEN HARWOOD, PA 2200 Jack Warner Parkway, Suite 200 Post Office Box aty Tuscaloosa, AL 35403-2727 Pension Benefit Guaranty Corporation Attn: Michael Strollo 1200 K St. NW Washington, DC 20005 crcm Nelson Brothers LLC Attn: Jason K. Baker 820 Shades Creek Pkwy Ste 2000 Birmingham, AL crcm 35209 Tracy, CA 95378 intp Michael Bazley PO Box 20 GE Capital Information Technology Solutions, Inc f/d/b/a IKON Financial Services intp Bankruptcy 1738 Bass Road P O Box 13708 Macon, GA 31208-3708 Administration WHH Real Estate, LLC intp c/o Lee R. Benton Benton & Centeno, LLP 2019 3rd Avenue Birmingham, AL 35203 North Alabama Gas Corporaton c/o Brian R. Walding Walding LLC 2227 First Avenue South, Suite cr Birmingham, AL 35233 2759 County Road 63 South Jewel D Chaney Berry, AL 35546 cr 73280 Shadow Mountain Dr Unit D Robert Makohin Palm Desert, CA 92260 intp Tuscaloosa, AL 35406 intp Albert Plus, LLC 407 Vantage Point EXLP Operating, LLC Stephen B. Porterfield Sirote & Permutt, P.C. 2311 Highland Avenue cr Birmingham, AL 35205 intp University of Notre Dame c/o Lee R. Benton Benton & Centeno, LLP 2019 3rd Avenue North Birmingham, AL 35203 cr KyKennKee, Inc P.O. Box 290 Vance, AL 35490 Official Committee of Retired Employees of Walter Energy, Inc. Adams and Reese LLP 1901 6th Avenue crcm Birmingham, AL 35203 UNITED STATES OF AMERICA North, Suite 3000 Jasper, AL 35402-2390 Alabama Surface Mining Commission P. O. Box 2390 cr c/o Kristofor D. Sodergren Charles M. Cassidy Group, LLC Rosen Harwood, P.A. 2200 Jack cr Warner Parkway, Suite 200 P.O. Box 2727 Tuscaloosa, AL 35403-2727 Alabama Department of Conservation and Natural Resources c/o Kristofor D. Sodergren Rosen Harwood, cr P.A. P.O. Box 2727 Tuscaloosa, AL 35403-2727 1000 N West Street Suite 1200 Direct Fee Review LLC W. Joseph Dryer Wilmington, DE 19801 ex Birmingham Water Works 3600 1st Avenue N Birmingham, AL 35222 ba Maynard, Cooper, & Gale, P.C. Birmingham, AL 35203–2618 1901 Sixth Avenue North 2400 Maynard, Cooper and Gale aty AmSouth Harbert Plaza Southeast Fabricators, Inc. c/o Kristofor D. Sodergren Rosen Harwood, P.A. P.O. Box cr 2727 Tuscaloosa, AL 35403 Citizens' Water Service, Inc. PO Box 670 Vance, AL 35490 cr Frankie R. Cicero PO Box 126 Sumiton, Al 35148 intp Preston B. Burnett S. Scott Hickman, Atty at Law, LLC c/o S. Scott cr Tuscaloosa, Al 35401 Hickman 2600 Oracle America, Inc. c/o Shawn M. Christianson 55 Second Street, 17th **Buchalter Nemer** cr Ca, 94105 SAN FRANCISCO San Francisco Barbara Ann Chism 14123 Freeman Rd Tuscaloosa, AL 35405-9579 intp c/oTN Atty General, Bankruptcy Div PO Box 20207 TN Dept of Revenue Nashville, TN cr 37202-0207 AixPartners LLP 2000 Town Center Ste 2400 James A. Mesterharm, Managing Director Southfield, op MI 48075 The Segal Company (Eastern States), Inc. 1920 N Street NW Suite 400 Washington, DC op ACE American Insurance Company (Creditor) c/o David B. Anderson 505 N. 20th Street, Suite cr 1450 Birmingham

United States of America Joyce White Vance United States Attorney 1801 Fourth Avenue cr Birmingham, AL 35203 North Keightley & Ashner LLP 700 12th Street NW Washington, DC 20005 fa Jeffrey Blankenship 1901 6th Ave N Ste 1200 aud Ernst & Young LLP Birmingham, AL 35203 Ronnie Hodges 5023 Jiim Gogganus Rd Dora, AL 35062 intp intp Terry Eulenstein 12116 Narrow Lane Brookwood, AL 35444 Vicki R. Craig 1801 Green Street Selma, AL 36703 intp Barbara Warren 116 Daventry Dr Calera, AL 35040 intp Jeffrey Brian Watts P O Box 505 Resaca, GA 30735 intp Franklin Perdue Birmingham, AL 35207 intp 3105 29th Ave N Regions/FNBT c/o Robert A. Morgan ROSEN HARWOOD, PA PO Box 2727 Tuscaloosa, intp AL 35403-2727 intp University of Notre Dame du Lac c/o Robert A. Morgan ROSEN HARWOOD, PA PO Box 2727 Tuscaloosa, AL 35403-2727 intp Regions Bank c/o Robert A. Morgan ROSEN HARWOOD, PA PO Box 2727 Tuscaloosa, AL 35403-2727 1800 M St NW Ste 200 fa Berkeley Research Group LLC Washington, DC 20036 De-Gas c/o Jesse S. Vogtle, Jr. Balch & Bingham LLP PO Box 306 Birmingham, AL 35201 cr Caldwell & Berkowitz, PC Pardee Minerals LLC Baker, Donelson, Bearman 420 North 20th cr Street **Suite 1400** Birmingham, AL 35203 Airgas USA, LLC PO Box c/o Kathleen M. Miller Smith, Katzenstein & Jenkins, LLP cr Wilmington, DE 19801 c/o Eric T. Ray, Esq. Balch & Bingham Alabama Power Company P. O. Box cr Birmingham, AL 35201-0306 George Hunter Enis c/o Kyle B. Fonville Burnett Plaza, Suite 2000 801 Cherry Street, Unit cr Fort Worth, TX 76102 Kforce, Inc. Cabaniss Johnston 2001 Park Place North Suite 700 Birmingham, AL 35203 cr 1229-15th Place SW John Jenkins Birmingham, AL 35211 intp 420 20th Street North CSX Transportation, Inc. c/o James H. White, IV Suite cr Birmingham, AL 35203 1400 Strata Mine Services, LLC c/o James H. White, IV Baker Donelson 420 20th Street cr North Suite 1400 Birmingham, AL 35203 LLoyd, Gray, Whitehead & Monroe, P.C. c/o Lee R. Benton Benton & Centeno, LLP 2019 3rd intp Birmingham, AL 35203 Avenue North Liquidity Solutions, Inc. Liquidity Solutions, Inc. 1 University Plaza, Suite 312 cr Hackensack, NJ 07601 fa KPMG LLP **Howard Steinberg** 1350 Avenue of the Americas New York, NY 10019 750 Third Avenue 27th Floor fa FTI Consulting, Inc. New York, NY 10017 1801 Green Street Selma, AL 36703 Vicki R. Craig mv New York, NY 10017 fa PJT Partners LP 280 Park Avenue Morrison & Foerster LLP 250 West 55th Street New York, NY 10019-9601 aty Paul, Weiss, Rifkind, Wharton & Garrison 1285 Avenue of the Americas aty Allan J. Arffa New York. NY 10019-6064 Amelia C. Joiner aty Morgan, Lewis & Bockius LLP One Federal St Boston, MA 02110-1726 Andrew I. Silfen Arent Fox PLLC 1675 Broadway New York, NY 10019 aty Beth Brownstein Arent Fox PLLC 1675 Broadway New York, NY 10019 aty Bobby H Cockrell, Jr Cockrell & Cockrell 1409 University Blvd Tuscaloosa, AL 35401-1633 aty Bobby H Cockrell, Jr Cockrell & Cockrell 1409 University Blvd Tuscaloosa, AL 35401-1633 aty aty Brett Miller MORRISON & FOERSTER LLP 250 West 55th Street New York, NY 10019-9601 Bruce D. Buechler Lowenstein Sandler LLP 65 Livingston Avenue Roseland, NJ 07068 aty aty Charles B. Sklarsky Jenner & Block LLP 353 North Clark Street Chicago, IL 60654-3456 Charles L. Kerr MORRISON & FOERSTER LLP 250 West 55th Street New York, NY 10019-9601 aty 815 Walker St. #1040 Houston, TX 77002 atv Chris D. Lindstrom Cooper & Scully, P.C. Crystal R. Axelrod Morgan, Lewis & Bockius LLP 1000 Louisiana Street, Suite 4000 Houston, TX aty 77002-5005 Dan Youngblut Paul, Weiss, Rifkind, Wharton & Garrison 1285 Avenue of the Americas New aty York, NY 10019-6064 aty Daniel J. Leffell Paul, Weiss, Rifkind, Wharton & Garrison 1285 Avenue of the Americas New York, NY 10019-6064 David R. Jury Five Gateway Center Room 807 aty United Steelworkers Pittsburgh, PA 15222 Taube Summers Harrison Taylor Meinzer Br Eric J. Taube 100 Congress Avenue Suite 1800 Austin. aty TX 78701 Erica J. Richards MORRISON & FOERSTER LLP 250 West 55th Street New York, NY aty 10019-9601 Harold L. Kaplan 321 North Clark St Ste 2800 Chicago, IL 60654-5313 aty J. Alexander Lawrence MORRISON & FOERSTER LLP 250 West 55th Street New York, NY aty 10019-9601 James A. Newton MORRISON &FOERSTER LLP 250 West 55th Street New York, NY aty 10019-9601 MORRISON & FOERSTER LLP New York, NY Jennifer L. Marines 250 West 55th Street aty 10019-9601 1701 Market Street John C. Goodchild, III Morgan, Lewis & Bockius LLP Philadelphia, PA aty 19103-2921 McGuireWoods LLP Gateway Plaza 800 East Canal Street John H. Maddock, III Richmond, aty VA 23219 aty John R. Mooney Mooney, Green, Saindon, Murphy & Welch, 1920 L Street NW Suite Washington, DC 20036 Cooper & Scully, P.C. 815 Walker St. #1040 aty Julie M. Koenig Houston, TX 77002

aty Kyle B. Fonville DECKER JONES, P.C. Burnett Plaza, Suite 2000 801 Cherry Street, Unit Fort Worth, TX 76102 46 Landon S. Raiford Jenner & Block LLP 353 North Clark Street Chicago, IL 60654-3456 aty 250 West 55th Street MORRISON & FOERSTER LLP aty Lorenzo Marinuzzi New York, NY 10019-9601 Mark R. Sommerstein Ropes & Gray LLP New York, NY aty 1211 Avenue of the Americas 10035-8704 Melissa Y. Boey Morgan, Lewis & Bockius LLP 101 Park Avenue New York, NY 10178-0060 aty Michael E. Collins Manier & Hood One Nashville Place 1500 Fourth Ave N Ste aty 2200 Nashville, TN 37219 Nicole M. Brown Lowenstein Sandler LLP 65 Livingston Avenue Roseland, NJ 07068 aty Morgan, Lewis & Bockius LLP Boston, MA 02110-1726 P. Sabin Willett One Federal Street aty aty Paul Kizel Lowenstein Sandler LLP 65 Livingston Avenue Roseland, NJ 07068 1920 L Street NW Suite Paul A. Green Mooney, Green, Saindon, Murphy & Welch, aty 400 Washington, DC 20036 Peter E. Ferraro 1011 W 10th St Austin, TX 78703 aty Lowenstein Sandler LLP Phillip J. Gross 65 Livingston Avenue aty Roseland, NJ 07068 Rachel Jaffe Mauceri Morgan, Lewis & Bockius LLP 1701 Market Street Philadelphia, PA aty 19103-2921 Richard M Seltzer Cohen, Weiss & Simon LLP 330 West 42nd Street New York, NY 10036 aty Robert N. Kravitz Paul, Weiss, Rifkind, Wharton & Garrison 1285 Avenue of the Americas New aty York, NY 10019-6064 Winter McFarland LLC 205 McFarland Circle North Ruth McFarland Tuscaloosa, AL 35406 aty S. Jason Teele Lowenstein Sandler LLP 65 Livingston Avenue Roseland, NJ 07068 aty Sam H. Poteet, Jr. Manier & Hood One Nashville Place 1500 Fourth Ave N Ste aty Nashville, TN 37219 2200 Samantha Martin MORRISON & FOERSTER LLP 250 West 55th Street New York, NY aty 10019-9601 Scott C. Williams Manier & Hood One Nashville Place 1500 Fourth Ave N Ste aty 2200 Nashville, TN 37219 Sharon L. Levine Roseland, NJ 07068 Dallas, TX 75202 aty Lowenstein Sandler LLP 65 Livingston Avenue Cooper & Scully, P.C. 900 Jackson, Suite 100 aty T. Michah Dortch Thomas N Ciantra Cohen, Weiss & Simon LLP 330 West 42nd Street New York, NY 10036 aty Thomas Corbett BA Birmingham 1800 5th Avenue North Birmingham, AL 35203 smg c/o Akin Gump Strauss Hauer & Feld LLP Steering Committee One Bryant Park Bank of America smg New York, NY 10036-6745 Tower

TOTAL: 156